PDF hosted at the Radboud Repository of the Radboud University Nijmegen

The following full text is a publisher's version.

For additional information about this publication click this link. http://hdl.handle.net/2066/82618

Please be advised that this information was generated on 2017-12-06 and may be subject to change.

Problem-solving treatment during general practice residency

feasibility, and effectiveness for patients with emotional symptoms in primary care

Lieke J.A. Hassink-Franke

12-3

Problem-solving treatment during general practice residency

feasibility, and effectiveness for patients with emotional symptoms in primary care

een wetenschappelijke proeve op het gebied van de Medische Wetenschappen

Proefschrift

ter verkrijging van de graad van doctor aan de Radboud Universiteit Nijmegen op gezag van de rector magnificus prof. mr. S.C.J.J. Kortmann, volgens besluit van het college van decanen in het openbaar te verdedigen op dinsdag 12 oktober 2010 om 15.30 uur precies

door

Lieke Jacoba Aleida Hassink-Franke geboren op 4 september 1977 te Oldenzaal

Promotor:	Dhr. Prof. dr. C. Van Weel
Copromotores:	Mw. dr. E.M. van Weel-Baumgarten Dhr. dr. P.L.B.J. Lucassen Mw. dr. M.M.L. Beek
Manuscript- commissie:	Mw. Prof. dr. A.E.M. Speckens (voorzitter) Dhr. Prof. dr. G.J.M. Hutschemaekers Mw. Prof. dr. H.E. van der Horst (VUMC)
Paranimfen:	Dhr. drs. Tim C. olde Hartman Dhr. Gert J. Franke, BA

Voor mijn ouders

Contents

	Chapter 1	General introduction	9
Part I. Feasibility of problem-solving treatment training during general practice residency	Chapter 2	Feasibility of training in problem-solv- ing treatment for general practice registrars Eur J Gen Pract 2007;13(4):243-5	21
	Chapter 3	Problem-solving treatment in general practice residency: a focus group study of registrars' views Patient Educ Couns, in press	28
	Chapter 4	General practice registrars and re- search - Attitudes toward participation Austr Fam Physician 2008;37(4):276-9	29
Part II. Effectiveness of problem- solving treatment for patients with emotional symptoms in primary care	Chapter 5	Problem-solving treatment for emo- tional problems in primary care – a Cochrane review Submitted for publication	46
	Chapter 6	Effectiveness of problem-solving treatment by general practice regis- trars for patients with emotional symp- toms: a controlled trial Submitted for publication	126
	Chapter 7	General discussion	142
		Summary / Samenvatting Acknowledgments / Dankwoord List of publications Curriculum Vitae	158 166 172 174

18-9

1 General introduction

1 10-11

Background

Mental health problems are common, with one in four adults experiencing a problem.¹ They might suffer from formal mood or anxiety disorders but symptoms can also be at a sub threshold level, for instance depressed mood, anxiety, irritability, stress, sleeping problems or psychosocial problems. Prevalence rates in primary care, for instance in the Continuous Morbidity Registration (CMR) Niimegen database, vary from 12 to 33%.²⁻⁶ These rates are increasing in all sociodemographic layers of the population.⁷ The World Health Organization (WHO) Global Burden of Disease Survey estimates that by the year 2020, major depression will be second only to ischemic heart disease in the amount of disability experienced by sufferers.⁸ Disability levels in terms of impaired functioning and number of work days lost are high.⁹ Quality of life is often influenced negatively and in many patients symptoms are recurrent or chronic,¹⁰ leading to frequent visits to their general practitioner (GP).¹¹ Therefore, mental health problems account for a substantial proportion of the GP's workload

GPs are to provide and coordinate continuous and comprehensive care for the patient within the context of the patient's physical, psychological, social, cultural and existential dimensions. Therefore, GPs use a person-centered and holistic approach within their patient contacts.¹² An essential ingredient is the GP-patient partnership, because it offers the opportunity to empower patients to become active in managing their health. It has been advocated that patient empowerment should be considered as one of the essential characteristics of family medicine.13 Patient empowerment is a process of helping patients to assert control over factors that affect their health. This process begins with information and education and includes active participation of the patient in treatment decisions.¹⁴ When patients are more informed, involved, and empowered, they interact more effectively with healthcare providers and strive to take actions that will promote healthier outcomes. Contrary to the biomedical model, the empowerment model requires that patients are viewed as experts on their own lives who are responsible for their own health. This may in case of somatic illness, for instance diabetes, lead to the one patient deciding to stop smoking and the other patient deciding to lose weight. In the case of mental illness it is important to empower patients so that they learn how to deal better with their emotional symptoms. This fits in with the wish of many patients with emotional symptoms to be treated non-pharmacologically.15-17

Most patients with mental health problems are treated adequately by their GP and only a minority of patients is being referred.¹⁸ GPs often prescribe medication, usually benzodiazepines or antidepressants,18 but medication is not always appropriate: it is not always indicated, has important side effects, 19,20 patient adherence is low,²¹ the effectiveness of benzodiazepines is limited,²² and the effectiveness of antidepressants is being disputed.²³ Pharmacological treatment mainly stems from the biomedical approach, whereas patients more frequently desire the

patient-empowering approach as they prefer non-pharmacological treatments.¹⁵⁻¹⁷ Many GPs already incorporate non-pharmacological approaches in their mental health treatment, because counseling is nearly always part of the treatment by GPs.²⁴ Nevertheless, both GPs and GP registrars have expressed the wish to learn a more structured way to manage patients with mental health problems in general practice.²⁵⁻²⁸ Although some educational interventions for GPs to manage mental health problems have been studied, patient outcomes have not significantly improved.29,30 Problem-solving treatment (PST) could be an attractive non-pharmacological treatment option. It has a highly patient empowering character because the aim of PST is to teach patients how they can deal themselves with every-day problems. Patients are provided with tools to directly manage their actual behaviour on dealing with problems, rather than to manage their cognitions - as is common in cognitive behavioural therapy. Furthermore, it is a brief and practical treatment with a treatment protocol specifically designed for use by primary care workers.

Problem-solving treatment

Problem-solving treatment (PST) is a brief and practical psychological treatment, derived from cognitive behavioural therapy (CBT). In the seventies, the American psychologists D'Zurilla and Goldfried described problem-solving *therapy* as a process with five stages³¹: problem orientation, problem definition, generation of alternative solutions, decision making, and solution implementation. This therapy was based on the principles of CBT. In the nineties, the British psychiatrist Dr Mynors-Wallis developed an abbreviated form for usage in a primary care setting, and called it problem-solving *treatment*. This thesis is about the latter form.

The aim of PST is to teach patients how to use their own skills to cope with everyday life problems in a systematic way. It is assumed that symptoms reduce when patient (re) gain control over their problems. Common emotional symptoms include depressed mood, feeling stressed, loss of enjoyment in life, worries, poor concentration, feeling hopeless, and irritability. Common physical symptoms include sleep problems, bad appetite, tiredness, headaches, and non-specific pains. Most frequently mentioned areas of problems in everyday life are relational problems with partner, children, family or friends; work; money; housing; health; few recreation activities.

As Mynors-Wallis set out in his manual,³² "the first goal of PST is to increase the patient's understanding of the link between their current symptoms and their current everyday problems, and that control over such problems will help to improve symptoms. The second goal is to increase the patient's ability to clearly define their current problems. Also, the importance of setting concrete and realistic goals is stressed and practised. The third goal is to teach the patient a problem-solving procedure how to solve their problems in a structured way. The fourth goal is to generate more positive experiences regarding the patient's ability to solve problems. In this way patient's confidence in their problem-solving

1 12-13

ability and their feelings of self-control during problematic situations increase. After PST the patient should be able to cope with current and future problems and thus minimise emotional symptoms. It is a collaborative treatment with an active role for the patient who takes more control in subsequent sessions whereas the therapist's role gradually decreases." These treatment goals are aimed for via seven specific stages during each session. The seven stages are:

- 1. Explanation of the treatment and its rationale;
- Definition and breaking down of the problem;

- 3. Establishing achievable goals;
- 4. Generating multiple possible solutions;
- Evaluating and choosing the solution(s);
- Implementing the preferred solution(s);
- 7. Evaluating the outcome after the solution has been implemented.

In stage 6, the therapist and patient formulate an action plan for implementation of the preferred solution with the actual implementation happening between two sessions. Stage 7, evaluation of the implemented solution, takes place in the subsequent session and is

Case vignette

Intake

Frank is a 37-year old man. He is married and has two sons (6 and 8 years old). He visited my practice for a low mood, irritability and long lasting low back pain. Physical examination, an X-ray and a visit to the neurologist had not revealed any abnormalities of his back. He has been in sick leave for six weeks now and is not starting to feel any better. He works in a big car factory, where many colleagues have been fired due to the economic recession. Therefore, the atmosphere has changed negatively over the last few months. Furthermore, his manager is complaining about his sick leave and wants him to come back to work. At home he's feeling tired all the time, not doing any sports or social activities. He is feeling a bad father for his children and he's feeling an unpleasant husband for his wife.

First PST session

In the first session I ask him which of these problems he wants to address first. He chooses the feeling of being a bad father. We work this out via the specific problem-solving stages and define them as following:

Problem: "I'm feeling a 'bad father' because I don't do any nice things with my children anymore." Goal: "To do at least two nice activities with my children within the next two weeks."

Brainstorm: "To go to the play garden; to the cinema; cycling; swimming; painting; cooking pan cakes" Homework: "Taking them to the play garden, and preparing pan cakes with them."

Follow-up session

They ate pan cakes together, and he went to the play garden with his two children. He was very satisfied about that and had enjoyed it a lot. When asking him how he had been feeling in the play garden, he looked surprised at me and said 'Actually I felt much better at that time'.

followed by addressing a new problem via the specific stages. The treatment consists of four to six sessions over a period of approximately 8-12 weeks with a duration of no more than 30 minutes, except for the first session which may last 60 minutes. During each session the therapist uses a PST work sheet: see appendix A.

PST is effective in depression.33-36 Also, there is some evidence that it has positive effects for anxiety.^{37,38} unexplained physical symptoms,³⁹ deliberate self-harm,⁴⁰ personality disorders,⁴¹ and palliative care.⁴² Various self-management programmes that include problem-solving, for instance for diabetes, low back pain and osteoarthritis, showed improved self efficacy and patient outcomes.43-⁴⁵ PST has been shown to be effective when delivered by different therapists, varying from psychiatrists and psychologists to nurses and research GPs. without relevant differences in effectiveness. These therapists were, however, all unfamiliar to the patient. There is no evidence available for PST being provided by patients' own physicians. Furthermore, most studies included distinct groups of patients - such as major and minor depression or dysthymia - whereas GPs are mostly confronted with a typical mixture of emotional symptoms or with patients not meeting diagnostic (DSM-IV) criteria for depressive and/or anxiety disorders. For this category of patients there is only little evidence for or against PST.^{37,46} There have been performed three systematic reviews on PST.⁴⁷⁻⁴⁹ However, these reviews did not specifically focus on studies performed in primary care nor did they specifically address the effectiveness

of PST for the broad range of mental health problems seen in primary care.

The fact that PST is effective for many (mental) health problems implies that problem-solving skills can be used more widely as a tool for patient empowerment in general. Together with its brief character, this makes problem-solving an attractive technique for GPs. However, PST was not yet available in the Dutch primary care setting.

This study

Worldwide, teaching of the management of mental health problems in residency programmes has largely focused on accurate diagnostic skills and appropriate prescribing,28 whereas GP registrars would welcome learning more counseling skills and an effective tool for non-pharmacological treatment of emotional symptoms.⁵⁰ By introducing PST in the GP residency programme we expected two advantages: a. the registrars' need could be met in an early stage, and b. the residency programme would provide optimal training and supervision circumstances. In the Netherlands, GP registrars follow a three-year residency programme. In the first and third year they are attached to a primary care practice in the community. In the second year they do a rotation between emergency room, nursing home and mental health care. Throughout the whole programme they come back to university one day every week for supervision and training in groups. An American pilot study with 11 GP registrars being trained in PST showed that registrars can be trained successfully in PST but the authors

1 14 - 15

recommended further investigation with a larger sample of registrars and evaluation of patient outcomes.⁵¹ We aimed to include the broad range and/ or mixture of emotional symptoms which characterises general practice as GPs often see mixed symptoms rather than specific, full blown DSM-IV disorders.⁵²

Therefore, in this thesis the following research questions are addressed:

- I. What is the feasibility of PST training during GP residency?
- II. What is the effectiveness of PST delivered by trained GP registrars for patients with recurrent or chronic emotional symptoms?

We planned to compare the effectiveness of PST versus usual care for patients with emotional symptoms in a pragmatic randomised controlled clinical trial. PST and usual care were provided by GP registrars who were in their last year before qualifying as a GP. As we aimed to include the broad range of emotional symptoms typically seen in the primary care setting, we defined emotional symptoms as sub threshold as well as formal disorders of depressed mood, anxiety, stress, irritability, sleep disturbance, and psychosocial problems. With the terms emotional symptoms, emotional problems and mental health problems we meant the same category of patients. Thus, in this thesis the terms emotional symptoms, emotional problems and mental health problems are interchangeable.

Outline of this thesis Part I. Feasibility of PST training during GP residency

Chapter two describes a feasibility study of GP registrars who were randomised to PST training. By observation and questionnaires, we measured registrars' participation in the training, the number of patients treated with PST, and registrars' opinions about the training and treatment.

Chapter three shows the views of GP registrars on PST in general practice and on PST during GP residency. For this objective we performed a qualitative focus group study.

Chapter four addresses the attitudes of GP registrars toward participation in research. One of the barriers we experienced during our study was the resistance of registrars to participate in the research project. We, therefore, were interested in their views on participation in research in general. We assessed their participation and opinions through observation and a questionnaire.

Part II. Effectiveness of PST for patients with emotional symptoms in primary care

Chapter five shows the results of a systematic review according to the principles of the Cochrane Collaboration. We aimed to systematically review the evidence for and against the effectiveness of PST and to perform a meta-analysis, which generally provides a better overall estimate of a clinical effect than the results from individual studies. The review focused on the effectiveness of PST provided by specifically trained GPs, nurses or other primary care health workers in primary care patients with all kinds of emotional problems.

Chapter six describes the results of an effectiveness study. We performed a controlled clinical trial to compare the effectiveness of PST and usual care in primary care patients with emotional symptoms. The main outcomes were symptom severity and quality of life.

Chapter seven contains a general discussion about the methods and findings of this thesis and ends with recommendations for general practice and GP residency programmes and for further research.

References

- 1 Verhaak PF, Hoeymans N, Garssen AA, Westert GP. Mental health in the Dutch population and in general practice: 1987-2001. Br J Gen Pract 2005 Oct;55(519):770-5.
- 2 van Rijswijk E. Mental Health Problems in General Practice an explorative study on diagnosis and treatment. Thesis. Radboud University Nijmegen Medical Centre; 2005.
- 3 Gulbrandsen P, Fugelli P, Hjortdahl P. Psychosocial problems presented by patients with somatic reasons for encounter: tip of the iceberg? Fam Pract 1998 Feb;15(1):1-8.
- 4 Charles J, Britt H, Fahridin S, Miller G. Mental health in general practice. Aust Fam Physician 2007 Apr,36(4):200-1.
- 5 olde Hartman T, van Rijswijk E, van Ravesteijn H, Hassink-Franke L, Bor H, van Weel-Baumgarten E, et al. Mental health problems and the presentation of minor illnesses: data from a 30-year follow-up in general practice. Eur J Gen Pract 2008;14 Suppl 1:38-43.

- 6 King M, Nazareth I, Levy G, Walker C, Morris R, Weich S, et al. Prevalence of common mental disorders in general practice attendees across Europe. Br J Psychiatry 2008 May;192(5):362-7.
- 7 Hodiamont PP, Rijnders CA, Mulder J, Furer JW. Psychiatric disorders in a Dutch Health Area: a repeated cross-sectional survey. J Affect Disord 2005 Jan;84(1):77-83.
- Murray CJ, Lopez AD. Alternative projections of mortality and disability by cause 1990-2020: Global Burden of Disease Study. Lancet 1997 May 24;349(9064):1498-504.
- 9 Lecrubier Y. The burden of depression and anxiety in general medicine. J Clin Psychiatry 2001;62 Suppl 8:4-9.
- 10 van Weel-Baumgarten EM, van den Bosch WJ, van den Hoogen HJ, Zitman FG. The longterm perspective: a study of psychopathology and health status of patients with a history of depression more than 15 years after the first episode. Gen Hosp Psychiatry 2000 Nov;22(6):399-404.
- 11 Katon W, Von KM, Lin E, Lipscomb P, Russo J, Wagner E, et al. Distressed high utilizers of medical care. DSM-III-R diagnoses and treatment needs. Gen Hosp Psychiatry 1990 Nov;12(6):355-62.
- 12 Wonca Europe. The European Definition of General Practice / Family Medicine. Barcelona, Spain; 2002.
- 13 Mola E, De Bonis JA, Giancane R. Integrating patient empowerment as an essential characteristic of the discipline of general practice/family medicine. Eur J Gen Pract 2008;14(2):89-94.
- 14 Lau DH. Patient empowerment--a patientcentred approach to improve care. Hong Kong Med J 2002 Oct;8(5):372-4.
- 15 Lowe B, Schulz U, Grafe K, Wilke S. Medical patients' attitudes toward emotional problems and their treatment. What do they really want? J Gen Intern Med 2006 Jan;21(1):39-45.
- 16 Corney RH. A survey of professional help sought by patients for psychosocial problems. Br J Gen Pract 1990 Sep;40(338):365-8.

- 17 Fritzsche K, Armbruster U, Hartmann A, Wirsching M. Psychosocial primary care - what patients expect from their General Practitioners A cross-sectional trial. BMC Psychiatry 2002 May 8;2(1):5.
- 18 van Rijswijk E, Borghuis M, van de Lisdonk E, Zitman F, van Weel C. Treatment of mental health problems in general practice: a survey of psychotropics prescribed and other treatments provided. Int J Clin Pharmacol Ther 2007 Jan;45(1):23-9.
- 19 Lader MH. Limitations on the use of benzodiazepines in anxiety and insomnia: are they justified? Eur Neuropsychopharmacol 1999 Dec;9 Suppl 6:S399-S405.
- 20 Williams JW, Jr., Mulrow CD, Chiquette E, Noel PH, Aguilar C, Cornell J. A systematic review of newer pharmacotherapies for depression in adults: evidence report summary. Ann Intern Med 2000 May 2;132(9):743-56.
- 21 van Geffen EC, Gardarsdottir H, van Hulten R, van Dijk L, Egberts AC, Heerdink ER. Initiation of antidepressant therapy: do patients follow the GP's prescription? Br J Gen Pract 2009 Feb;59(559):81-7.
- 22 Lader M. Effectiveness of benzodiazepines: do they work or not? Expert Rev Neurother 2008 Aug;8(8):1189-91.
- 23 Kirsch I, Deacon BJ, Huedo-Medina TB, Scoboria A, Moore TJ, Johnson BT. Initial severity and antidepressant benefits: a meta-analysis of data submitted to the Food and Drug Administration. PLoS Med 2008 Feb;5(2):e45.
- 24 Robinson JW, Roter DL. Counseling by primary care physicians of patients who disclose psychosocial problems. J Fam Pract 1999 Sep;48(9):698-705.
- 25 Kerwick S, Jones R, Mann A, Goldberg D. Mental health care training priorities in general practice. Br J Gen Pract 1997 Apr;47(417):225-7.
- 26 Sahhar D, O'Connor D. How well do Australian medical schools prepare general practitioners to care for patients with mental disorders? Australas Psychiatry 2004 Mar,12(1):26-30

- 27 Oopik P, Aluoja A, Kalda R, Maaroos HI. Family doctors' problems and motivating factors in management of depression. BMC Fam Pract 2006;7:64.
- 28 van Marwijk H. How to improve mental health competency in general practice training?--a SWOT analysis. Eur J Gen Pract 2004 Jun;10(2):61-5.
- 29 Gask L, Dowrick C, Dixon C, Sutton C, Perry R, Torgerson D, et al. A pragmatic cluster randomised controlled trial of an educational intervention for GPs in the assessment and management of depression. Psychol Med 2004 Jan;34(1):63-72.
- 30 Huibers MJ, Beurskens AJ, Bleijenberg G, Van Schayck CP. The effectiveness of psychosocial interventions delivered by general practitioners. Cochrane Database Syst Rev 2003;(2):CD003494.
- 31 D'Zurilla TJ, Goldfried MR. Problem solving and behavior modification. J Abnorm Psychol 1971 Aug;78(1):107-26.
- 32 Mynors-Wallis L. Problem-solving treatment for Anxiety and Depression: A practical guide. Oxford University Press; 2005.
- 33 Mynors-Wallis LM, Gath DH, Lloyd-Thomas AR, Tomlinson D. Randomised controlled trial comparing problem solving treatment with amitriptyline and placebo for major depression in primary care. BMJ 1995 Feb 18;310(6977):441-5.
- 34 Mynors-Wallis LM, Gath DH, Day A, Baker F. Randomised controlled trial of problem solving treatment, antidepressant medication, and combined treatment for major depression in primary care. BMJ 2000 Jan 1;320(7226):26-30.
- 35 Dowrick C, Dunn G, Ayuso-Mateos JL, Dalgard OS, Page H, Lehtinen V, et al. Problem solving treatment and group psychoeducation for depression: multicentre randomised controlled trial. Outcomes of Depression International Network (ODIN) Group. BMJ 2000 Dec 9;321(7274):1450-4.

- 36 Barrett JE, Williams JW, Jr., Oxman TE, Frank E, Katon W, Sullivan M, et al. Treatment of dysthymia and minor depression in primary care: a randomised trial in patients aged 18 to 59 years. J Fam Pract 2001 May;50(5):405-12.
- 37 Catalan J, Gath DH, Anastasiades P, Bond SA, Day A, Hall L. Evaluation of a brief psychological treatment for emotional disorders in primary care. Psychol Med 1991 Nov;21(4):1013-8.
- 38 Mynors-Wallis L, Davies I, Gray A, Barbour F, Gath D. A randomised controlled trial and cost analysis of problem-solving treatment for emotional disorders given by community nurses in primary care. Br J Psychiatry 1997 Feb;170:113-9.
- 39 Wilkinson P, Mynors-Wallis L. Problemsolving therapy in the treatment of unexplained physical symptoms in primary care: a preliminary study. J Psychosom Res 1994 Aug;38(6):591-8.
- 40 Townsend E, Hawton K, Altman DG, Arensman E, Gunnell D, Hazell P, et al. The efficacy of problem-solving treatments after deliberate self-harm: meta-analysis of randomised controlled trials with respect to depression, hopelessness and improvement in problems. Psychol Med 2001 Aug;31(6):979-88.
- 41 Huband N, McMurran M, Evans C, Duggan C. Social problem-solving plus psychoeducation for adults with personality disorder: pragmatic randomised controlled trial. Br J Psychiatry 2007 Apr;190:307-13.
- 42 Wood BC, Mynors-Wallis LM. Problem-solving therapy in palliative care. Palliat Med 1997 Jan;11(1):49-54.
- 43 Didjurgeit U, Kruse J, Schmitz N, Stuckenschneider P, Sawicki PT A time-limited, problemorientated psychotherapeutic intervention in Type 1 diabetic patients with complications: a randomised controlled trial. Diabet Med 2002 Oct;19(10):814-21.

- 44 van den Hout JH, Vlaeyen JW, Heuts PH, Zijlema JH, Wijnen JA. Secondary prevention of work-related disability in nonspecific low back pain: does problem-solving therapy help? A randomised clinical trial. Clin J Pain 2003 Mar,19(2):87-96.
- 45 Heuts PH, de Bie R, Drietelaar M, Aretz K, Hopman-Rock M, Bastiaenen CH, et al. Self-management in osteoarthritis of hip or knee: a randomised clinical trial in a primary healthcare setting. J Rheumatol 2005 Mar;32(3):543-9.
- 46 Schreuders B, van Marwijk H, Smit J, Rijmen F, Stalman W, van Oppen P. Primary care patients with mental health problems: outcome of a randomised clinical trial. Br J Gen Pract 2007 Nov;57(544):886-91.
- 47 Bell AC, D'Zurilla TJ. Problem-solving therapy for depression: a meta-analysis. Clin Psychol Rev 2009 Jun;29(4):348-53.
- 48 Malouff JM, Thorsteinsson EB, Schutte NS. The efficacy of problem solving therapy in reducing mental and physical health problems: a meta-analysis. Clin Psychol Rev 2007 Jan;27(1):46-57.
- 49 Cuijpers P, van SA, Warmerdam L. Problem solving therapies for depression: a meta-analysis. Eur Psychiatry 2007 Jan;22(1):9-15.
- 50 Ratcliffe J, Gask L, Creed F, Lewis B. Psychiatric training for family doctors: what do GP registrars want and can a brief course provide this? Med Educ 1999 Jun;33(6):434-8.
- 51 Hegel MT, Dietrich AJ, Seville JL, Jordan CB. Training residents in problem-solving treatment of depression: a pilot feasibility and impact study. Fam Med 2004 Mar;36(3):204-8.
- 52 Nease DE, Jr., Volk RJ, Cass AR. Investigation of a severity-based classification of mood and anxiety symptoms in primary care patients. J Am Board Fam Pract 1999 Jan;12(1):21-31.

Problem-solving worksheet

1. Problem

2. Goal(s)

,			•										•	• •				 • •	• •	ł								-	 •	•		-		-					• •	•		 •	• •				•	•	• •	• •	• •	• •	
		• •	•		 ·	• •	• •	•	• •	• •	• •	• •	•	• •	•		•	 	• •	• •	 	• •	•	• •	•	•	•	÷	 •	• •	•		• •	•	1	•	•	•	• •	• •	• •	 •	• •	•	-	• •	• •	•	• •		• •	• •	
	 	•													•	 • •												ł	 ł				1							• •								ł					

3. Solutions

a)	••••••	pros:	cons:
b)		pros:	cons:
C)		pros:	cons:
d)		pros:	cons:
e)	••••••	pros:	cons:

4. Choice of solution(s)

.....

5. Steps to achieve solution (homework)

a)	• •	• •	• •	• •		• •	• •	• •	•			• •	• •		• •	• •	•		• •		• •	• •		• •	•		• •				• •	•	•	• •	•		• •			• •	•	• •	• •	•			 	• •	• •	•••	• •	• •	• •		• •				
	• •	• •	• •	• •	•	• •	• •	• •	•		• •	• •			• •	• •	·							• •	•	• •	• •			• •	• •	• •	• •			• •			-	• •	•		• •	•	• •	• •	 • •	• •		• -	• •	• •			• •		•		• •
b)	• •		• •				• •	• •	•	• •		• •			• •	• •	•	• •	• •		• •	• •		• •			• •		• •	• •		•		• •			• •	• •		• •					• •			• •			• •	• •			• •	• •			•••
	• •		• •	• •					•	• •					• •		·	• •				• •					• •		• •	• •		•					• •	• •							• •	• •	 	• •			• •	• •						• •	•••
C)	• •	• •	• •	• •		• •		• •	•	• •	• •		• •				•	• •		•	• •	• •				• •	• •	•	• •	• •		•	•	• •	•	• •	•••	• •			·	• •		•	• •	• •	 	• •		• •	• •	• •		• •	• •		•		•••
	• •	• •	• •	• •	• • •	• •	• •	• •	•	• •	• •	• •	• •	·	• •		·	• •	• •	·	• •	• •	•	• •	•	• •	• •	•	• •	• •			• •	• •	•	• •	• •	• •		• •	•	• •	• •	•	• •	• •		• •		• •	• •	• •		• •	• •	• •		• •	•••
d)	• •	• •	• •	• •		• •	• •	• •	•			• •	• •		• •	• •	•		• •		• •	• •		• •			• •		• •	•	• •	•	•	• •	•		• •			• •	•	• •	• •	•			 	• •			• •	• •	• •		• •				•••
	• •	• •	• •	• •			• •	• •	•			• •			• •	• •	·							• •	•		• •				• •	• •							-	• •	•			•	• •		 • •	• •		• -	• •	• •					•		• •
Ne	Next appointment																																																										

6. Evaluation

•••••	

2 20 - 21

2 Feasibility oftraining in problem-solving treatmentfor generalpractice registrars

Lieke J. A. Hassink-Franke Evelyn M. van Weel-Baumgarten Peter L.B. J. Lucassen Mechtild M.L. Beek Laurence M. Mynors-Wallis Chris van Weel

Published in: European Journal of General Practice 2007;13(4):243-5

Presented at: WONCA Europe Conference of Family Doctors 2004, Amsterdam, the Netherlands WONCA World Conference of Family Doctors 2004, Orlando, Florida, USA

2 22 - 23

Introduction

In primary care patients frequently present with emotional symptoms and/or psychosocial problems.¹ Most of these patients are treated without specialist referral.² Usual care by general practitioners (GPs) mostly consists of counseling and medication.^{2,3} However, medication is not always the best option^{4,5} nor does it meet patients' preferences,6 and the effectiveness of counseling is unclear. Many GPs and GP registrars have expressed the need for more practical skills training in managing emotional symptoms.^{7,8} We therefore introduced training in problemsolving treatment (PST) into the GP residency programme. PST is a brief, psychosocial intervention teaching the patient a systematic way of dealing with problems of everyday life. The treatment has shown effectiveness in depression and other emotional symptoms,⁹⁻¹¹ and it also improved self-efficacy in patients with diabetes, low-back pain, and osteoarthritis.12-¹⁴ This implies that problem solving skills can be used as a tool for patient empowerment in general. Together with its brief character, this makes problem solving an attractive technique for GPs. In this pilot study, we assessed the feasibility of PST training in GP residency.

Methods

We performed an observational study with GP registrars in the GP residency programme in the Netherlands. We randomly selected 21 third-year GP registrars, who were in their last year of training to qualify as a GP. After training in PST, registrars selected and treated patients with emotional or psychosocial problems with PST in their teaching practice. By observation and questionnaires, we measured: registrars' participation in the training; time spent on training, supervision, and treatment; number of patients treated with PST; registrars' performance in treating patients; and registrars' opinions about the training and treatment.

Problem-solving treatment and Training

PST is a brief psychological treatment. derived from cognitive behavioural therapy, specifically developed for primary care.¹⁵ Patients receive an explanation about the link between their emotional symptoms and current psychosocial problems. They learn how to use their own skills to resolve problems and improve their symptoms. PST comprises the following stages: clarification of the problems; establishing achievable goals; generating solutions; selecting and implementing preferred solution; and evaluation. The full treatment consists of four to six sessions of about 30 minutes (first session 60 minutes). The registrars received a two-day training by experienced PST trainers, followed by five months' supervision on treatment of patients in the registrars' practice.

Results

Twenty registrars completed the twoday training and evaluated it as useful (see Table 1). They all practised PST and treated in total 52 patients under supervision (median 2.5 patients per registrar; range 1-5). The median number of sessions per patient was 3 (range 1-6). All 20 registrars received feedback on their treatment techniques. Average attendance rate during the three group-feedback sessions (1½ hours each) was 70%. Sixteen registrars received individual feedback (5-30 minutes per contact). Overall, the supervisor reported good performance of PST in 18 registrars.

At the end of the supervised treatment period, 17 registrars completed a questionnaire (see Table 1). Aspects that registrars liked most included the clarifying character of the treatment (n=7), thinking in terms of practical solutions (n=7), the active role of the patient (n=4), and helping patients to increase insight in and control over their own problems (n=4). Aspects that registrars liked less included the extra time they needed for treatment (n=10) and the difficulty in defining their role as a GP whilst providing psychological treatment (n=3). Some registrars (n=4) liked the structure of the treatment; others (n=5) disliked it. At the end of their third year, 15 registrars completed the last questionnaire: 14 assessed education in PST as positive, and 14 assessed the amount of time necessary for treatment as negative.

Table 1 Registrars' opinions about the PST training programme

Item of questionnaire	Average on 5-point scaleª (range)
	(20 respondents)
- Usefulness overall	3.8 (3-5)
- Relevance of lectures	4.0 (1-5)
- Relevance of role-play	4.2 (2-5)
- Relevance of video feedback	3.9 (1-5)
- Feeling of being sufficiently trained	3.8 (3-5)
Supervised treatment period (20 participants):	(17 respondents)
Usefulness of group feedback	3.6 (2-4)
- Usefulness of individual feedback	3.9 (3-5)
Difficulty with selecting appropriate patients	3.8 (2-5)
- Difficulty with patients willing to start	2.6 (1-4)
Difficulty with defining the problem clearly	3.7 (2-5)
Difficulty with stating achievable goals	3.6 (2-5)
- Difficulty with finding solutions	2.8 (1-4)

Discussion

This study demonstrated that training in PST is feasible during GP residency. Despite the compulsory character of the training and some critical comments, registrars stated that it brought them relevant, new skills. It met the need for practical skills training in the treatment of emotional symptoms in primary care. We therefore recommend PST training to be implemented as a core part in GP residency programmes. Experience from the US suggests that registrars will continue using PST, or parts of it, once qualified.¹⁶ This will be primarily for patients with mental health problems. However, the skills can be used more generically for a much larger part of the primary care population. It is most satisfying that our registrars themselves stressed the generic character of problem-solving skills: they appreciated having learnt

how to clarify patients' problems, to think of practical solutions, and to activate patients. These patientempowering skills are useful in many other patient contacts and do not have to take much time. The fact that registrars themselves expressed this potential of problem-solving skills supports our reason of introducing PST during residency.

Acknowledgments

We thank all GP registrars for their cooperation in this project. We also thank their tutors in the practices and their tutors at the Radboud University Nijmegen Medical Centre for giving the registrars the opportunity to participate in this project. We also thank ZonMw, The Netherlands Organisation for Health Research and Development, who financially supported the PST project.

References

- Gulbrandsen P, Fugelli P, Hjortdahl P. Psychosocial problems presented by patients with somatic reasons for encounter: tip of the iceberg? Fam Pract 1998 Feb;15(1):1-8.
- 2 van Rijswijk E, Borghuis M, van de Lisdonk E, Zitman F, van Weel C. Treatment of mental health problems in general practice: a survey of psychotropics prescribed and other treatments provided. Int J Clin Pharmacol Ther 2007 Jan;45(1):23-9.
- 3 van Weel-Baumgarten EM, van den Bosch WJ, Hekster YA, van den Hoogen HJ, Zitman FG. Treatment of depression related to recurrence: 10-year follow-up in general practice. J Clin Pharm Ther 2000 Feb;25(1):61-6.
- 4 Lader MH. Limitations on the use of benzodiazepines in anxiety and insomnia: are they justified? Eur Neuropsychopharmacol 1999 Dec;9 Suppl 6:S399-S405.
- 5 de Abajo FJ, Rodriguez LA, Montero D. Association between selective serotonin reuptake inhibitors and upper gastrointestinal bleeding: population based case-control study. BMJ 1999 Oct 23;319(7217):1106-9.
- 6 Fritzsche K, Armbruster U, Hartmann A, Wirsching M. Psychosocial primary care what patients expect from their General Practitioners A cross-sectional trial. BMC Psychiatry 2002 May 8;2(1):5.
- 7 Kerwick S, Jones R, Mann A, Goldberg D. Mental health care training priorities in general practice. Br J Gen Pract 1997 Apr;47(417):225-7.
- 8 Sahhar D, O'Connor D. How well do Australian medical schools prepare general practitioners to care for patients with mental disorders? Australas Psychiatry 2004 Mar;12(1):26-30.

- 9 Mynors-Wallis LM, Gath DH, Lloyd-Thomas AR, Tomlinson D. Randomised controlled trial comparing problem solving treatment with amitriptyline and placebo for major depression in primary care. BMJ 1995 Feb 18;310(6977):441-5.
- 10 Barrett JE, Williams JW, Jr., Oxman TE, Frank E, Katon W, Sullivan M, et al. Treatment of dysthymia and minor depression in primary care: a randomised trial in patients aged 18 to 59 years. J Fam Pract 2001 May;50(5):405-12.
- 11 Catalan J, Gath DH, Anastasiades P, Bond SA, Day A, Hall L. Evaluation of a brief psychological treatment for emotional disorder in primary care. Psychol Med 1991 Nov;21(4):1013-8.
- 12 Didjurgeit U, Kruse J, Schmitz N, Stuckenschneider P, Sawicki PT. A time-limited, problemorientated psychotherapeutic intervention in Type 1 diabetic patients with complications: a randomised controlled trial. Diabet Med 2002 Oct;19(10):814-21.
- 13 van den Hout JH, Vlaeyen JW, Heuts PH, Zijlema JH, Wijnen JA. Secondary prevention of work-related disability in nonspecific low back pain: does problem-solving therapy help? A randomised clinical trial. Clin J Pain 2003 Mar;19(2):87-96.
- 14 Heuts PH, de Bie R, Drietelaar M, Aretz K, Hopman-Rock M, Bastiaenen CH, et al. Self-management in osteoarthritis of hip or knee: a randomised clinical trial in a primary healthcare setting. J Rheumatol 2005 Mar;32(3):543-9.
- 15 Mynors-Wallis L. Problem-solving treatment a practical manual. Oxford: Oxford University Press; 2005.
- 16 Hegel MT, Dietrich AJ, Seville JL, Jordan CB. Training residents in problem-solving treatment of depression: a pilot feasibility and impact study. Fam Med 2004 Mar,36(3):204-8.

3 28 - 29

3 Problem-solving treatment in general practice residency: a focus group study of registrars' views

Lieke J.A. Hassink-Franke Tim C. olde Hartman Mechtild M.L. Beek Chris van Weel Peter L.B.J. Lucassen Evelyn M. van Weel-Baumgarten

Patient Education and Counseling, in press

Abstract

Introduction

General practitioner (GP) registrars often express the need for more expertise of non-pharmacological treatments for patients with mental health problems. Problem-solving treatment (PST) could be an attractive option. We aimed to explore GP registrars' views on PST training during residency and on the actual use of PST in general practice.

Methods

We performed a qualitative study with four focus groups, interviewing 18 Dutch registrars who had been trained in PST during residency. Data were analysed according to the principles of constant comparative analysis.

Results

Registrars thought that PST training during residency was feasible, interesting and helpful, but found that it took too much time in everyday practice and was not a GP's task. All registrars, however, said they would use specific elements in a variety of consultations, for instance concretising problems, brainstorming about practical solutions, and activating patients.

Discussion

Registrars regarded PST training during residency feasible and helpful. In daily practice they would apply specific elements of the treatment. We recommend residency programmes to offer training in PST or another psychological treatment with comparable elements. Training should fit in with the registrars' needs and level of training and experience.

Introduction

Many general practitioners (GPs) perceive educational needs with regard to the management of mental health problems. Mental health problems are common - with one in four adults experiencing a problem¹ and further increasing in prevalence in primary care. The vast majority of these patients are treated by their own GPs,² causing a considerable proportion of GPs' workload. In the teaching and training of (future) GPs the emphasis is on diagnostic skills and pharmacological treatment, at the expense of non-pharmacological treatment.³ Nevertheless, support and counseling are used in almost every case,³⁷ but interventions for GPs to manage mental health problems did not improve patient outcomes significantly.^{8,9} GPs and GP registrars - doctors in training to become a GP - have expressed the need for more specific, practical (counseling) skills training in managing emotional symptoms.^{3,10,11} This fits in with the many patients who prefer non-pharmacological treatment within primary care for their mental health problems.¹²⁻¹⁴ In this light, a different focus of mental health training in residency programmes has been proposed.15

Problem-solving treatment (PST) is a brief, practical psychological intervention, teaching the patient a systematic way to gain control over their own (mental health) problems of everyday life.¹⁶ (Re)gaining control will decrease symptoms. PST is effective in the treatment of depression and emotional symptoms.^{17,18} And there are indications that it is effective for unexplained physical symptoms¹⁹, in palliative care,²⁰ and in stroke patients.²¹ It has a maximum of six sessions, each limited to no more than 30 minutes. Furthermore, PST incorporates skills such as patient empowerment that can be used in many patient contacts outside mental health problems. This makes it an attractive technique for GPs. PST training could meet the needs of registrars to learn a non-pharmacological intervention in an early stage of their career. An American pilot study showed that registrars are able to learn PST.²²

We introduced training in PST into a Dutch GP residency programme. An earlier Dutch study confirmed that PST training was feasible during GP residency,²³ but a number of registrars did not apply it to eligeable patients.²⁴ The aim of this study was to explore the registrars' views on PST and its use in general practice for patients with emotional symptoms.

Methods

We conducted focus group interviews with GP registrars in Nijmegen, the Netherlands, nested in a clinical trial.²⁴ This trial had randomised 20 thirdyear registrars into PST treatment for emotional symptoms and provided PST training during their residency training. PST is a brief psychological treatment, derived from cognitive behavioural therapy, aiming to increase patients' awareness of a link between everyday life problems and symptoms, and to teach them to cope with these everyday problems in a systematic way.¹⁶ Each PST session comprises seven stages in which problems are identified and clarified, goals are set, and solutions are worked out (Figure 1). This is a collaborative process between GP and patient. with the patient gradually taking over control. By learning better problemsolving skills patients can deal better with current and future problems. The treatment consists of four to six consultations over a period of approximately 8-12 weeks with a duration of no more than 30 minutes each, except for the first session which may last 60 minutes. The registrars were trained by experienced PST trainers in a two-day course and trainers supervised treatment during 5 months through individual and group feedback sessions.²³ This was followed by a six-month period of the clinical trial.

Focus group interviews took place at the end of the trial (February 2004). We chose focus group interviews rather than individual interviews because of the benefit of group interaction which stimulates participants to explore and clarify their views into more depth.²⁵ We grouped the registrars into four focus groups, taking into account their gender and their attitude towards PST. We received information about their attitudes from one of the trainers. who met the registrars regularly during group feedback sessions in the 5-month supervised treatment period. By putting registrars with negative attitudes in one and the same group, they got ample space for their (negative) comments. The minority of male registrars were spread over the groups. Two GP registrars were unable to attend the focus group interviews. leaving 18 participants (12 females) for two groups with four and two with five registrars. Their mean age was 32 (range 27-41) and they had treated on average 4.3 patients with PST each (range 1-9).

The interviews lasted approximately 60 to 90 minutes and were facilitated by two independent skilled moderators (one psychologist, one GP) and by using an interview guidebook. Through the guidebook (see Table 1) registrars' views were explored on: a. PST as a treatment for patients with emotional symptoms; b. PST in general practice; c. PST during GP residency. Both enabling and disabling factors for its use were discussed. In the fourth focus group interview saturation

Figure 1. Problem-solving treatment (PST)

- A brief psychological treatment with 7 stages:
- 1.Explanation and rationale
- 2. Clarification and definition of the problems
- 3.Establishing achievable goals
- 4. Generating solutions through brainstorming
- 5. Selecting preferred solution
- 6.Implementing solution
- 7. Evaluation of progress

was reached with no new major themes arising. The interviews were audio-recorded with the participants' consent and ethical approval for the project was provided by the University of Nijmegen.

Analysis

Data collection and analysis was conducted as an iterative process which means that the three researchers (LH, ToH, EvW) added relevant topics to the guidebook after a preliminary analysis of each interview. All interviews were transcribed verbatim. The first author checked the transcripts and entered all data into Atlas ti, a software package to support the analysis of qualitative data. According to the principles of constant comparative analysis²⁶ in which transcripts are subsequently thematically coded, transcripts were read and re-read by LH, ToH and EvW to identify recurring themes. They independently made a first categorisation by coding meaningful sentences. Initial codes were discussed, seeking agreement on their content, and then grouped into themes to identify registrars' views on PST. Recurrent and important themes were frequently discussed and refined as part of an ongoing iterative process.²⁷ The emerging themes were then reread

To enhance the validity of our findings we triangulated our data by comparing them with the results of a questionnaire survey among registrars of the subsequent year.²⁸ The written answers to these open questions were qualitatively analysed by two researchers.

Results

Five major themes evolved from the focus groups: registrars' 1. views on GPs' role in mental health care; 2. positive experiences with PST; 3. negative experiences with PST; 4. views on PST in general practice; and 5. views on PST during residency. In general, those registrars who had treated more patients with PST were more positive about PST than those who had treated fewer patients. We did not find major differences between registrars who were grouped as having a more positive attitude and those with a more negative attitude.

Role of GPs in mental health care

Registrars expressed that mental health care certainly is part of the GP's function. Firstly, GPs should be able to diagnose mental health problems. And secondly, they should be able to manage mental health problems up to a certain level. They saw their role more in support than in treatment of these problems, and in particular in case of relatively new and uncomplicated problems. They would refer patients who had more complicated or chronic problems, or who needed more long-lasting counseling, or who did not improve. They emphasised the importance of the GP-patient relationship: in a good relationship they would tend to counsel longer, as this was often associated with better outcome. Applying a specific psychological treatment is not the task of a GP, according to the registrars, although it could be the interest of some GPs.

"I would like to be able to map the problems out and diagnose them and then it's either done or I can refer them or maybe I can help them - not in the sense of developing a long-term treatment plan but more in the sense of, uhm, offering support. I can't do much more than that." (14, male, FG4)

"I do think I can play a role but that role would be small. That means short treatments. I would do that myself [...] If I think that it is going to take longer or that the problems are more complicated, then I would be more inclined to refer patients to someone else." (01, male, FG1)

Positive experiences with PST

Most registrars appreciated the structure of PST as this provided them with a model how to set the agenda of the consultation. They were happy with addressing 'here and now' problems because it made the treatment very practical. Also, they all liked that PST forced them to make the problems of the patients as concrete as possible. This enabled them to define problems, goals and homework more precisely. Furthermore, they explicitly stated they liked the activating character because the treatment has a patientcentred approach in which the goals and solutions are set and carried out by the patient. Specific parts of PST they had appreciated were linking symptoms to problems, the patients' insight in their problems, focusing at only one problem during each session, setting goals, and brainstorming about solutions.

"I think that it is a good method for exploring things concretely and it gives people a tool by which they can start to work on things right away. It gets them started." (05, female, FG2)

"You guide the patient but, really, it's the patient that does the work (sounds of agreement); he or she comes up with the solutions." (07, female, FG2)

"What I think is really good is that you can help the patient to explore his or her own problems and they learn something from that. I think it's good that they then can start dealing with those problems on their own. So it promotes independence." (06, female, FG2)

"I do make that link now between the complaints and the, the reasons behind them - the way all things [symptoms and problems] are related to each other. That is definitely a useful, uh, useful element." (13, female, FG3)

They had experienced PST sometimes as a diagnostic tool, because in some cases many more problems became clear during treatment. They also appreciated that patients came to the insight that symptoms decrease when they work on their problems. Most registrars found it advantageous to teach patients how they can look at and manage problems, as these skills can be used in future problems too.

"It is not just the treatment at that point in time but also the method or the way of thinking that you try to give them so that they can later apply it themselves." (01, male, FG1)

Negative experiences with PST

The most important negative experiences were: 1.the structure of PST; 2. registrars' competence; 3. doubt about the effectiveness; 4. time investment.

Some registrars described the structure of PST as rigid or artificial and not giving them enough space to listen to patients' new stories, symptoms or problems because the psychological treatment kept them focusing on the problem of today's session. They felt hindered to react empathically. Furthermore, it was emphasised that PST can only address some problems and only to some extent, whereas some patients brought up many more (complicated) problems. "If people have had complaints again. then, all of the sudden, they end up sitting down with their general practitioner who then doesn't want to talk about physical complaints but actually wants to work on other problems, and I found that difficult sometimes. It was difficult to keep those two things separate for me but it was particularly difficult for the patients and it was hard to be clear about that [...] It was like you had to say. uh. 'If you want to deal with the physical complaints, then you need to come back for another appointment." (03, female, FG1)

"What I experienced every time, or at least almost every time, was that people come back with 'yeah, but' and then mention their fatigue again or whatever they came for, and, yeah, you

Table 1. Focus group interview guidebook

1.	Non-pharmacological treatment of emotional symptoms	 How do you regard your role in the non-pharmacological treatment of emotional symptoms? Is there a need for a tool to treat emotional symptoms? How did you treat emotional symptoms before PST training?
2.	Problem-solving treatment (PST)	 What do you think of PST? When or what patients suit PST? When not? What do you think about the role of the doctor versus the role of the patient? What do you think about the effectiveness of PST?
З.	Feasibility of PST in general practice	 What is the place of PST in general practice? Can GPs provide PST? Why (not)? What are the most useful elements of PST? What do you think of your competence of PST?
4.	Feasibility of PST during GP residency	 Do you think PST could and/or should be part of the residency programme? Why (not)? What elements of PST are especially helpful during residency?
don't really have enough time to go into those things because you have to stick to the PST protocol and I have to admit that I find that kind of restrictive." (06, female, FG2)

"It is that the method is so fixed, and I think that's what I struggle with [...] It's like it's all or nothing." (03, female, FG1)

Most registrars judged their competence of PST as unsatisfactory. They said they had had too little education and experience in this field. The perceived lack of competence demotivated some registrars to start PST. Furthermore, they found it hard to leave it to the patient to do the work.

"I think that I've only had a small taste of it, at least that's how it feels. And it kind of feels like that's a little unfair to patients because you offer the patients something that I am not sure I can give them in the way it is intended." (05, female, FG2)

"I think that, in general, we are 'do-ers' and so we want to determine how it can best be done." (16, female, FG4)

Registrars had doubts about the effectiveness of PST, because it did cost a lot of their energy and time but resulted in varying outcomes. Some believed the effect to be the result of time and attention. The likelihood that symptoms would improve over time without doing a formal treatment like PST was a barrier to start PST. Once treatment results were less positive than expected, some registrars perceived this as a barrier to apply PST in further patients. "Some patients have really been helped with it but there have also been a few patients whereby I really wasn't satisfied. I would like to see some more evidence." (03, female, FG1)

"I have consciously chosen not to suggest it but I've thought to myself 'What on earth am I subjecting this patient to with PST?' (sounds of agreement). I think that a lot of issues resolve themselves in time and with a few consultations instead of six, uh, half hour consultations. And it demands a lot from the patient, which doesn't always seem to be better." (02, female, FG1)

Registrars experienced time investment as the most important barrier. They thought that PST took too much time for an individual patient. They stated that the strong structure and repeating, long sessions did not fit into the common structure of Dutch general practice with a 10-minute consultation system. Some of them had solved this problem by doing only 3 to 4 sessions with each patient or giving the patient homework.

"I think that the way in which the treatment is conducted does not fit well with general practice. The number of consultations and the length of those consultations [...] I think it demands too much time from a family doctor's practice [...] and it messes up your consultation hours. It is a lot of hassle and it is very different work than what we are used to." (02, female, FG1)

"I have to say that I indeed held three consultations and then I was pretty much done with it. I dealt with two problems and when we got to the third, I sent the patient home with homework and said, 'Write up a list of problems and do your homework.' I wouldn't want to do a six session PST." (09, female, FG3)

We identified two groups of registrars in terms of more flexible and more rigid. The first group regarded PST as a helpful tool they tried to use in a tailor-made way for many patients. They were satisfied with their competence. The other group regarded PST as an inflexible method without any space for the patient. They saw only few indications for PST and did not feel very competent. The first group was more positive about PST than the latter group.

"I noticed that the more, uh, you get used to it and develop the skills, the more you can let go and that you then kind of make it your own thing." (07, female, FG2)

"You don't really consider who is sitting across from you. The procedure is pretty much the same for everyone." (01, male, FG1)

PST in general practice

Most registrars thought that PST did not fit into general practice because of the aforementioned time investment, structure and doubts about the effectiveness. There were some more barriers. Some registrars emphasised that they were not happy with their role during PST. They felt a 'double role' of being a psychologist and a GP at the same time. They thought it could be a barrier for patients to come to them as a GP again after PST, because patients might see them as a psychologist.

"Perhaps that's why people haven't come back to see me. Perhaps they've gone to another GP for their other complaints because I treated their emotional symptoms. I can imagine that patients would find it more difficult to come back to me for just a sore toe or something." (02, female, FG1)

Many registrars found it hard to decide on the proper indications for PST. They thought PST was applicable in only a small group of patients. They thought that the more longstanding the GPpatient relationship was, the easier it was to decide which patients would benefit from PST.

"I think that the training GPs are more suitable because they have already built a relationship with the patient and they are better able to assess the patient." (18, female, FG4)

Nevertheless, most registrars were able to specify the indications for PST. They thought it was especially appropriate for patients with recently arisen, daily life problems - such as problems with work, relationship, family, house, et cetera - or patients who put forward their problems so that it became a chaos resulting in symptoms of mild depression, burn-out or varying physical symptoms. These patients don't see where to start managing their problems. According to the registrars, PST could give them the feeling of being in control of their lives again. Some registrars stated that

they did not think PST was appropriate for patients with many or complex problems because they thought they could reach too few effects, whereas others were happy to make at least a starting point.

"I think that it is particularly useful for giving people a sense that they can have control over their own lives and that they can deal with their problems again, and, yeah, it is not really that you can treat someone's depression with it but it does give people something to hold onto and gives them the idea: 'I can do something about this.'" (11, male, FG3)

"You try to solve something and I find that that kind of falls short. People have such huge problems and you can only deal with a tiny little part of those problems in those few sessions." (05, female, FG2)

"But even if there are more problems, you have to start somewhere, right?" (03, female, FG1)

According to the registrars PST was less suitable for patients with moderate, severe or long-lasting depressive symptoms and contraindicated in suicidal patients. Also, patients with (alcohol) addiction, personality disorders or dependent personality traits, and patients who could not link symptoms to problems did not benefit from PST.

"A major depression or a minor depression: I don't think I can [handle that]." (02, female, FG1)

"What is contraindicated?" (Moderator)

"Alcohol abuse, addition." (02, female, FG1)

"I don't think it works that well with all patients because often they cannot make that link where, uhm, they move away from focusing on the complaints and towards dealing with the problems." (06, female, FG2)

Some registrars thought that it also depended on specific patient characteristics, for PST to be appropriate. They thought it was easier in patients with some intellectual capacities, whereas others thought that the patient's character and motivation was more important: patients must be willing to take the initiative to improve their situation.

"I don't think intelligence would be a barrier at all but it is more about whether or not it fits with a particular personality. [...] I think it really fits with someone who wants to take things into their own hands but with someone who is quite dependent on others and who, uhm, maybe is looking for a compliment, I don't think it fits as well with those kinds of people." (11, male, FG3)

"I would compare it to quitting smoking: If someone isn't motivated, I can put all my energy into trying to motivate him but I don't really want to do that, and I don't think it would have much effect." (01, male, FG1)

Because of the above reasons, not all GPs should learn PST, according to the registrars. Nevertheless, they emphasised that GPs are able to master PST and that this would be interesting for GPs with a particular interest in mental health care. Other GPs might in their view involve a (mental health) nurse or colleague-GP in their practice to actually perform the PST intervention. This would be interesting for patients, as registrars experienced that patients were happy to receive treatment for this type of problems in their own practice, rather than being referred.

"I think it is a good method and I think it would work just fine with a psychologist or a nurse but, as a GP, I just don't have enough time for it." (05, female, FG2)

"[Similar] to how we choose a certain drug, it's a method that needs to fit with somebody and if it does, then you choose that method." (05, female, FG2)

"I think that an advantage for the patient is the fact that it's offered at the GP's practice. It lowers the threshold significantly." (06, female, FG2)

Instead of using the entire treatment, registrars said that they would rather use elements of it, for instance concretising problems, setting goals, or brainstorming about solutions. Also, they preferred leaving more parts of PST as homework with the patient. By doing so, they thought that PST sessions could fit into regular double consultations of 20 minutes.

"I think that PST contains elements that I can use very well. For example, making things concrete and, uh, trying to work towards a goal and, uh, giving people homework." (03, female, FG1) "Maybe a double consultation, twenty minutes. I think that would probably be enough, you could quickly go through all the steps and then I would probably give people more homework so they do more of the work at home." (07, female, FG2)

PST during residency

Before PST training, registrars felt that their diagnostic skills in mental health problems were satisfying but that this was not the case for their management skills. All registrars expressed the need for a more practical, structured tool to manage mental health problems and they thought that the current residency programme did not provide enough tools.

"Despite the fact that we have known, for a long time, that many of the complaints that present to the GP have a psychological aspect, I think that the GP residency programme pays too little attention to that." (10, male, FG3)

"I have always been frustrated that I didn't have any tools to deal with that. It has always been limited to just talking through things and those conversations are relatively unstructured, [...] sure, people can get things off their chests but there was nothing concrete you could do for them." (09, female, FG3)

"I have never really been trained in how to conduct a, uhm, therapeutic treatment, not even during my psychiatry rotation where I had really hoped to have been taught something like this." (06, female, FG2) "It seems to be useful to have some kind of structure that, let's say, outlines how you should approach that." (18, female, FG4)

Registrars were positive about the PST training, because they found PST a useful and practical technique. Registrars thought that PST training during residency was feasible. However, in the opinion of the registrars the training should have been spread throughout the whole 3-year residency programme rather than being put solely in the 'overfilled' third year. Also, registrars had experienced the limitation of being in training: they felt that many patients with mental health problems consulted their own GP rather than a registrar. Whether PST should be a standard part of the residency programme was being discussed: some thought it would be useful, especially the twoday training. All registrars thought, however, it could well be an optional part for interested registrars.

"I found the [two day] course to be very useful." (06, female, FG2)

"I think that if it were to be incorporated in the residency programme, that it should not be crammed into the third year [...] but that maybe it should be spread out over the entire programme." (12, female, FG3)

"When I hear from four people who have done it that they will not use it in the future, then I wonder if it should be incorporated in the residency programme where registrars already have too little time to learn what they need to learn [...] Perhaps it should be an elective." (03, female, FG1)

Discussion and Conclusion

Discussion

This study aimed to explore the views of GP registrars on training of PST during residency and the use of PST in general practice for patients with emotional symptoms. It showed that registrars perceived a shortcoming in the residency programme with regard to providing them with skills to manage mental health problems. They stated that PST training during residency was feasible, interesting and helpful. They, however, thought that the training had been intensive within one year, and therefore suggested to spread the training over the full three years of their residency training. All said they would use particular elements of PST in general practice, rather than applying the entire treatment which they did not regard as part of the GP's tasks. They especially appreciated concretising problems, brainstorming about practical solutions, and activating patients. They thought that implementing specific elements could fit into relatively regular 20-minute consultations. This corresponds with Hegel's study in which registrars mostly used abbreviated versions of PST, when evaluated at 3-year followup after training.²² Furthermore, the mentioned elements fit in with the point of view on mental health care in general practice, published by the Dutch College of General Practitioners, stressing the importance of defining problems and helping patients to use their own

resources to solve problems.²⁹

Our findings fit in with research among experienced GPs. For instance, our registrars mentioned the same barriers and enablers of using PST as GPs did in an earlier Australian study about PST.³⁰ Both mentioned the useful, practical character of the treatment, which fits into the pragmatic nature of many GPs' approach. Both regarded time as a precondition. Also, both registrars and GPs experienced the change from giving advice to facilitating the patient finding a solution that the patient owns. Although our registrars thought this was sometimes difficult to do, they valued this treatment characteristic highly. In a Danish study, GPs considered non-specific factors such as attentive listening very important.³¹ Our registrars too thought that good listening is important but they thought that PST did not always give them enough space to do so, because they felt pressured by the structure of the treatment. They often experienced the structure as a handicap, which is striking since we know that most young physicians are positive about using highly structured guidelines for somatic illnesses.³² Perhaps they need some more confidence both in the contact with patient with emotional problems and in 'playing' with the treatment structure

In some aspects, the registrars expressed different experiences than experienced GPs. Because it takes time to build a good doctorpatient relationship and because this relationship is needed in treating patients with emotional symptoms, they experienced a disadvantage of working in the same training practice for just one year. Obviously, they were aware of the importance of the GP-patient relationship, which is known to correlate highly with patient outcomes.³³ Due to the short-standing relationships they had difficulties deciding on 'the adequate patients' for PST. On the other hand they were able to mention many indications for PST, which reflects the relatively general character of PST.

The registrars stated that applying a full psychological treatment did not belong to the GP's tasks. In the light of the registrars' experience that mental and physical health problems are interwoven frequently, combined with the strong wish of patients to be treated by their GP,³⁴ this attitude can be seen as somewhat unsatisfactory, as this interplay of mental and physical health problems was reason to propagate the integration of mental health into primary health care.³⁵ An important factor that might play a role is that some registrars are not intensively enough exposed to mental health problems, because these patients prefer to contact their own GP rather than the resident as patients value longstanding relationships. It stresses the need - for residency programmes - to secure that registrars see the whole range of primary care patients, including those with mental health problems. This may strengthen their perception of their role in mental health care. Furthermore, registrars might still be predominantly biomedically oriented in order not to miss any illness. Considering these arguments, PST training could

probably be optimised by offering them generic elements of PST in an early stage of residency and repeat these in more detail in a later stage when they perceive a greater need.

In our view, the registrars expressed an ambiguity towards their involvement in mental health. They complained of a lack of tools provided in the residency programme - a complaint that has been aired before.^{11,36} But at the same time. they objected to the compulsory nature of the PST training module, and expressed a restricted task for themselves and for GPs in general in mental health. In order to solve this ambiguity, as there is currently a strong advocacy to intensify GPs' and primary care's role in mental health,³⁵ we propose to insistently discuss this antinomy within residency programmes.

There were some limitations in our study. We evaluated in this focus group study the experiences with the very first Dutch PST training in residency. It is likely that this training suffered from its newness, influencing registrars' experiences negatively. On the other hand, the availability of a supervisor may have facilitated the use of PST. Also, registrars had been obliged to participate in the PST study, which caused a lot of resistance amongst them and made them experience PST training as 'something extra' in an already full programme. This influenced their opinions negatively.

It was a strength of our study that we involved a relatively large number of registrars who had all participated in the full training programme. By doing a qualitative study we explored and clarified what they thought about PST during residency and about PST in daily practice. We succeeded to collect both positive and negative experiences. Triangulation of our data showed that the positive and negative experiences identified in this focus group study were more or less identical to those expressed in a questionnaire of another group of registrars.

This qualitative study examined registars' perceptions about PST and not their actual behaviour. By studying these perceptions we identified the experienced needs, barriers and enablers which can help in the implementation of PST or other psychological treatments in residency and everyday practice. Tape-recording the discussions, multiple coding during analysis and our triangulation strategy added to the rigour of the study.

Conclusion

Registrars thought that PST training is feasible and helpful during GP residency, especially because they had perceived a lack during residency in the tools being provided for the management of emotional symptoms. All registrars liked to use specific elements of the treatment in a variety of consultations within general practice. They especially appreciated concretising problems, brainstorming about solutions, and activating patients. However, they thought that the entire treatment costs too much time in everyday practice and is not part of the GP's tasks.

Practice implications

We recommend residency programmes to offer training in PST or another psychological treatment with comparable useful elements. It should be part of the standard programme. rather than being part of a research project in order to avoid 'study resistance'. Based on the experiences with the registrars of these focus groups, we have implemented an adapted PST training programme in the first year of residency in Nijmegen since 2008. This programme focuses specifically on the generic elements of concretising, brainstorming and activating patients. We encourage our registrars to find multiple uses for their skills to increase the learning process and to transfer these skills into their practice patterns. Evaluation results of this new programme were good. Subsequently, we offer an optional programme with training in the full psychological treatment in the second and third year. This will be evaluated in a later stage. Offering a training programme with generic skills at the start of residency and more specific skills in a later stage seems to fit in with the registrars' needs and their level of training and experience.

In general, we highly recommend residency programmes to address the ambiguity of registrars' need for practical skills training against their restricted view on GPs' task in mental health.

Based on our results and those of an Australian study with GPs experiencing PST as useful,³⁰ PST training might also be useful for GPs as a support in their management of mental health problems. In the Netherlands, the Dutch College of General Practitioners is offering a PST training since 2007. We recommend ongoing follow-up with booster sessions to consolidate the new skills, because training then is most likely to be effective.³⁷

Acknowledgments

We kindly thank all GP registrars who took the time and effort to contribute to our study and our focus group interviews.

Reference List

- Verhaak PF, Hoeymans N, Garssen AA, Westert GP. Mental health in the Dutch population and in general practice: 1987-2001. Br J Gen Pract 2005 Oct; 55(519):770-5.
- 2 van Rijswijk E. Mental Health Problems in General Practice - an explorative study on diagnosis and treatment. Thesis. Radboud University Nijmegen Medical Centre; 2005.
- 3 Sahhar D, O'Connor D. How well do Australian medical schools prepare general practitioners to care for patients with mental disorders? Australas Psychiatry 2004 Mar;12(1):26-30.
- 4 Robinson WD, Geske JA, Prest LA, Barnacle R. Depression treatment in primary care. J Am Board Fam Pract 2005 Mar;18(2):79-86.
- van Weel-Baumgarten EM, van den Bosch WJ, Hekster YA, van den Hoogen HJ, Zitman FG.
 Treatment of depression related to recurrence:
 10-year follow-up in general practice. J Clin
 Pharm Ther 2000 Feb;25(1):61-6.
- 6 Rijswijk van E. Mental Health Problems in General Practice - an explorative study on diagnosis and treatment Nijmegen; 2005.
- 7 Robinson JW, Roter DL. Counseling by primary care physicians of patients who disclose psychosocial problems. J Fam Pract 1999 Sep;48(9):698-705.
- 8 Gask L, Dowrick C, Dixon C, Sutton C, Perry R, Torgerson D, et al. A pragmatic cluster randomised controlled trial of an educational intervention for GPs in the assessment and management of depression. Psychol Med 2004 Jan;34(1):63-72.
- 9 Huibers MJ, Beurskens AJ, Bleijenberg G, Van Schayck CP. The effectiveness of psychosocial interventions delivered by general practitioners. Cochrane Database Syst Rev 2003;(2):CD003494.

- 10 Kerwick S, Jones R, Mann A, Goldberg D. Mental health care training priorities in general practice. Br J Gen Pract 1997 Apr;47(417):225-7.
- 11 Ratcliffe J, Gask L, Creed F, Lewis B. Psychiatric training for family doctors: what do GP registrars want and can a brief course provide this? Med Educ 1999 Jun;33(6):434-8.
- 12 Priest RG, Vize C, Roberts A, Roberts M, Tylee A. Lay people's attitudes to treatment of depression: results of opinion poll for Defeat Depression Campaign'just before its launch. BMJ 1996 Oct 5;313(7061):858-9.
- 13 Lowe B, Schulz U, Grafe K, Wilke S. Medical patients' attitudes toward emotional problems and their treatment. What do they really want? J Gen Intern Med 2006 Jan;21(1):39-45.
- 14 Fritzsche K, Armbruster U, Hartmann A, Wirsching M. Psychosocial primary care - what patients expect from their General Practitioners A cross-sectional trial. BMC Psychiatry 2002 May 8:2(1):5.
- 15 van Marwijk H. How to improve mental health competency in general practice training?--a SWOT analysis. Eur J Gen Pract 2004 Jun;10(2):61-5.
- 16 Mynors-Wallis L. Problem-solving treatment a practical manual. Oxford: Oxford University Press; 2005.
- 17 Mynors-Wallis LM, Gath DH, Lloyd-Thomas AR, Tomlinson D. Randomised controlled trial comparing problem solving treatment with amitriptyline and placebo for major depression in primary care. BMJ 1995 Feb 18;310(6977):441-5.
- 18 Dowrick C, Dunn G, Ayuso-Mateos JL, Dalgard OS, Page H, Lehtinen V, et al. Problem solving treatment and group psychoeducation for depression: multicentre randomised controlled trial. Outcomes of Depression International Network (ODIN) Group. BMJ 2000 Dec 9;321(7274):1450-4.

- 19 Wilkinson P, Mynors-Wallis L. Problem-solving therapy in the treatment of unexplained physical symptoms in primary care: a preliminary study. J Psychosom Res 1994 Aug;38(6):591-8.
- 20 Wood BC, Mynors-Wallis LM. Problem-solving therapy in palliative care. Palliat Med 1997 Jan;11(1):49-54.
- 21 Robinson RG, Jorge RE, Moser DJ, Acion L, Solodkin A, Small SL, et al. Escitalopram and problem-solving therapy for prevention of poststroke depression: a randomised controlled trial. JAMA 2008 May 28:299(20):2391-400.
- 22 Hegel MT, Dietrich AJ, Seville JL, Jordan CB. Training residents in problem-solving treatment of depression: a pilot feasibility and impact study. Fam Med 2004 Mar;36(3):204-8.
- 23 Franke LJ, van Weel-Baumgarten EM, Lucassen PL, Beek MM, Mynors-Wallis L, van Weel C. Feasibility of training in problemsolving treatment for general practice registrars. Eur J Gen Pract 2007;13(4):243-5.
- 24 Hassink-Franke LJA, van Weel-Baumgarten EM, Lucassen PLBJ, Wierda E, Engelen MWM, Beek MM, et al. Effectiveness of Problemsolving treatment by family practice registrars for patients with emotional symptom: a controlled trial. submitted 2010.
- 25 Kitzinger J. Qualitative research. Introducing focus groups. BMJ 1995 Jul 29;311(7000):299-302.
- 26 Glaser BG, Strauss AL. The discovery of grounded theory. Chigago: Aldine; 1967.
- 27 Britten N. Qualitative interviews in medical research. BMJ 1995 Jul 22;311(6999):251-3.
- 28 Casey D, Murphy K. Issues in using methodological triangulation in research. Nurse Res 2009;16(4):40-55.
- 29 NHG-Standpunt Toekomstvisie Huisartsenzorg. GGZ in de huisartsenzorg. 2007.

- 30 Pierce D, Gunn J. GPs' use of problem solving therapy for depression: a qualitative study of barriers to and enablers of evidence based care. BMC Fam Pract 2007;8:24.
- 31 Davidsen A. Experiences of carrying out talking therapy in general practice: a qualitative interview study. Patient Educ Couns 2008 Aug;72(2):268-75.
- 32 Wiersma D, Geijer RRM. Practice Guidelines 2010 from the Dutch College of General Practitioners. Houten: Bohn Stafleu van Loghum; 2009.
- 33 Lambert MJ, Barley DE. Research summary on the therapeutic relationship and psychotherapy outcome. Psychotherapy 2001;38(4):357-61.
- 34 Brody DS, Khaliq AA, Thompson TL. Patients' perspectives on the management of emotional distress in primary care settings. J Gen Intern Med 1997 Jul;12(7):403-6.
- 35 WHO. Integrating Mental Health Into Primary Health Care: A Global Perspective. World Health Organization; 2008.
- 36 Gask L, Dixon C, May C, Dowrick C. Qualitative study of an educational intervention for GPs in the assessment and management of depression. Br J Gen Pract 2005 Nov;55(520):854-9.
- 37 Gask L, Goldberg D, Lewis B. Teaching and learning about mental health. In: Gask L, Lester H, Kendrick T, Peveler R, editors. Handbook of primary care mental health. London: Gaskell; 2009.

4 46 - 47

4 General practice registrars and research Attitudes toward participation

Lieke J.A.Hassink- Franke Thijs Kommers Evelyn M. van Weel-Baumgarten Peter L.B.J. Lucassen Mechtild M.L. Beek Henk J.M. van den Hoogen Chris van Weel

Published in: Australian Family Physician 2008;37(4):276-9

Presented at: NAPCRG Annual Meeting 2005, Quebec City, Canada WONCA World Conference of Family Doctors 2007, Singapore, Singapore

4 48 - 49

Abstract

Introduction

Early exposure of general practitioners to research is recommended to increase family medicine research capacity. However, vocational training programmes encounter difficulties in engaging general practice registrars in research projects. We investigated registrars' opinions of research and their participation in research in daily practice.

Methods

Sixty- seven Dutch general practice registrars participated in a trial concerning patients with emotional symptoms. We assessed the registrars' participation and opinions through observation and a questionnaire.

Results

Response rate was 82%. Registrars recruited 208 patients. The participants liked learning a new skill and participating in research. Obligatory participation, lack of time and difficulties with patient recruitment were important barriers to participation.

Discussion

Registrars report that participation in research during vocational training is interesting but that it should not be compulsory, and that they prefer to choose their own research subjects. We recommend implementing an attractive research program during vocational training.

Introduction

Primary care research is important¹ but there is general concern about the difficulty of linking research to patient care.^{2,3} Research development is even more problematic in primary care than in other disciplines.⁴ Policies to enhance general practice research include the creation of research networks,⁵ collaborations with research institutes,6 and early exposure to research during undergraduate teaching and specialty training.⁷ Early exposure prompts students to consider research as part of their future career, and better equips future practitioners to deliver evidence based patient care.8 The importance of early exposure to research is generally accepted by both general practice registrars and directors of vocational training programmes.^{7,9} Most programmes include research curricula or related activities, including training in knowledge and skills, conducting research projects, or participation in research in daily practice.^{8,10} Although research curricula create more positive attitudes toward research,¹¹ there are no indications that more registrars are participating in research,¹² and we still know little about the long term effects of such curricula.^{8,12} Furthermore, most studies have assessed changes in registrars' attitudes toward performing a research project or undertaking education in research skills during vocational training, rather than assessing registrars' participation in research during daily practice.^{9,11} This study analyses registrars' participation in research tasks during their daily work with patients. The aim was to assess patient recruitment, factors influencing recruitment, and

registrars' views and suggestions with regard to participation.

Methods

Setting & design

Dutch general practice registrars undertake a 3 year specialty training program. They spend the first and third year in a training practice in the community, and in the second year they rotate between hospital posts. In 2003 and 2004 our training program included participation of all third year registrars (70) in a controlled clinical trial as part of their core program.

The trial

A controlled clinical trial of the effectiveness of registrars using problem solving treatment (PST) - a brief psychological treatment to teach patients how to use their own skills to cope with problems - for patients with emotional symptoms. It is theoretically assumed that symptoms are reduced if problems can be resolved.13 Registrars were randomly allocated to either the intervention group or the control group. Both groups recruited patients with emotional symptoms during their regular clinical work. We asked each registrar to recruit 4-6 patients who had presented for three or more consultations in the past 6 months, had a score of four or more on the 12 item general health questionnaire (GHQ-12), and who experienced emotional symptoms. Exclusion criteria were severe medical illness, current contact with psychiatric services (or contact in the past year), current psychological treatment or past cognitive behavioral therapy, severe mental disorder, organic psychiatric disorder or

4 50-51

substance misuse, active suicidal ideas, and lack of sufficient Dutch language to participate. Registrars in the intervention group received a 2 day training course in PST¹⁴ and provided the psychological treatment to the patients they had recruited within 8 months. Registrars in the control group provided 'care as usual' and were asked to complete their patient recruitment within 4 months. The trial design was approved by the Medical Ethics Committee of the Radboud University Nijmegen Medical Centre, The Netherlands.

Outcomes and analysis

We administered a self developed questionnaire to explore registrars' opinions about their participation in the trial, barriers they experienced in patient recruitment, and their opinions and suggestions with regard to enhancing research participation. Recruitment data were obtained from the trial records

Recruitment data and scaled answers from the questionnaire were analysed with descriptive statistics and independent sample t-tests using

SPSS statistical analysis software. The answers to open ended questions were independently ordered into categories.

Results

Sixty-seven of the 70 registrars participated in the trial (37 in intervention group [27 women] and 30 as controls [18 women]). Registrars randomly allocated to the intervention or control aroup in 2003 expressed resistance to obligatory participation. We modified the process for the 2004 cohort. offering registrars an individual choice to participate in the training. Registrars who participated in PST training comprised the intervention group (17); the others, providing usual care, were regarded as the control aroup. The registrars were also offered more research assistance, and both groups were given 8 months to recruit patients.

Patient recruitment

The registrars in the 2003 intervention group recruited 83 patients; the registrars in the control group

Table 1 Recruitment of patients

	2003		2004		p-value
	Mean number of patients per registrar	Range	Mean number of patients per registrar	Range	
Intervention group Control group* Total	4.2 1.0 3.0	1-9 0-3	3.9 2.5 3.2	2-6 0-5	0.653 0.007 0.781

* Control group recruitment in 2003 took 4 months; in 2004 8 months (intervention group recruitment in both years 8 months).

recruited 11 patients. The registrars in the 2004 intervention group recruited 66 patients; the control group 48 (Table 1). We explicitly asked registrars in 2004 to describe the difficulties they had experienced recruiting patients (Table 2).

Registrars' opinions and suggestions

The questionnaire had a response rate of 84% (30 of 37 registrars in the intervention group, 26 of 30 registrars in the control group [p=0.54]). Positive points reported by the registrars included the interesting and relevant nature of the topic, the opportunity to learn a new skill, becoming acquainted with and contributing to research and evidence based medicine, good research support, becoming more attentive to diagnosing and treating emotional problems, and developing a critical view.

Negative points included the time investment required (this was especially mentioned by intervention group registrars). All registrars in 2003 criticised obligatory participation; only control group registrars were negative about obligatory participation in the 2004 cohort.

To improve participation, registrars suggested they be allowed to choose between several research projects to better match the research topic with their personal interests. They felt they needed to spend enough time in clinical practice training, early and good information about a research project, and involvement of their general practitioner tutors in patient recruitment.

Response to question 'Did the following reason play	Numbers of registrars (%)		
a role in the inclusion of patients?' in questionnaire	Yes	Νο	Missing
Lack of patients who met inclusion criteria	18 (58)	12 (39)	1 (3)
Lack of time	16 (52)	14 (45)	1(3)
Patients refusing to start the intervention*	10 (67)	4 (27)	1(7)
Patients refusing to participate in research	12(39)	18 (58)	1(3)
Too many administrative actions	12(39)	17 (55)	2 (6)
Difficulties in explaining the research	8 (26)	23 (74)	
Difficulties leaving the role as a GP and asking	8 (26)	22(71)	1 (3)
patients for research participation			
Lack of patients with emotional symptoms	3 (10)	28 (90)	

 Table 2
 Important factors in the recruitment of patients (2004, 31 respondents)

* Intervention group (15 respondents)

In 2003, all respondents said that they wanted to be involved in research in the future; in 2004 this was the case for 14 of 15 intervention group respondents and nine of 13 control group respondents (Table 3).

Discussion

Sixty-seven registrars recruited 208 patients in total. Registrars expressed an interest in participation and appreciated contributing to research. They enjoyed learning a new skill and being more attentive to a particular disease and/or symptoms. Nevertheless, their patient recruitment rate was below our expectations. Initially the obligatory nature of participation was considered to be an important barrier. Engaging registrars by offering the choice to voluntarily take part in the training did not however, result in major improvements in recruitment: the doubled patient recruitment rate in the control group can be attributed to the doubled recruitment period. Registrars suggested that the option to choose an interesting and relevant research topic, the opportunity to learn new skills and a report of the research results would make participation in research projects more attractive.

The recruitment rate in this trial does not differ from the moderate recruitment rates recorded by GPs: Peto et al,¹⁵ for instance, found an average rate of 3.7 patients recruited per GP per annum. Registrars' barriers and wishes are also comparable with those of GPs: moderate patient recruitment because of time pressures, the need for interesting,

Table 3Requirements that make research projects attractive for
participation during vocational training (2004, 31 respondents)

Requirements of re	Mean score on 5-point scale*	
Most important	The topic should be interesting	4.7
	There should be a report of the research results	4.6
Less important	l should have enough time	4.4
	l should learn something, e.g. a skill	4.2
	The project should be well adapted to the practice	4.0
	There should be feedback on my own performances	4.0
Least important	Participation should contribute to my own career	3.2
	l should have a say in the project	2.8
	There should be a financial reward	2.4

* 1= very unimportant, 5=very important

practice oriented and relevant projects, personal support, good information, and good feedback about the research results.^{3,15-17}

Strengths and limitations

As far as we are aware, this is the first study exploring general practice registrars' actual performance in research. The study explored the attitudes of registrars toward research rather than opinions of GPs or directors of vocational training programmes. Furthermore, it studies registrars' opinions about participation in a trial within their routine practice rather than requiring them to conduct a research project themselves. The study is limited however, by the modest sample of registrars, all of whom belonged to the same training program and who participated in a single trial alone. However, the findings are similar to those of other studies concerning registrars' appreciation of research experience.^{7,11} The study was compromised by the change made in 2004 to select registrars on the basis of their motivation. In 2003, however, we observed variation in the registrars' selection and recruitment of patients. which was related to their individual motivation as expressed during supervision sessions. For this reason we believe the actual effects of the change were limited.

Implications for general practice

Research experience during medical school is associated with postgraduate research involvement.¹⁸ Assuming that this applies to registrars too – and assuming a desire for research to be part of the culture of family medicine – we suggest that researchers and training programmes should offer research in such a way that registrars will find it an attractive activity in which to participate.

This requires attention to the wishes of registrars and availability of resources,¹¹ and the development of a culture that motivates registrars to prioritise research rather than a culture that views research as 'unnecessary'.¹⁹

The way in which the training environment values research is an important factor in how registrars respond to participation. This environment includes both the training program, training practice and the profession at large. Faculty play an important role in this.7 With their expertise and experience, enthusiastic faculty could successfully integrate research into vocational training.²⁰ Finally, registrars might be motivated by colleague peers, namely registrars actively involved in research such as those with an academic registrar position.21

Creating attractive research programmes should motivate registrars to voluntarily participate in research. Research networks, departments of family medicine, and residency training programmes must collaborate to develop programmes that offer registrars the opportunity to participate in distinct research projects. Whether this increases registrars' participation in research activities is a question for further study.

Acknowledgments

We thank ZonMw, the Netherlands organisation for health research and development, which funded our study, and we kindly thank all the registrars who participated.

References

- 1 De Maeseneer JM, van Driel ML, Green LA, van Weel C. The need for research in primary care. Lancet 2003; 362(9392):1314-1319.
- 2 Clark J, Smith R. BMJ Publishing Group to launch an international campaign to promote academic medicine. BMJ 2003; 327(7422):1001-1002.
- 3 Rosemann T, Szecsenyi J. General practitioners' attitudes towards research in primary care: qualitative results of a cross sectional study. BMC Fam Pract 2004; 5(1):31.
- 4 Mendis K, Solangaarachchi I. PubMed perspective of family medicine research: where does it stand? Fam Pract 2005; 22(5):570-575.
- 5 Sullivan F, Butler C, Cupples M, Kinmonth AL. Primary care research networks in the United Kingdom. BMJ 2007; 334(7603):1093-1094.
- 6 van Weel C, Rosser WW. Improving health care globally: a critical review of the necessity of family medicine research and recommendations to build research capacity. Ann Fam Med 2004; 2 Suppl 2:S5-16.
- 7 Temte JL, Hunter PH, Beasley JW. Factors associated with research interest and activity during family practice residency. Fam Med 1994; 26(2):93-97.
- Hebert RS, Levine RB, Smith CG, Wright SM.
 A systematic review of resident research curricula. Acad Med 2003; 78(1):61-68.
- 9 Dehaven MJ, Wilson GR, Murphree DD, Grundig JP. Family practice residency program directors' views on research. Fam Med 1997; 29(1):33-37.
- 10 Recommended core educational guidelines for family practice residents. Research and scholarly activity. American Academy of Family Physicians. Am Fam Physician 1992; 46(5):1564-1566.
- 11 Dehaven MJ, Wilson GR, O'Connor-Kettlestrings P. Creating a research culture: what we can learn from residencies that are successful in research. Fam Med 1998; 30(7):501-507.

- 12 Smith M. Research in residency: do research curricula impact post-residency practice? Fam Med 2005; 37(5):322-327.
- 13 Peto V, Coulter A, Bond A. Factors affecting general practitioners' recruitment of patients into a prospective study. Fam Pract 1993; 10(2):207-211.
- 14 Mynors-Wallis LM, Gath DH, Day A, Baker F. Randomised controlled trial of problem solving treatment, antidepressant medication, and combined treatment for major depression in primary care.[see comment]. BMJ 2000; 320(7226):26-30.
- 15 Franke L, van Weel-Baumgarten E, Lucassen P, Beek M, Mynors-Wallis L, van Weel C. Feasibility of training in Problem-solving treatment for general practice registrars. Submitted.
- 16 Levinson W, Dull VT, Roter DL, Chaumeton N, Frankel RM. Recruiting physicians for officebased research. Med Care 1998; 36(6):934-937.
- 17 Huibers M, Windt van der D, Boeke J. The participation of general practitioners in scientific research [De deelname van huisartsen aan wetenschappelijk onderzoek]. Huisarts en Wetenschap 2002; 45(9):454-458.
- 18 Segal S, Lloyd T, Houts PS, Stillman PL, Jungas RL, Greer RB, III. The association between students' research involvement in medical school and their postgraduate medical activities. Acad Med 1990; 65(8):530-533.
- 19 Neale AV. A national survey of research requirements for family practice residents and faculty. Fam Med 2002; 34(4):262-267.
- 20 Carek PJ, Araujo D, Nalin PM. Scholarly activity and residency training: seeking strategic partnerships. Ann Fam Med 2005; 3(6):560-561.
- 21 olde Hartman TC, Franke LJ, Borghuis MS, Lucassen PL. Becoming a research-oriented family physician. Fam Med 2005; 37(10):686-687.

5 Problem-solving treatment for emotional problems in primary care – a Cochrane review

Lieke J.A. Hassink-Franke Joe Costa Peter L.B.J. Lucassen Reinier P. Akkermans Laurence M. Mynors-Wallis Chris van Weel Evelyn M. van Weel-Baumgarten

Submitted for publication

Presented at:

Dutch College of General Practitioners, Scientific meeting 2009, Utrecht, the Netherlands NAPCRG Annual Meeting 2009, Montreal, Canada

Abbreviations:

BDI	Beck Depression Inventory
CCDAN-QRS	Cochrane Collaboration Depression, Anxiety and Neurosis
	Review Group - Quality Rating Scale: range 0-42 points
CIS	Clinical Interview Schedule
EPDS	Edinburgh Postnatal Depression Scale
EQ-5D	Euroqol - 5 Dimensions
GHQ-12	12-item General Health Questionnaire
GHQ-28	28-item General Health Questionnaire
GP	General Practitioner
HADS	Hospital Anxiety & Depression Scale
HDRS	Hamilton Depression Rating Scale
HSCL-D	Hopkins Symptom Checklist Depression Scale
	Jadad scale: range 0-5 points
PHQ	Patient Health Questionnaire
PSE	Present State Examination
PSI	Problem Solving Inventory
PST	Problem-solving treatment
PSYCHLOPS	Psychological Outcomes Profiles
RCT	Randomised Controlled Trial
SAS	Social Adjustment Scale
SCID	Structured Clinical Interview for DSM-Disorders
SF-36	Short Form-36
SPSI-R	Social Problem Solving Inventory - Revised
SSI	Social Support Inventory
TicP	Trimbos/iMTA questionnaire for Costs associated with
	Psychiatric IIIness
WAYS	Ways of Coping Questionnaire

5 58-59 WAYS

Abstract

Background

Many patients present to their general practitioner with 'emotional problems'. Emotional problems may include true mental disorders but may also include psychosocial problems or psychological symptoms not severe enough to reach thresholds for a formal diagnosis. Current evidence-based care consists of medication and/or some form of psychological treatment. Evidence for psychological treatment, however, comes largely from studies in the secondary care setting, yet there is considerable demand for psychological treatments within primary care. Problem-solving treatment (PST) is a brief psychological treatment which has shown effectiveness in primary care patients with depression.

Objectives

To assess the effectiveness of PST versus usual care, medication, or other psychological treatments in adult patients with emotional problems in primary care.

Search methods

We searched The Cochrane Collaboration Depression Anxiety and Neurosis group Controlled Trials Register (CCDAN-CTR), Medline, EMBASE, CINAHL, LILACS, PsycInfo, databases of ongoing trials, and reference lists. Date of search was January 2008.

Selection criteria

Randomised controlled trials comparing PST in a primary care setting with any other intervention for adult patients with emotional problems. Data collection and analysis Two reviewers read all abstracts, assessed quality and extracted data independently. The standard mean difference was used to pool continuous data, and odds ratios were used to pool dichotomous data, using a random effects model.

Results

We included twelve trials (2261 participants). Control conditions consisted of usual care (8 studies). antidepressant medication (3 studies), placebo medication (2 studies), or another psychological treatment (5 studies). Four studies were of good quality, five studies of moderate quality and three studies of low quality. Overall, PST was not different from usual care, antidepressants, placebo or other psychological treatments. For patients with major depression, however, two good quality trials showed that PST was more effective than usual care, placebo medication and group psychoeducation. For patients with other emotional problems we found evidence in one moderate quality trial favouring PST above usual care whereas six other trials did not show differences between PST and control treatments. Costs and health care use did not show major differences

Authors' conclusions

We consider PST more effective than control treatments for major depression in primary care. For emotional problems other than major depression however, there was insufficient evidence to show statistically significant differences between PST and usual care, medication, placebo, or other psychological treatments.

Background

Description of the condition

Mental health disorders and other psychological problems are highly prevalent in primary care. The WHO Collaborative Study on Psychological Problems in General Health Care reported a pooled 21% point prevalence of disorders such as major depression, anxiety disorders, somatoform disorders and substance dependence in consecutive attenders of primary care facilities across 14 countries (Ormel 1994). These disorders are the cause of considerable disability. Moreover, these disorders frequently occur as comorbid conditions in primary care (Wittchen 1999). Additionally, psychological symptoms that do not reach thresholds for formal diagnoses are generally excluded from such prevalence estimates despite their prevalence, the accompanying burden to patients, and the resultant increased use of health services (Gureje 2002). Finally, many patients present psychosocial problems to the general practitioner that are not classified into strictly mental categories but are labelled as: problems with work, problems with parents or children and so on. For the purposes of this study, we refer to all of the above disorders and psychological symptoms or psychosocial problems as 'emotional problems'.

Current evidence-based care for patients with emotional problems consists of medication (mainly antidepressants and/or sedatives) and/or some form of psychological treatment. Antidepressant medication, cognitive behaviour therapy and interpersonal psychotherapy have been proven effective in depressive disorders: both antidepressants and cognitive-behaviour therapy are effective in panic disorder; cognitive therapy and some antidepressants are effective in generalized anxiety disorder (Kumar 2002; Geddes 2002; Gale 2002). However, the evidence for the use of psychological treatments comes largely from studies in secondary care. Additionally, psychological interventions are often relatively lengthy (12-16 sessions), scarce, and delivered only by specialists with a specific interest in a particular intervention. Despite the lack of evidence and potential barriers to more wide-spread use, there is considerable patient and physician demand for psychological treatments in primary care (Fritzsche 2002; Oopik 2006). Consequently, there is a need to both identify psychological interventions applicable to the treatment of emotional problems in primary care, as well as to assess the potential effectiveness of their use

Description of the intervention

Problem-solving treatment (PST) is a psychological intervention 'focusing on the here and now' (Mynors-Wallis 2000) and may be a suitable treatment to deal with the broadness of emotional problems seen in primary care. PST teaches patients to use their own skills and resources to improve functioning (Gath 1997). A link is established between emotional symptoms and current everyday problems and it is explained that if patients succeed in solving and gaining control over (part of) their problems, symptoms might improve. The treatment is relatively brief, with the first introductory session usually lasting one hour, and remaining sessions lasting maximum half an hour.

In primary care, PST has a maximum of 6 treatment sessions (Gath 1997) and can be administered by general practitioners, practice nurses, or other primary care workers. PST is thus considered a 'generic treatment' aiming to improve the individual's skills to solve problems encountered in everyday life. It is this generic character that makes PST uniquely applicable to the wide range of problems seen in the primary care setting.

Why it is important to do this review

Three recently published systematic reviews have focused specifically on PST. Bell and D'Zurilla conducted a meta-analysis of controlled outcome studies on the efficacy of PST in reducing depressive symptomatology. PST was found to be as effective as other psychosocial therapies and medication, and more effective than support groups or no treatment (Bell 2009). Similarly, Cuiipers et al in 2007 conducted a meta-analysis of the effectiveness of problem solving therapies for treating depression. Their results showed a mean standardized effect size for PST of 0.83 in the random effects model, with very high heterogeneity (Cuijpers 2007). Because of the high heterogeneity, the authors recommended more research to ascertain the conditions and subjects in which these positive effects may be realized. A third review by Malouff aimed to assess the efficacy of PST across all types of mental or physical health problems (Malouff). This meta-analysis showed that PST is significantly more effective than no treatment (d=1.37), treatment as usual (d=0.54), and attention placebo (d=0.54), but not significantly more

effective than other evidence-based treatments offered as part of a study (d=0.22). While not statistically significant, there was a trend in favour of PST over other treatments in this meta-analysis. Ultimately, none of these three reviews specifically focused on studies performed in primary care and all of them recommended further research.

Three additional Cochrane reviews have addressed the treatment of emotional problems in primary care. Den Boer et al (Den Boer 2005) focused on the effectiveness of any kind of psychological treatment delivered by 'paraprofessionals' in community, primary and secondary care settings for the treatment of anxiety and depressive disorders. They reported positive outcomes for treatments delivered by paraprofessionals as compared to no treatment. In another review, Bower and colleagues (Bower 2001) aimed to assess the effectiveness and cost effectiveness of counseling interventions for patients with psychological and psychosocial problems considered suitable for counseling in primary care. The review showed a modest effectiveness of counseling in short term outcomes (standardised mean difference -0.28, 95% CI -0.43 to -0.13, n = 772). A third review (Huibers 2003) assessed the effectiveness of a variety of psychosocial interventions delivered by general practitioners and concluded that there is little evidence for the use of psychosocial interventions by general practitioners. Of the interventions reviewed, PST for depression seemed the most promising tool for general practitioners, although the reviewers emphasised that stronger evidence is necessary

and effectiveness in routine practice has yet to be demonstrated (Huibers 2003). None of the above three reviews specifically addressed the effectiveness of PST for the broad range of emotional problems seen in primary care.

Responding to the requirement for a stronger evidence-base for PST in primary care, our review focuses on the effectiveness of PST provided by specifically trained general practitioners, nurses or other primary care health workers in patients with the broad range of emotional problems typically seen in the primary care setting.

Objectives

We aim to assess the effectiveness of PST (delivered by general practitioners, practice nurses or other primary care workers) versus any other treatment in adult patients with emotional problems in primary care.

Methods

Criteria for considering studies for this review

Types of studies

We included randomised controlled trials on PST versus any other treatment. Non-randomised studies were excluded as they tend to be biased in unpredictable directions (Kunz 1998). Studies with both PST delivered to individuals and PST delivered in a group format were included. Studies on group treatment were included only if they used cluster randomization in their design.

Types of participants

The review was intended to be broad

with respect to the diversity of problems and was not confined to patients with DSM-IV diagnoses such as major depression or anxiety disorder. We also aimed to include patients with 'below threshold disorders'. minor depression, dysthymia and psychosocial problems. Therefore, all studies including adult patients with 'emotional problems' treated in a primary care setting were eligible for the review. We defined 'emotional problems' as 1) major depression according to DSM-IIIR/DSM-IV or ICD-9/ICD-10 classification, 2) anxiety disorders according to DSM-IIIR/DSM-IV or ICD-9/ICD-10 classification, 3) minor depression according to the DSM-IV classification, 4) dysthymia according to the DSM-IV classification, 5) psychological symptoms not reaching thresholds for classification systems, 6) psychosocial problems: problems with work, problems with relations, problems with (the acceptance of) disease, or 7) somatoform symptoms.

Types of interventions

This review included studies using individualized or group-based PST as the experimental condition. PST consists of a maximum of 6 sessions with 7 stages (Mynors-Wallis 2005). The stages are: explaining the rationale; clarification of the problems; establishing achievable goals; generating solutions; selecting and implementing preferred solution; and evaluation. There was no restriction concerning the nature of control conditions. To be included in the review the article had to state that the therapists (general practitioner, psychologist, nurse or social worker) were trained in problem solving

5 62-63

techniques. Patients were required to have had at least three treatment sessions.

Types of outcome measures

Primary outcomes

Given the broad range of problems and diversity of study designs, numerous outcome measures were used. We considered symptom severity (as measured with symptom scales) and recovery rates as primary outcome measures.

Secondary outcomes

As secondary outcomes we included quality of life measures and cost effectiveness. With respect to costeffectiveness, we measured days off work and financial costs. Adverse effects such as deterioration of symptom scores, increased use of psychotropic medication, or more days off work were analysed separately.

Search methods for identification of studies

Electronic searches

In order to identify studies for review, we searched the following databases (search date August 2009 - no language restriction):

1) The Cochrane Collaboration Depression Anxiety and Neurosis group Controlled Trials Register (CCDAN-CTR). This database contains records on trials comparing treatments within the scope of the CCDAN, including mood disorders, anxiety disorders, somatoform disorders, dissociative disorders, and eating disorders. The CCDAN-CTR also includes clinically significant problems, such as deliberate self-harm and suicide attempt, which are often associated with these disorders. The register is updated quarterly and includes results from searches of CINAHL, EMBASE, LILACS, MEDLINE, NRR, PSYCLIT, PSYCINFO, PSYNDEX and SIGLE. Additional quarterly screening of conference proceedings as well as hand searching of relevant journals are also conducted by the CCDAN. We also separately searched 2) MEDLINE, 3) EMBASE, 4) CINAHL, 5) PsycINFO, 6) NRR, 7) Lilacs, and 7) Psyndex.

Search strategy:

We used the search strategy as modified by CCDAN (www.iop.kcl.ac.uk/ iop/ccdan.index.htm). The syntaxes for the other databases were modified according to the properties of those databases.

The syntaxes for CCDAN-CTR, MEDLINE, EMBASE, CINAHL and PsycINFO can be found in the Appendices.

Searching other resources

Additionally searches were conducted by checking reference lists of retrieved publications as well as other reviews on PST. Finally we identified unpublished studies by personal communication with experts in the field and researchers currently performing trials on PST.

Data collection and analysis

Selection of studies

Abstracts of records retrieved by the electronic searching process were assessed for inclusion in the review independently by two reviewers (LH and JC) (Edwards 2002). If necessary the full article was read. Disagreements were resolved by discussion, or via mediation by a third reviewer (PL) when necessary

Data extraction and management

Data extraction of selected trials was performed by two reviewers independently on pre-coded electronic forms. Disagreements were resolved by discussion. The following data were extracted:

- General information: published/ unpublished, title, authors, source, country, contact address, language of publication, year of publication, duplicate publication, sponsoring,
- 2. Trial characteristics: design, duration, randomisation method, allocation concealment, blinding of outcome assessors.
- 3. Interventions: integrity of PST (patient attended three or more sessions), number of PST sessions, qualification of providers (general practitioner, nurse, social worker), adequacy of provider training (provider received practical skills training in PST), number of providers, type of control intervention
- 4. Participants: sampling (convenience, random), inclusion and exclusion criteria, number of participants in intervention and control group at baseline, gender, age, baseline characteristics, duration of the emotional problem, definition of the emotional problem, withdrawals/ losses to follow up, subgroups,

5. Outcomes, adverse effects. The primary outcomes were changes in symptom scores as measured with any validated scale and recovery rates. Secondary outcomes were quality of life measures, resolution of psychosocial outcomes such as work related or relational problems and cost effectiveness. Finally, we recorded adverse events such as deterioration of symptom scores and adverse effects of medication.

We attempted to include any necessary missing data by directly contacting investigators.

Assessment of risk of bias in included studies

Methodological quality

In order to determine study quality we used the validated 5-item Jadad scale (Jadad 1996) and the 23-item CCDAN Quality Rating Scale as a fieldspecific quality rating scale (ORS, see Table 1) (Moncrieff 2001). The Jadad scale contains items on randomization method, allocation concealment, blinding and completeness of followup. There is empirical evidence that allocation concealment, generation of allocation sequences, blinding and completeness of data is associated with trial results (Juni 2001). The scale has a range from 0 to 5, with higher scores indicating good quality. The ORS includes 23 items on sample size, allocation, use of diagnostic criteria, compliance, attrition and statistical analysis. Each item at the ORS is scored 0, 1 or 2, which gives a total score range of 0 to 46. Methodological quality was scored independently by two reviewers.

Disagreements were resolved by discussion or by consulting a third reviewer when discussion did not result in consensus.

We considered generation of allocation sequences adequate if the resulting sequences are unpredictable. Unpredictability can be achieved by one of the following methods: computer generated random numbers, table of random numbers, drawing lots or envelopes, coin tossing, shuffling cards or throwing dice. Inadequate methods are: sequences resulting from case record numbers, date of birth, date of admission or alternation (Juni 2001). We considered concealment of allocation sequences adequate 'if patients and enrolling investigators cannot foresee assignment': central randomisation (performed at a site remote from trial location) or sequentially numbered, sealed, opaque envelopes (Juni 2001). We included trials with inadequate allocation concealment in the meta-analysis, but planned to perform sensitivity analyses to study the influence of inadequate allocation concealment on outcomes. Blinding of patients and care providers is not possible in studies on psychological treatments. The only possible form of blinding in trials on PST is blinding of outcome assessors. which was recorded.

Patients lost to follow-up after allocation to treatment groups may lead to attrition bias. Adequate handling of attrition bias presupposes analyses according to the intention to treat principle. Correct handling of losses to follow-up requires the application of the last observation carried forward principle (LOCF). Trials with > 20% lost to follow-up were considered inadequate. Trials with > 10% difference in dropout rate between experimental and control group were considered inadequate as well. We planned to further examine the influence of quality on the results of the meta-analysis with sensitivity analyses (see Statistical analysis).

Measures of treatment effect

Continuous data were analysed if (i) means and standard deviations were available and (ii) there was no clear evidence of a skewed distribution i.e. if the data were normally distributed according to the original publication. As a measure of treatment effect for continuous data we used weighted mean differences (WMD). If data was skewed, measures were dichotomised according to the median value. Where different instruments or scales were used to measure the same clinical outcome, standardised mean differences (SMD) were calculated and combined across studies

We used the differences from baseline to endpoint as the actual measure of effect of all continuous variables. The standard deviations of these differences are essential for the data to be included in the meta-analysis. If the standard deviation (SD) of the difference was not reported, we calculated the SD of the difference with the following formula: SD_{pareddifference} = $\sqrt{[(SD1)^2 + (SD2)^2 - 2 \times r \times SD1 \times SD2]}$. SD_{paireddifference} = standard deviation of the difference (pre-/post-treatment), SD1 = Standard deviation of the pretreatment value, SD2 = Standard deviation of the post-treatment value, r = correlation coefficient between pre/ post treatment. We used a conservative correlation coefficient of 0.4. Dichotomous data were analysed by calculation of the relative risk with the

95% confidence interval. If applicable we used NNT with 95% confidence intervals.

For the dichotomous and the continuous outcomes, a random effects model was used to allow for the expected heterogeneity. In pooling the data, we weighted the effect estimates by the inverse of their variance. Thus, the larger the trial the greater the weight the study received in the metaanalysis. We considered pooling not appropriate when the data were too heterogeneous as indicated by the l² test and by inspection of the clinical characteristics.

Unit of analysis issues

There were no unit of analysis issues because all trials randomised participants individually.

Dealing with missing data

If outcome data had remained missing despite our attempts to obtain complete outcome data from authors, we performed an available-case analysis, based on the numbers of patients for whom outcome data were known.

Assessment of heterogeneity

Consistency of results was assessed visually and by examining I², a quantity which describes approximately the proportion of variation in point estimates that is due to heterogeneity rather than sampling error (Higgins 2002).

Assessment of reporting biases

As an indication for publication bias we examined funnel plots of effect size against study precision. (Egger 1997).

Data synthesis

If possible, and where appropriate, overall effects were calculated, using Cochrane Collaboration software (current version: RevMan 5).

Subgroup analysis and investigation of heterogeneity

We intended to separately analyse the effects of the intervention in the various emotional problems. We considered the following subgroups particularly relevant: major depression, anxiety disorders, dysthymia, somatoform symptoms and psychological problems not reaching thresholds for classification systems. Also, we intended to separately analyse the effects of the intervention by type of therapists, for instance nurses versus GPs.

Sensitivity analysis

Sensitivity analyses were conducted to determine the impact of study quality on outcome. The sensitivity of the results for good versus low quality was determined by subgroup analyses to explore the influence of the quality of the studies.

Interpretation of Results

We based our conclusions about effectiveness on the differences in psychological outcomes and quality of life outcomes. For this purpose we considered primarily the statistically significant results. For further refinement we discussed the clinical importance of the effects. Statistical significance was based on the 95% confidence intervals. Clinical importance was based on the 'minimal clinically important difference' (MCID) which we defined as a change of 15% or more on dimensional scales of primary outcomes (symptom severity, quality of life) or as rate of recovery for specific syndromes. We therefore regarded numbers of recoveries as clinically relevant when the difference between PST and control group was 15% or more. For continuous outcomes, we looked at the extent of clinical importance by considering the relationship of the MCID with the point estimate and the 95% confidence interval (CI) surrounding it (Man-Son-Hing 2002):

- 1. definite, when the MCID is smaller than the lower limit of the 95% Cl;
- 2. probable, when the MCID is greater than the lower limit of the 95% CI, but smaller than the point estimate of the efficacy of the intervention;
- 3. possible, when the MCID is less than the upper limit of the 95% CI, but greater than the point estimate of the efficacy of the intervention;
- 4. definitely not, when the MCID is greater than the upper limit of the 95% Cl.

With regard to SMDs, we considered 0.5 and higher as the value for the MCID.

Results

Description of studies

Results of the search

The searches of the databases yielded a total of 2403 abstracts: 170 in the CCDAN-controlled trial register, 536 references in MEDLINE, 773 in CINAHL, 621 in EMBASE, 161 in PsycINFO, 4 in Lilacs, 127 in Psyndex and 11 in the National Research Register. Of these 2304 abstracts, we studied, after removing duplicates, the full-text version of 130 articles for definite judgment on inclusion. We contacted 13 authors for additional information.

Included studies

Finally, 12 studies met eligibility criteria and were included in the review (Barrett 2000; Catalan 1991; Dowrick 2000; Kendrick 2005; Liu 2007; Lynch 1997; Lynch 2004; Mynors-Wallis 1995; Mynors-Wallis 1997; Mynors-Wallis 2000; Schreuders 2007; Tezel 2006). The studies by Williams and Barrett both belonged to the Treatment Effectiveness Project and were therefore described as one study.

Patients, therapists and control treatments

The twelve studies included 2261 subjects with 803 in the PST condition and 1458 in the control conditions. Participants were recruited from the community or in general practice during the consultation or in the waiting room. Most patients were between 18 and 65 years old; a minority were elderly people. Two studies included patients with major depression (Mynors-Wallis 1995; Mynors-Wallis 2000); three studies included patients with dysthymia, mild or minor depression (Barrett 2000 ; Lynch 1997; Lynch 2004); six studies included patients with a range of mental health problems, including depressed mood, anxiety, tension, or reaction to life difficulties (Catalan 1991; Dowrick 2000; Kendrick 2005; Liu 2007; Mynors-Wallis 1997; Schreuders 2007); and

one study included only women who were postpartum (Tezel 2006). Most studies used 'care as usual' as the control condition. Other studies used antidepressant medication, placebo medication or other psychological treatments All control treatments were applied individually except the group psychoeducation in one study (Dowrick 2000). PST was generally applied as an individual treatment consisting of a maximum of 6 sessions with 7 stages. In 2 studies PST was applied by telephone (Lynch 1997; Lynch 2004). PST was mostly delivered by nurses; some studies used research GPs, psychiatrists, or psychologists. In all but two studies (Catalan 1991: Tezel 2006) it was clear that therapists were trained in PST. In three studies the adherence to the treatment protocol was checked by audio/videotaping (Dowrick 2000; Kendrick 2005; Schreuders 2007). In six studies there was supervision in order to optimize treatment quality. The number of sessions in PST and control interventions varied between 2 and 8. Three studies did not report the attendance rates (Lynch 1997; Lynch 2004; Tezel 2006). In the other studies, more than 2/3 of the patients had had 4 or more sessions. The only study with less than 3 sessions, which is considered as the minimum number needed for treatment, was the study by Liu with 2.3 sessions on average. Follow-up varied from 6 to 52 weeks. For the meta-analyses, we categorised four trials with a duration of 6-16 weeks for practical reasons in the 3-month follow-up category (Liu 2007; Lynch 1997; Lynch 2004; Tezel 2006) and one trial of 9-months follow-up in the 12-month follow-up (Schreuders 2007).

Excluded studies

Other studies were excluded from the review (see excluded studies table), mainly because they were not performed in primary care or did not investigate PST. We excluded studies in which PST was part of a (stepped) care programme, e.g. the IMPACT study and the PEARLS study (Unutzer 2001: Ciechanowski 2004) because we could not calculate PST effects separately. The included studies table describes the characteristics of the trials, including the characteristics of participants, the interventions, outcome measures, and methodological quality. We did not find additional references via reference checking.

Risk of bias in included studies

Overall methodological quality

All data related to quality assessment and outcome was extracted by two review authors, who then reached consensus on final ratings. Two studies scored 1 point on the 5-point Jadad scale, three studies 2 points, and seven studies 3 points (mean 2.4, SD 0.8). All studies scored zero points at the items 'described as double blind' and 'blindness adequately described'. The overall ORS mean score for included studies was 29.8 (SD 7.9), ranging from 17 (Lynch 1997; Tezel 2006) to 38 (Mynors-Wallis 2000), see Table 2 and Table 3. Four trials were of high quality (highest quartile of the QRS), five were of moderate quality and three of low quality. Most trials scored low on blinding of subjects and assessors, sample size, duration of follow-up, and intention-to-treat analysis.

Allocation

Randomisation and allocation concealment

Although all studies described the allocation of participants to groups as 'randomised', Lynch 1997, Lynch 2004 and Tezel did not specify their methods for generating the random sequence. Eight studies adequately concealed allocation of patients and three studies did not describe the method of allocation (Lynch 1997; Lynch 2004; Tezel 2006). Catalan used drawing cards from a random pack to assign participants to groups but it was not clear whether allocation was concealed from the investigators and therefore considered inadequate.

Blinding

Blinding of treatment

All studies scored zero points at 'blinding of subjects' because in psychological treatments it is not possible to keep patients and therapists blinded. Trials investigating PST versus antidepressant medication and placebo medication could keep medication blinded; PST, however, not. Therefore, we scored all trials zero points.

Blinding of outcome measurement

The assessor was blinded in six studies (Barrett 2000; Catalan 1991; Liu 2007; Lynch 1997; Mynors-Wallis 1995; Mynors-Wallis 2000). Blinding was, however, not tested. Lynch 2004 did not report any blinding at all. In five studies this item was not applicable because of self rating outcome measures; these were rated as zero points.

Incomplete outcome data

Losses to follow up

We separated the attrition rate in 'non-compliance of patients with treatment' (non-completers, i.e. those who withdrew before the end of treatment) and 'lost to follow-up from study assessments' (those with incomplete follow-up data, i.e. until the very last follow-up of the regarding study). Non-compliance ranged from 0% to 33% (average 20%). Lost to follow-up numbers ranged from 0% to 45% (mean 27%, median 28%), with Lynch 1997 and Lynch 2004 being exceptionally high with 43% respectively 45%. Tezel did not report data on non-compliance; Barrett and Catalan did not report data on lost to follow-up. Reasons of non-completion were only given in four studies (Barrett 2000; Catalan 1991; Mynors-Wallis 1995; Mynors-Wallis 2000).

Effects of interventions

Overview of results

We calculated effect sizes for 158 comparisons in ten studies. The studies of Lynch 1997 and Lynch 2004 did not provide sufficient data to do so. The statistically significant results concerned comparisons on main outcome (24), cost-effectiveness (9) and patient satisfaction (1). All other comparisons (124) yielded effect sizes that were not statistically significant. Twenty-four comparisons on main outcomes showed statistically significant differences in outcome between PST and control treatment. Of these, nine were in favour of the control treatment and fifteen in favour of PST. The majority (13) of the statistically significant differences in favour of PST

resulted from two studies (Mynors-Wallis 1995: Dowrick 2000). In these studies PST was effective compared with placebo medication at 3-month follow-up, compared with usual care at 6- and 12-month follow-up and compared with group psychoeducation at 12-month follow-up. The nine statistically significant differences not in favour of PST resulted from 3 studies: antidepressants were more effective than PST for dysthymic or minor depressed patients, postnatal care was more effective than PST in women at risk for postpartum depression, and at one outcome generic mental health nurse care was more effective than PST for emotional symptoms

The seven statistically significant effects of PST resulting from *cost analyses* demonstrated less days off, less consultations, less consultation costs and less medication costs but higher total treatment costs in the PST group (Kendrick 2005; Liu 2007; Mynors-Wallis 1997).

The significant difference resulting from measurement of patient satisfaction was in favour of PST as compared with usual care. We performed 49 meta-analyses on effect sizes of 10 studies. Forty-six meta-analyses did not vield significant differences between PST and control condition. The three meta-analyses that did vield a significant difference favoured PST above usual care in quality of life outcomes at 6-month follow-up and favoured antidepressant medication above PST in psychological and quality of life outcomes at 3-month follow-up. Heterogeneity between studies was large (I² varying from 0% to 91%).

Detailed description of results

PST versus usual care

3-month follow-up

Seven trials compared PST and usual care at 3-month follow-up (694 patients: 65 with minor depression/ dysthymia; 629 with emotional symptoms). In individual studies we found statistically significant differences in 1 of 22 comparisons (with 13 different outcome measures in total). This difference was in favour of PST with a magnitude of 3 points on a scale ranging from 0-240 (Catalan 1991).

Trial data of Lynch 1997 and Lynch 2004 of patients with minor depression were not detailed enough to use the meta-analysis. In the first trial, BDI scores after treatment were significantly better in the PST group than in the usual care group. The second trial did not find any statistically significant difference in outcomes between PST and usual care. We performed 10 meta-analyses: 6 with psychological symptom scores and 4 with quality of life scores. None of these vielded a statistically significant difference between PST and usual care. So, there is no evidence that PST was significantly more effective than usual care at 3-month follow-up.

6-month follow-up

Four trials compared PST and usual care at 6-month follow-up (602 patients: 317 with major depression; 285 with emotional symptoms). In individual studies we found 6 statistically significant differences in 17 comparisons (with 12 different outcome measures in total). These differences came from two studies and were all in favour of PST. Catalan found a statistically significant WMD at the PSE of -4.70 (95% CI -8.35 to -1.05, n=47) in favour of PST. Dowrick's study, with mostly major depressed patients, found that PST patients had statistically significant better BDI scores (63-point scale) and SF-36 scores (mental role, social function, and mental health; 100-point scales). More PST patients than usual care patients recovered at the BDI (RR 2.54, 95% CI 1.85 to 3.50, n=270).

We performed 8 meta-analyses. The meta-analyses of psychological symptom scores did not show statistically significant differences. The meta-analysis of 3 trials measuring quality of life showed a statistically significant advantage for PST patients (SMD 0.33, 95% Cl 0.15 to 0.51, n=497).

So, at 6-month follow-up, quality of life was significantly better for patients who received PST than for those who received usual care. Overall, there was no difference in effect on psychological symptoms.

12-month follow-up

Two trials compared PST and usual care at 12-month follow-up (492 patients: 317 with major depression; 175 with emotional symptoms). In individual studies we found 1 statistically significant difference in 9 comparisons (with 7 different outcome measures in total). Depressed patients had statistically significant better scores at the 100-point SF-36 social function scale after PST than after usual care (WMD 14.01, 95% CI 5.51 to 22.51, n=218).

We performed 3 meta-analyses. SF-36 mental health scores did not differ significantly between groups. Also, psychological outcomes and quality of life outcomes did not differ significantly between PST and usual care (SMD -0.11, 95% Cl -0.36 to 0.14, n=345 respectively 0.15, 95% Cl -0.06 to 0.36, n=345).

So, after 12 months, there were no differences between PST and usual care.

In conclusion

At 3-month follow-up there were no statistically significant differences between PST and usual care. At 6-month follow-up, PST patients' quality of life was significantly better than usual care patients' quality of life and patients with major depression had higher chances of recovery after PST than after usual care. At 12-month follow-up, however, we did not find statistically significant overall differences in quality of life nor psychological outcomes.

PST versus antidepressant medication

3-month follow-up

Three trials compared PST and antidepressant medication at 3-month follow-up (571 patients: 136 with major depression; 435 with minor depression/dvsthvmia). Medication consisted of an SSRI in two studies and amitriptyline in one study. In individual studies we found 4 statistically significant differences in 26 comparisons (with 8 different outcome measures in total). For 18-59 year-old patients with dysthymia or minor depression symptom resolution in the first 2 weeks was more rapid in the SSRI group than in the PST group but in week 2-11 this difference disappeared. HSCL-D scores improved significantly
more in the SSRI aroup. For dysthymic patients of 60 years and older HSCL-D scores improved significantly more too in the SSRI group. For minor depressed or dysthymic patients of 60 years and older HDRS scores improved significantly more in the SSRI group than in the PST group. The trial with major depressed patients did not show factors predictive for which patients might benefit from either PST or amitriptyline (Mynors-Wallis 1995). We performed 11 meta-analyses. Two showed significant differences in favour of antidepressant medication: psychological symptom scores (SMD 0.23, 95% CI 0.01 to 0.44 n=461) and the change in SF-36 Mental health score (WMD -2.70, 95% CI -5.20 to -0.19, n=329). Recovery rates did not differ significantly between groups: RR 0.98, 95% CI 0.83 to 1.17. n=564. The other 8 meta-analyses were not significantly different neither. Sensitivity analyses on tricyclic antidepressants and SSRI's did not alter these findings.

6-month follow-up

At 6-month follow-up, one trial compared PST with an SSRI for 286 patients with minor depression or dysthymia and found no statistically significant difference in the numbers of recovered patients at the HDRS (RR 0.90, 95% CI 0.65 to 1.26, n=286).

12-month follow-up

One trial compared PST versus an SSRI for 75 patients with major depression at 12-month follow-up (5 comparisons, 4 outcomes). PST did not lead to statistically significant more recovered cases than medication (RR 1.11, 95% CI 0.76 to 1.63, n=75). Other psychological outcomes were not significantly different. Quality of life outcomes were not reported. Metaanalyses were not possible.

In conclusion

There is some evidence that antidepressant medication is more effective for dysthymic or minor depressed patients than PST at the short-term. At the long-term there are no differences between PST and antidepressant medication for patients with dysthymia or major or minor depression.

PST versus placebo medication

3-month follow-up

Two trials reported outcomes at 3-month follow-up (499 patients: 60 with major depression; 439 with minor depression/dysthymia). In individual studies we found 7 statistically significant differences in 21 comparisons (with 7 different outcome measures in total). Major depressed patients in the PST group had better scores at the PSE, SAS, BDI and HDRS and they had higher recovery rates than patients in the placebo group (RR 2.25, 95% Cl 1.16 to 4.36, n=60). Dysthymic patients of 60 years and older had better SF-36 Mental health scores after placebo medication than after PST. Also, symptom resolution for patients aged 18-59 was quicker from week 0 to 2 at placebo medication. This difference disappeared from week 2 to 11.

We performed 6 meta-analyses. The proportion of recovered patients nor the psychological symptom scores and quality of life scores differed significantly between treatment groups (RR 1.06, 95% CI 0.77 to 1.46, n=505). Sensitivity analyses on age nor diagnosis showed statistically significant differences.

6-month follow-up

One trial reported recovery rates at 6-month follow-up (290 patients with dysthymia or minor depression). There were no statistically significant differences between PST and placebo for dysthymia nor minor depression (RR 1.02, 95% CI 0.75 to 1.39, n=290).

In conclusion

Overall, there were no statistically significant differences in outcome between PST and placebo medication. In the study investigating patients with major depression, PST was more effective than placebo medication.

PST versus other psychological treatment

3-month follow-up

Four trials reported outcomes at 3-month follow-up (436 patients: 98 with minor depression/dysthymia: 338 with emotional symptoms). In 3 trials, we found 2 statistically significant differences in 12 comparisons (with 10 different outcome measures in total). Both differences came from one trial, investigating PST versus postnatal care in women at risk for postpartum depression. The trial showed statistically significant more recoveries and a significant better improvement at the BDI in the postnatal care group than in the PST group (RR 0.49, 95%) CI 0.31 to 0.76, n=62; WMD 6.30, 95% CI 2.84 to 9.76, n=62). We performed 3 meta-analyses. Psychological symptom scores did not show statistically significant differences for emotional symptoms nor overall (overall SMD 0.22, 95% CI

-0.31 to 0.74, n=335). Quality of life outcomes neither differed significantly (SMD 0.12, 95% CI -0.11 to 0.36, n=273).

So, there is no evidence for statistically significant differences between PST and other psychological treatments. In the single study investigating PST in women at risk for postpartum depression, there was evidence that PST gave less improvement than postnatal care. In the other 3 trials, there were no differences in outcomes between PST or any other psychological treatment.

6-month follow-up

Two trials reported outcomes at 6-month follow-up. One regarded depressed patients (n=236, mainly major depression) and compared PST with group psychoeducation for depression. The other study regarded emotional symptoms (n=169) and compared PST with generic mental health nurse care. In the individual studies we found 1 statistically significant difference in 11 comparisons (with 10 different outcome measures in total): in the study with patients with emotional symptoms, Kendrick found statistically significant more improvement of 4 points on the 57-point CIS for generic mental health nurse care than for patients who received PST (WMD 4.00, 95% CI 0.12 to 7.88, n=133). We performed 2 meta-analyses. Psychological outcomes did not differ significantly between PST and control

significantly between PST and control groups (SMD 0.04, 95% CI -0.53 to 0.62, n=305). Neither existed statistically significant differences in quality of life scores (SMD 0.11, 95% CI -0.11 to 0.34, n=305). So, at 6-month follow-up there were no statistically significant differences between PST and general mental health nurse care or group psychoeducation.

12-month follow-up

One study reported 12-month followup and included 236 depressed patients mainly suffering from major depression. It compared PST with group psychoeducation for depression. Three of 5 comparisons (with 4 different outcome measures in total) were significantly in favour of PST: 4 points on the 63-point BDI (WMD -4.15, 95% CI -7.01 to -1.29, n=172); 8 points on the 100-point SF-36 mental health score (WMD 7.69, 95% CI 1.25 to 14.13, n=172) and 14 points on the 100-point SF-36 mental role (WMD 13.61, 95% CI 0.97 to 26.25, n=172). The proportion of recoveries did not differ significantly between the PST group and the psychoeducation group (RR 1.17, 95% CI 0.90 to 1.51, n=172).

So, at 12-month follow-up PST for patients with major depression was more effective than group psychoeducation for depression, mainly in terms of quality of life.

In conclusion

Overall, short-term outcomes after PST do not significantly differ from outcomes after other psychological treatments. PST might be less effective than postnatal care for women at risk for postpartum depression. The single trial with long-term outcomes, for patients with major depression, favours PST above group psychoeducation.

Costs and Health care use

Of the twelve studies in this review, five reported data on costs or health care use. Of these, three trials (Kendrick 2005: Mynors-Wallis 1997: Schreuders 2007) performed a formal cost analysis in terms of health care use, costs of disability days and costs of treatment, one study (Liu 2007) reported the number of GP consultations at 3-months follow-up and one study (Catalan 1991) briefly described health care use regarding GP consultations, prescription of psychotropic medication and referral to specialist psychiatric treatment. We found 9 statistically significant differences between PST and control treatment in 28 comparisons (with 7 different outcome measures in total). We performed 5 meta-analyses.

Disability days

In the three trials reporting numbers of disability days and costs there was 1 statistically significant difference. Mynors-Wallis found a statistically significant lower number of 12 disability days at 6-month follow-up in the PST group than in the usual care group (WMD -11.88, 95% CI -20.99 to -2.77, n=70). The meta-analysis on total number of days off did not differ significantly between PST and usual care.

Health care use and costs

Four trials compared numbers of GPconsultations during PST and during usual care. Kendrick found significantly less GP-consultations and hence lower consultation costs in PST than usual care patients. Mynors-Wallis found significantly lower medication costs for PST than for usual care patients. Mynors-Wallis and Kendrick, however, found higher total costs for PST than for usual care. Costs of hospital admissions did not differ significantly. The four meta-analyses did not show

5 **74 - 75**

significant differences.

Two trials compared numbers of GPconsultations during PST and during other psychological treatments. Both found significantly lower consultations rates, and hence costs, in the PST than the control treatment group. However, including the numbers of treatment sessions - additional to the GP-consultations - total costs between PST and other psychological treatment did not differ. Costs of medication and hospital admissions were not significantly different between groups either.

In conclusion

There are no clinically or socially relevant differences between PST and control treatments with regard to the numbers and costs of disability days, nor total health care use and costs.

Satisfaction outcomes

Four trials reported outcomes on patient satisfaction about treatment (Catalan 1991; Kendrick 2005; Mynors-Wallis 1997; Mynors-Wallis 2000). The 3 trials comparing PST with usual care all reported that patients experienced PST as more helpful (Kendrick 2005; Mynors-Wallis 1997), or that patients perceived PST as more positive (Catalan 1991) or that patients were more likely to recommend PST to a friend (Kendrick 2005; Mynors-Wallis 1997). In the trial comparing PST with antidepressant medication, PST did not result in higher patient satisfaction nor in a quicker or greater resolution of the patient's perception of the severity of their problems. Neither did PST result in a quicker or greater sense of mastery and self-control as rated by patients (Mynors-Wallis 2000).

In conclusion

Patients were more satisfied after PST than after usual care or after antidepressant medication. Satisfaction levels of patients who received generic mental health nurse care were comparable with those of PST patients.

Clinical importance of the outcome effects

For the continuous psychological outcomes and quality of life outcomes. we also looked at the clinical importance besides the statistical significance according to the method of Man-Son-Hing. We found 21 statistically significant differences in 111 comparisons. Of the 21 statistically significant differences 13 were in favour of PST: 1 was definitely clinically important. 11 were probably important and one was possibly important. Definitely clinically important was the SF-36 social function score compared to usual care at 6-month follow-up (Dowrick 2000). With regard to the numbers of recoveries, we found 3 statistically significant differences in 21 comparisons. Two were clinically relevant in favour of PST compared to usual care at 6-month follow-up (Dowrick 2000) and compared to placebo medication at 3-month followup (Mynors-Wallis 1995). The third was relevantly in favour of postpartum nurse care (Tezel 2006).

Of the 90 non-statistically significant differences 51 were in favour of PST: 5 were probably important and 46 were possibly important. Twenty differences were definitely not clinically important.

Heterogeneity

Heterogeneity was high. Firstly,

there were several types of patients included, varying from women at risk for postpartum depression to patients with anxiety, sleeping problems, psychosocial problems, dysthymia, minor or major depression. Secondly, types of comparison group varied from usual care to antidepressants, placebo, and various other psychological treatments. Thirdly, the length of follow-up varied from 6 weeks to 52 weeks with different follow-up moments between baseline and endpoint and many different outcome measures. Finally, the types of therapists varied from nurses to GPs, psychiatrists, psychologists and students. I² ranged from 0% to 91%. Because of this high heterogeneity we described studies individually too.

Subgroup analyses

We intended to conduct subgroup analyses of the effects of the intervention on the various emotional problems, however, all subgroup data were derived from post-hoc analyses, and we therefore chose not to carry out these analyses. Data limitations also prevented us from performing subgroup analyses by type of therapists. According to the 'eye-ball test' we did not discover relevant differences in outcome between PST by nurses (7 studies), GPs (2 studies), psychiatrists (2 studies), psychologists (1 study), and (medical/psychology) students (1 study).

Sensitivity analyses

We performed sensitivity analyses on the following items: study quality, allocation concealment and integrity of treatment. With regard to the first we repeated our analyses but omitted the three low quality studies (Lynch 1997; Lynch 2004; Tezel 2006). These studies scored worst at losses to follow-up. With regard to allocation concealment we repeated our analyses without these three studies and the study of Catalan because the adequacy of allocation concealment in these four studies was not clear. With regard to integrity of treatment, we repeated the analyses without Liu's study because the average delivered number of sessions in this study was less than 3. None of the sensitivity analyses altered our conclusions.

Publication bias

We had too few studies to create reliable funnel plots for examining publication bias.

Discussion

Summary of main results

In our review we included 12 studies with 2261 patients. Two studies did not provide sufficient data to calculate differences between treatment groups. Three studies showed statistically significant differences favouring PST. three showed significant differences favouring control treatment and six did not show significant differences between PST and control treatments. For patients with major depression we found positive effects favouring PST over placebo medication (Mynors-Wallis 1995). For patients with emotional symptoms PST was favourable over usual care or group psychoeducation (Catalan 1991; Dowrick 2000). In Dowricks study, 71% of participants with emotional symptoms had major depression. The positive evidence came from two trials of high quality (Dowrick 2000: Mynors-Wallis 1995) and one trial of moderate quality (Catalan

1991). PST affected quality of life more than psychological outcomes, which is consistent with the premise that PST affects quality of life improvements through addressing daily problems. The type of therapist does not seem to determine the outcome in these trials as Dowrick had nurses and Catalan and Mynors-Wallis had GPs or a psychiatrist as PST therapists. Two trials favoured control treatment, one of high quality (Barrett 2000) and one of low quality (Tezel 2006): we found positive effects favouring (placebo) medication above PST in dysthymic or minor depressed patients (Barrett 2000); and for women at risk for postpartum depression postnatal care was favourable above PST (Tezel 2006). One high quality trial showed only one significant favourable effect for mental health nurse care above PST for patients with emotional symptoms (Kendrick 2005). Overall, we found few statistically significant differences between PST and usual care, (placebo) medication or other psychological treatments for patients with emotional symptoms. However, when looking at the clinical relevance of the non-significant differences, we cannot rule out the existence of positive effects in favour of PST.

Overall completeness and applicability of evidence

Although we included only twelve studies (2261 participants in total), a broad range of patients were represented, varying from emotional symptoms, to at risk for post partum depression, to minor and major depression. We included trials using several types of control interventions varying from usual care, (placebo) medication and other psychological treatments. Also, all studies used valid and reliable outcome measures that are relevant for primary care. Ultimately, we consider the external validity of the evidence to be *good*.

Quality of the evidence

The mean score on the Jadad scale was 2.4 (range 1-3) and on the ORS 29.8 (range 17-38). According to the ORS, four studies were of high quality. five of moderate quality and three of low quality. There was no clear relation between quality and outcome because both low and good quality trials pointed both in positive and neutral directions for PST, which was confirmed by our sensitivity analyses on high/moderate versus low quality. The evidence in our review had some limitations. Firstly. most sample sizes were small. This led to many wide confidence intervals in the differences between outcomes which made interpretation of clinical relevance often inconclusive because many significant and non-significant differences in outcomes were 'possibly' or 'probably' clinically important. Secondly, there was only one study with 'time and attention' as control condition (Mynors-Wallis 1995). The lack of 'time and attention' control aroups in the other studies might have caused an overestimation of the effect of PST. From the five studies with a psychological treatment as control group (Dowrick 2000; Kendrick 2005; Liu 2007; Lynch 2004; Tezel 2006), in which time and we presume attention was provided too, only Dowrick showed relevant differences in favour of PST. We, therefore, cannot exclude the possibility that structural time and attention is more important than the specific type of intervention provided. Thirdly, there was no blinding of participants in the trials but this is

inherent to studies of psychological treatments. If applicable, however, outcome assessors were blinded. Fourthly, there was not any study in which the patient's own GP applied PST. As a good doctor-patient relationship contributes to the effectiveness of treatments in non-specific ways, this lack of patient-doctor relationship might have given an underestimation of the effect of PST and this might explain the good outcome of usual care. Fifthly, in three studies the same therapists delivered both PST and a control intervention (Mynors-Wallis 2000; Lynch 2004; Tezel 2006) which might have led to contamination i.e. the application of PST principles in the control intervention, leading to an underestimation of the treatment effect of PST. Sixthly, six studies had a very short follow-up, ranging from 6 to 16 weeks. In medication studies, e.g. Barrett 2000, an 11-week followup is too short to make sure you have reached the optimal effect, even more as in some cases the dose was raised in week 6-8 to the final dosage. For PST we would suggest a follow-up of at least 6 months as optimal effect is likely seen between 3 and 6 months after baseline. Finally, eight studies had more than 20% of patients lost to follow-up. This, however, is common in trials with psychological interventions (Huibers 2003).

Potential biases in the review process

We feel confident we were able to capture the existing body of evidence regarding PST in primary care for several reasons. Compared with the reviews of Cuijpers and Malouff we did not miss any relevant study and our review differed primarily only in so much as we included studies with adults in primary care. Further, no additional studies were identified via reference checking after electronic searches suggesting our search strategy was sufficiently broad. Lastly, as one of our authors (LMW) is a pioneer in the field of PST and a leading authority regarding the application of PST to primary care, he is likely to be familiar with available literature on the subject.

There were, however, potential biases to the review process. Unfortunately we were unable to obtain additional data from two authors in spite of our request (Lynch 1997; Lynch 2004; Tezel 2006). These were, however, low quality trials and the results therefore likely less relevant. Similarly, we were unable able to carry out comprehensive meta-analyses because studies were very heterogeneous in terms of types of participants, types of control treatments, outcome measures, and moments of follow-up. We therefore assigned more weight to the findings of individual studies rather than the limited number of meta-analyses we were able to carry out.

Agreements and disagreements with other studies or reviews

To our knowledge three other systematic reviews on PST have been performed (Cuijpers 2007; Malouff 2007; Bell 2009). Bell included 21 studies (1264 participants) looking at the efficacy of PST in decreasing depressive symptomatology. They showed PST to be equally effective as other psychosocial therapies and medication treatments, and significantly more effective than no treatment and support control groups. Cuijpers included 13 studies (1133 participants) investigating PST in

5 **78 - 79**

depression and found favourable results for PST in most studies. Contrary to our findings, they found that PST had smaller effects in major depression. This may be due to the fact that they included minor depression studies with positive effects that we did not include in our review because they did not take place in primary care. Heterogeneity in their study was very high. The authors therefore concluded that PST has varying effects on depression but that more research is needed to determine whether PST has larger or smaller effects. Malouff included 31 studies (2895 participants) investigating PST in mental and physical health problems. The review showed that PST was significantly more effective than placebo treatment and treatment as usual, but not significantly more effective than other bona fide treatments such as antidepressant medication or other psychological therapies. These findings correspond with our own results.

Authors' conclusions

Implications for practice

We suggest that care providers involved in the treatment of patients with emotional problems in primary care consider PST, along with other therapies of proven effectiveness (cognitive behaviour therapy, interpersonal therapy, antidepressant medication), for the treatment of major depression. It is preferable to be able to offer the patient several equally effective alternatives, because different patients prefer different treatments. Some patients prefer antidepressant medication, notwithstanding recent debate about their effectiveness (Kirsch 2008) or adverse effects (Williams 2000). Most patients seem to prefer nonpharmacological treatments (Fritzsche 2002) however. Ultimately, the treatment choice should be the result of a shared decision process in which the patient's preference is considered after being appropriately informed by the primary care provider. If PST is the treatment of choice, it can be delivered by any appropriately trained therapist as we did not find relevant differences between types of therapists. We encourage interested general practitioners to learn PST as it is one of the briefest psychological treatments available and therefore fits well into the setting of daily practice. Further, a good patient-doctor relationship, as fostered by PST delivered by the patient's own physician, is likely to positively influence the treatment effect of PST (Kaptchuk 2008; Van Os 2005). For emotional problems other than major depression, only one moderate quality trial favoured PST above usual care (Catalan 1991). In six other trials we did not find evidence for statistically significant differences between PST and usual care, placebo, medication or other psychological treatments. Unfortunately, not knowing the exact content of 'usual care' in most trials, it is difficult to determine what specific or non-specific elements were effective. An alternative explanation is that other emotional problems may have relatively favourable outcomes with symptoms simply improving over time. This seems plausible as patients in this review did indeed improve after placebo and 'usual care'. Similarly, the same reasoning may explain why we found significant differences only for major depressed patients and not other milder forms

of depression or mood disorders. However, based on the results of clinical importance we could not exclude positive effects of PST above control treatments

Implications for research

The effectiveness of PST in patients with emotional problems should be studied further considering both the high prevalence and associated disability burden (Wittchen 1999; Lecrubier 2001) as well as our finding that clinical relevant outcomes. are probable and possible. Future researchers should endeavour to provide larger and more homogeneous samples. In addition, as our findings showed improvement of quality of life in PST-treated patients, we recommend further research in homogenous groups stratified by disability levels in order to quantify outcomes and facilitate subgroup analysis. Future studies should also include a control group in which an equal amount of 'time and attention' is provided to participants in an effort to provide a more comparative control for PST. Researchers should also consider studying the effects of PST as delivered by the patient's own GP in order to better assess

the effects of the patient-doctor relationship. Study outcomes should include costs, including both disability days and health care use, particularly for patients with major depression, as this information is currently lacking. Similarly, as long-term outcomes are limited in the current body of evidence on PST in primary care, and given that the optimal effect of PST is likely seen between 3 and 6 months after baseline. we recommend follow-up moments at 3. 6 and 12 months. Lastly, any future comparison with other treatments of proven effectiveness should be designed as equivalence trials (Hermens 2003).

Acknowledgments

We would like to thank the following people: all authors and investigators who were willing to answer our questions and who provided us with additional data; Floris van de Laar who contributed in formulating the research strategy; Madelien van de Beek who helped with double data entry; Rachel Churchill, Vivien Hunot, Hugh McGuire and Jane Dennis who took care of a smooth coordination process with the CCDAN.

Characteristics of studies

Characteristics of included studies

Barrett 2000

Methods	Treatment Effectiveness Project. USA: Multicentre RCT to compare 3 conditions. Blocked and stratified randomisation (by site and diagnosis). Recruitment by referrals from GPs. Assessments on 3 occasions: baseline, 6 and 11 weeks.
Participants	9 DSM-III-R symptoms of major depression >4 weeks) recruited from GPs in 4 cities. Mean age: 61; males: 53%
Interventions	T1 (n=218): Problem-solving treatment. T2 (n=217): Paroxetine (initiated at 10 mg/day and increased at week 2 to the target dose of 20 mg/day. At week 4 or 6, the dose could be increased to 30 mg/day and at week 6 or 8 to 40 mg/day for patients who showed partial or no improvement) and general support. C (n=221): Placebo (titrated as paroxetine) and general support. T1: 6 sessions over 11 weeks, first session 1 hour, subsequent visits 30 minutes each. Antidepressant medication use was prohibited; T2 and C: 6 sessions over 11 weeks, 10-15 minutes each. T1, T2, C: individual treatment. Therapists: PST-providers: 7 PhD psychologists and 5 masters prepared therapists. Medication therapists: psychiatrists (fellows), PC-physicians (residents), or general internists. Therapists received training in PST. Supervision of therapists during treatment unknown.
Outcomes	11-week follow-up: HDRS, HSCL-D, SF-36. In the 18-59 year-old group: Paroxetine and to a lesser degree PST improved remission of dysthymia more than the use of placebo plus nonspecific clinical management. For minor depression, the 3 interventions were equally effective. In the 60 years and older group: PST did not show significantly greater improvement but did show more rapid late-course resolution of symptoms than placebo plus clinical management. PST led to functional improvement for fewer patients than paroxetine.
Notes	Less females than males. Antidepressant prescription forbidden in PST group. PST fidelity check: no. Non-compliance (<4 sessions): T1: 18%; T2: 22%; C: 15%; overall 18%. Lost to follow-up: not reported. Compliance/attendance: 82% attended at least 4 treatment sessions and 79% completed all scheduled treatment sessions (T1: 84% completed all 6 sessions; T2 and C: 94% achieved target dose). T2, C: at each visit, patients self-reported medication adherence. ITT/PP analysis: ITT and PP. Quality Rating: Jadad scale = 3; CCDAN-QRS = 37.

Allocation concealment? Adequate

USA: Singlecentre RCT to compare 2 conditions. Recruitment by referrals from GPs. Assessments on 4 occasions: baseline, 4, 11 and 28 weeks.
113 Patients (aged 18-65) with recent onset complaints (anxiety, tension, depressed mood, irritability, sleep disturbance, or somatic symptoms) recruited from 26 GPs in 16 practices. During the next 4 weeks the GPs were free to give any treatment of their choice. At the end of week 4 patients were assessed at interview; those suitable were randomly allocated to PST or control treatment for the next 6 weeks. At week 4, 66 (58%) were low risk and 47 (42%) high risk. All high risk pts entered the trial. Mean age: 33.5; males: 32%
T1 (n=21): Problem-solving treatment. C (n=26): Usual GP care (any treatment of the GP's choice, whether psychological, social, pharmacological or no treatment.) T1: 4 sessions over 6 weeks; T2: not specified. T1, T2: individual treatment. Therapists: 1 research psychiatrist. Therapists received PST training; PST supervision not reported.
28-week follow-up: PSE, GHQ-28, checklist of helpful components. The PST group showed significant greater reductions in syptoms, both at the end of treatment and at follow-up.
Small sample size, only one psychiatrist delivering intervention. PST fidelity check: no. Non-compliance: T1: 0%; C: 0%; overall: 0%. Lost to follow-up: not reported. Attendance: median number of PST sessions 4. ITT/PP analysis: unclear. Quality Rating: Jadad scale = 3; CCDAN-QRS = 29.

Allocation concealment? Unclear

Dowrick 2000	
Methods	Outcomes of Depression International Network. Europe (Ireland, UK, Norway, Finland, Spain): Multicentre RCT to compare 3 conditions. Recruitment in community. Assessments on 3 occasions: baseline, 6 and 12 months.
Participants	425 Patients (aged 18-65) with a depressive episode (according to ICD-10, dysthymia or adjustment disorder; depressive disorders according to DSM-IV, dysthymia adjustment disorder, bereavement, or other depressive disorders): 52% single major depressive disorders; 19% recurrent major depressive disorders; 16% dysthymia; 4% adjustment disorders; 9% others. Age: 5% 18-25 year-old, 46% 26-45 year-old, 50% 46-65 year-old; males: 35%
Interventions	T1 (n=128): Problem-solving treatment. T2 (n=128): Group psychoeducation for depression (promoting relaxation, positive thinking, pleasant activities and social skills). C (n=189): no intervention from the research team. T1: 6 30-60 minute sessions over 3 months; T2: 8 2,5 hour sessions; C: usual care of GP. T1, C: individual treatment; T2: group treatment. Therapists: Mental health facilitators with qualifications in psychology, nursing, or allied health professionals (unknown number). Therapists were trained and received supervision during treatment.

Dowrick 2000

Outcomes	12-month follow-up: BDI, SF-36. Compared with the control treatment, PST and group psychoeducation participants were less likely to remain cases of depression and more likely to report improved subjective mental and social functioning. Participants assigned to PST were less likely to report depressive symptoms.
Notes	Concurrent antidepressant use was not an exclusion criterion. PST fidelity check; yes. Non-compliance: T1: 37%; T2: 56%; C: 0%; overall 29%. Lost to follow-up: T1: 30%; T2: 23%; C: 32%; overall 39%. Attendance: completed treatment T1: 63%; T2: 44%. ITT/PP analysis: ITT. Quality Rating: Jadad scale = 2; CCDAN-QRS = 34.

Allocation concealment? Adequate

Kendrick 20	<i>i</i> 0:	5
-------------	-------------	---

Methods	UK: Singlecentre RCT to compare 3 conditions. Stratified randomisation (by referring GP). Recruitment by referrals from GPs. Assessments on 3 occasions: baseline, 8 and 26 weeks.
Participants	247 Patients (aged 18-65) with a new episode of anxiety, depression or reaction to life difficulties (duration of symptoms 4 weeks to 6 months; GHQ- 12 score 3 or more): around 42% mixed anxiety and depressive disorder, around 34% a diagnosis of moderate/severe depressive episode, 19% primarily anxiety disorder. Referred by 98 GPs in 62 practices. Mean age: 35; males: 30%
Interventions	T1 (n=90): Care from nurses trained in problem-solving treatment. T2 (n=79): Generic mental health nurse care (Nurses were asked to use whatever treatments they thought appropriate for the patient's problems). C (n=78): Usual GP care (GPs were asked not to refer patients for psychological treatments during the first 8 weeks). T1&T2 first session 1 hour, sessions 2-6 30-45 minutes each. T1, T2, C: individual treatment. Therapists: 53 community mental health nurses (29 T2; 24 T1). Therapists were trained and received supervision during treatment. All patients remained free to consult their GPs throughout the study, and to be prescribed psychotropic drug treatments as the GP thought fit.
Outcomes	26-week follow-up: CIS, GHQ-12, HADS, SAS, EQ-5D, Patient satisfaction. No significant differences between groups in effectiveness at either point. Satisfaction was significantly higher in both nurse-related groups, so were costs.
Notes	PST fidelity check: yes. Non-compliance (less than 4 sessions): T1: 38%; T2: 27%; C: 0%; overall: 22%. Lost to follow-up: T1: 20%; T2: 19%; C: 31%; overall 23%. Attendance: T1: 62% received 4 or more sessions; T2: 73% received 4 or more sessions. ITT/PP analysis: ITT analysis. Quality Rating: Jadad scale = 3; CCDAN-QRS = 37.

Allocation concealment? Adequate

Liu 2007	
Methods	Taiwan: Singlecentre RCT to compare 3 conditions. Blocked randomisation. Recruitment by referrals from nonpsychiatric physicians. Assessments on 2 occasions: baseline and 16 weeks.
Participants	254 Patients (aged 18-70) with common mental disorders (CIS-R score 12 or more): major depressive disorder (53.1%), mixed anxiety and depressive disorder (26.8%; defined as a mixture of anxiety, depression, and various somatic symptoms with the sum of scores for each CIS-R section 12 or more, but not meeting criteria for other specific anxiety or depressive disorder), and anxiety disorders (20.1%). Recruited in general medical clinics in a general hospital (19 family physicians, 12 internists). Mean age: 43.7; males: 19%
Interventions	T1 (n=84): Problem-solving treatment plus usual care. T2 (n=85): Psychiatric consultation plus usual care. C (n=85): Usual care. T1: up to six sessions over 16 weeks, duration of sessions not reported; T2: consultation of a psychiatrist with follow-up visits to the psychiatrist at his/her discretion; C: patients continued seeing their treating physician as usual, and they were permitted to see mental health professionals. T1, T2, C: individual treatment. Therapists T1: 4 psychologists, 2 psychiatric social workers, psychiatric nurse. Therapists were trained; PST supervision not reported. Therapists T2: 6 psychiatrists. All patients remained free to consult their usual physicians throughout the study, and these physicians could prescribe psychotropic drugs.
Outcomes	16-week follow-up: CIS-R, HDRS, SF-36. No statistically significant differences between at 16 weeks on any of the measures.
Notes	Low mean number of PST sessions (2.27) PST fidelity check: unknown. Non-compliance (all dropt out before start of treatment): T1: 41%; T2: 19%; C: 20%, overall: 27%. Lost to follow-up: T1: 25%; T2: 9%; C: 22%; overall 19%. Attendance: on average 2.3 sessions. ITT/PP analysis: ITT. Quality Rating: Jadad scale = 2; CCDAN-QRS = 34.

Allocation concealment? Adequate

Lynch 1997

Methods	USA: Singlecentre RCT to compare 2 conditions. Stratified randomisation (by sex). Recruitment in waiting room of general practice. Assessments on 2 occasions: baseline and 7 weeks.
Participants	29 Patients (aged 18 or older) with a subthreshold or minor depression (as screened by the MOS-Depression Screening Inventory) recruited in the waiting room of a family practice. Mean age: 48; males: 14%
Interventions	T1 (n=15): Telephone-based problem-solving treatment C (n=14): Usual GP care T1: 6 sessions, administered by telephone, once a week, 20 minutes. T1, C: individual treatment. Therapists: 2 student therapists (one a second-year medical student, one graduate nursing student). Therapists were trained and received supervision by psychiatrist.
Outcomes	7-week follow-up: HDRS, BDI, Duke Health Profile, PSI. The PST group had significantly lower post-intervention scores on the HDRS compared with their pre-intervention scores; scores did not differ significantly over time in the control group.

Notes	Small sample, brief follow-up, comparison group treatment not described, PST fidelity check: no. Non-compliance: T1: 27%; C: 0%; overall 14%. Lost to follow-up (complete data): T1: 53%; C: 36%; overall: 45%. Lost to follow-up, just HDRS: T1: 27%; C: 7%; overall: 17%. Attendance: not reported. ITT/PP analysis: unclear.
	Quality Rating: Jadad scale = 2; CCDAN-QRS = 17.

Allocation concealment? Unclear

Lynch 2004

Methods	USA: Singlecentre RCT to compare 3 conditions. Stratified randomisation (by Hamilton scores (11-17, 18-26) and antidepressant use (use, nonuse). Recruitment in waiting room of general practice. Assessments on 2 occasions: baseline and 6 weeks.
Participants	54 Patients (aged 18 or older) with minor or moderate depression (HDRS score 11-26), recruited in the waiting room of 3 family medicine practices. Mean age: 38.5; males: 17%
Interventions	T1 (n=18): Telephone-based problem-solving treatment T2 (n=18): Stress-management intervention (as attention-control group; identifying sources of stress in one's life, discussing importance of diet and exercise in coping with stress.) C (n=18): Usual GP care (whatever the PC-physician deemed appropriate) T1, T2: 6 sessions, administered by telephone, once a week, 20 minutes. T1, T2, C: individual treatment. Therapists: nurses (unknown number). Therapists were trained and received supervision by psychiatrist.
Outcomes	6-week follow-up: HDRS, BDI, Duke Health Profile. No significant differences in the amount of decrease between groups on any scores.
Notes	Small sample, high dropout rate. The same therapists for both psychological treatments. PST was not clearly described. PST fidelity check: no. Non-compliance: T1: 50% (all dropt out before start of treatment); T2: 50% (all dropt out before start of treatment); C: 0%; overall 33%. Lost to follow-up: T1: 50%; T2: 50%; C: 28%; overall: 43%. Attendance: not reported. ITT/PP analysis: unclear. Quality Rating: Jadad scale = 1; CCDAN-QRS = 20.

Allocation concealment? Unclear

Methods	UK: Singlecentre RCT to compare 3 conditions. Stratified randomisation (by severity of depression). Recruitment by referrals from GPs. Assessments on 3 occasions: baseline, 6 and 12 weeks.
Participants	91 Patients (aged 18-65) with major depression (low mood, 4 key depression symptoms, > 2 weeks, HDRS 13 or more): mean HDRS-scores in all groups around 18-19 = major depression. Referred by 26 GPs in 15 local practices. Mean age: 37.1; males: 23%
Interventions	T1 (n=30): Problem-solving treatment T2 (n=31): Amitriptyline (50 mg was prescribed for two nights, followed by an increase of 25 mg every third night until 150 mg was being taken) C (n=30): Placebo (same dosage scheme as T2) T1, T2, C: 6 sessions over 3 months (weeks 1, 2, 3, 5, 7, and 11). One additional therapy session could be offered at the therapist's discretion; first session about 60 minutes, session 2-6 about 70 minutes. T2, C: Amitriptyline and placebo were prescribed as if amitriptyline was being given, and both patient and therapist were blind to the contents of capsules. Specific psychological interventions were avoided, but non-specific interventions such as listening, encouraging, and sympathising were included. T1, T2, C: individual treatment. Therapists: 3 therapists (a psychiatrist experienced in PST and 2 GPs who had received training in PST and in a standardised method of drug administration). Therapists received training; supervision during treatment not reported.
Outcomes	12-week follow-up: HDRS, BDI, PSE, SAS. PST was significantly superior to placebo at both six weeks and 12 weeks on all outcome measures. No significant difference between PST and amitriptyline.
Notes	PST fidelity check: no. Non-compliance: T1: 3% (less than 4 sessions); T2: 13%; C: 13%; overall 10%. Lost to follow-up: not reported per group; overall 29%. Attendance: 4 sessions 90% (T1: 97%; T2: 87%; C: 87%), 6 sessions 71.4% (T1: 93%; T2: 81%; C: 40%). ITT/PP analysis: PP, only data of patients with 4 or more sessions were analysed. Quality Rating: Jadad scale = 3; CCDAN-QRS = 35.

Allocation concealment? Adequate

Mynors-Wallis 1997

Methods	UK: Singlecentre RCT to compare 2 conditions. Recruitment by referrals from GPs. Assessments on 3 occasions: baseline, 8 and 26 weeks.
Participants	70 Patients (aged 18-65), of 4 primary health care centres, with emotional disorders of at least a month's duration, identified by their GP. Complaints: anxiety, tension, depressed mood, irritability and sleep disturbance, somatic symptoms not apparently due to a physical disorder. The GP reassessed the patients 4 weeks after the initial consultation and only referred those with persistent symptoms: mild depression (17%); moderate depression (40%); severe depression (9%); generalised anxiety disorder (3%); mixed anxiety depression (11%); no psychiatric diagnosis (20%). Patients did not have to meet additional severity criteria. Mean age: 38; males: 23%
Interventions	T1 (n=40): Problem-solving treatment C (n=30): Usual GP care T1: 4 or 5 sessions over 8 weeks, unknown duration. T1, C: individual treatment. Therapists: 6 nurses (4 practice nurses, 1 district nurse, 1 health visitor). After treatment (week 8-26) all patients received usual GP care. Therapists received training and video-taped supervision during treatment.

Mynors-Wallis 1995

Outcomes	26-week follow-up: CIS, GHQ-28, SAS, Self-report measure of patient satisfaction, number of disability days, number of days off work. No significant differences at clinical outcomes between groups at either follow-up assessment. The cost analysis showed that PST was more costly in terms of direct healthcare costs but there were significantly less disability days in the PST group.
Notes	No demographic characteristics per group. PST fidelity check: no. Non-compliance: T1: 28%; C: 0%; overall: 16%. Lost to follow-up: T1: 45%; C: 13%; overall: 31%. Attendance: T1: 73% 4 or more sessions. ITT/PP analysis: ITT. Quality Rating: Jadad scale = 3; CCDAN-QRS = 26.
Allocation concealment?	Adequate

Mynors-Wallis 2000

Methods	UK: Singlecentre RCT to compare 4 conditions. Stratified randomisation (by severity and chronicity of depression). Recruitment by referrals from GPs. Assessments on 4 occasions: baseline, 6, 12 and 52 weeks.
Participants	151 Patients (aged 18-65) with major depression (HDRS 13 or more, symptoms > 4 weeks): mean HDRS-scores in all groups around 19-20 = major depression. Referred by 24 GPs. Mean age: 35; males: 23%
Interventions	T1 (n=39): Problem-solving treatment by GP T2 (n=41): Problem-solving treatment by nurse T3 (n=36): SSRI by GP (20 mg paroxetine or 100 mg fluvoxamine once daily. Aim was to encourage patients' compliance with medication in a supportive and encouraging framework but with avoidance of specific psychological interventions). T4 (n=35): SSRI by GP and Problem-solving treatment by nurse T1, T2: 6 sessions over 12 weeks; first 1 hour, session 2-6 30 minutes; T3: 6 sessions over 12 weeks; T4: 6 PST sessions and 6 drug sessions over 12 weeks. In all groups one extra treatment session could be offered if the therapist thought it clinically necessary. T1-T4: individual treatment. Therapists: 3 research GPs and 2 research practice nurses for all 4 groups. Therapists received training and supervision during treatment.
Outcomes	52-week follow-up: HDRS, BDI, CIS, SAS, credibility of treatment. All four groups improved during treatment. There were no significant differences between the four treatment groups at 6, 12, or 52 weeks.
Notes	In our review we only used data of T1 (PST by GP) and T3 (SSRI by GP). The same research GPs delivered both PST and drug sessions. PST fidelity check: no. Non-compliance: T1: 36%; T2: 22%; T3: 17%; T4: 17%; overall 23%. Lost to follow-up: T1: 36%; T2: 32%; T3: 17%; T4: 14%; overall: 25%. Attendance: completed treatment T1: 82%; T2: 61%; T3: 83%; T4: 83%. ITT/PP analysis: ITT. Quality Rating: Jadad scale = 3; CCDAN-QRS = 38.
Allocation concealment?	Adequate

Schreuders	2007
------------	------

Methods	The Netherlands: Singlecentre RCT to compare 2 conditions. Blocked randomisation with random permuted blocks of 4. Recruitment in waiting room of general practice. Assessments on 3 occasions: baseline, 3 and 9 months.
Participants	175 Patients (aged 18 or older) with mental health problems (GHQ-12 score 4 or more, 3 or more GP consultations in past six months): the 3 most frequently occurring diagnoses were major depressive disorder, panic disorder, and somatic disorder which was not further specified. Recruited in waiting rooms of 12 general practices. Mean age: 52.8; males: 29%
Interventions	T1 (n=88): Problem-solving treatment by nurse C (n=87): Usual GP care T1: up to 6 sessions over 3 months; first 1 hour, session 2-6 30 minutes. T1-C: individual treatment. Therapists: 12 nurses with experience in mental health care. Therapists received training and supervision during treatment.
Outcomes	9-months follow-up. HADS, PHQ, SPSI-R, SF-36, PSYCHLOPS, EQ-5D, costs and health care utilization. No significant differences between groups.
Notes	PST fidelity check: yes. Non-compliance: T1: 31%; C: 21%; overall 26%. Lost to follow-up: T1: 33%; C: 22%; overall: 27%. Attendance: T1: on average 4 sessions. ITT/PP analysis: ITT and PP. Quality Rating: Jadad scale = 3; CCDAN-QRS = 33.
Allocation concealment?	Adequate

Tezel 2006

Methods	Turkey: Single centre RCT to compare 2 conditions. Random assignment, matched regarding mean scores on the BDI, parity and education level. Recruitment through screening (EPDS) in 10 primary health care units (BDI 10 or more). Assessments on 2 occasions: baseline and 6 weeks.
Participants	62 Women (postpartum one week) with a risk of postpartum depression (EPDS > 11) but without exhibiting major depression symptoms (as resulted from BDI and SCID). Mean age: 25; males: 0%
Interventions	T1 (n=32): Problem-solving training T2 (n=30): Care Group Intervention (Postnatal care was tailored flexibly to individual needs, using activities from the Nursing Interventions Classification). T1: one visit weekly for 6 weeks, 30–50 minutes. T1, T2: individual treatment. Therapist: one nurse researcher for both interventions. The therapist was trained in PST according to the principles of D'Zurilla. Supervision of therapist unknown.
Outcomes	6-week follow-up: BDI. T2 was effective women for with depressive symptoms and PST was also effective. Utilizing the BDI, it was found out that the nursing care was more effective than education alone.
Notes	No placebo group; very small groups, short follow-up. The same researcher delivered both interventions and interviewed all patients. PST fidelity check: no. Non-compliance: not reported. Lost to follow-up: T1: 0%; T2: 0%; overall: 0%. Attendance: not reported. ITT/PP analysis: ITT. Quality Rating: Jadad scale = 1; CCDAN-QRS = 17.
Allocation concealment?	Unclear

Characteristics of excluded studies

Reason for exclusion

Alexopoulos 2003	not performed in primary care
Allen 2002	not performed in primary care (hospital & oncology clinics)
Arean 1993	not performed in primary care (senior centres)
<u>Atha 1992</u>	no PST investigated
Audrain 1999	not performed in primary care, no PST investigated
Avuso-Mateos 2007	Predictors of compliance with psychological interventions
Bali 2002	no PST investigated (memory training, reasoning training, speed-of-processing-
	training)
Bamblina 2006	PST in control aroup too (supervised vs unsupervised PST)
Barrett 1999	No RCT (description of Barrett 2001 that is included)
Bennun 1985	No RCT (description of 2 bsvcholodical approaches)
Berdiuna 1997	No PS I Investidated
BiumberaLabiaus 2001	No PST as experimental condition investigated
Brouwers 2007	Not ealaby parformed in primary core
Ciochanowski 2004	PEAPLS Study, stapped care model without rendemination at the stort
Colline 1997	No RCT no PST investigated, no emotional symptoms investigated
Del Pino 2004	No PST investigated
Den Boer 2007	No PST not performed in primary care
Doorenbos 2006	No PST investigated
Elliott 2008	Not solely performed in primary care
Elliott 2009	No PST investigated (a combined treatment)
Elliott 2009-b	Not performed in primary care
Fitzpatrick 2005	No PST investigated (one session only)
Friedli 1998	No RCT
Gallo 2000	No RCT (Editorial)
Gellis 2007	Patients not recruited in primary care
Gellis 2008	Patients not recruited in primary care
Gensichen 2006	No RCT
Glasaow 2005	No RCT
Grant 1999	No PST investigated
<u>Grant 2002</u>	Not performed in primary care
Hussain 1981	Not berformed in brimary care (hursing home)
IJTT 2007	Not solely PST investidated (a combined treatment)
IMPACT Study 2001	Stepped care model without randomisation at the start
Inerney 2004	Not performed in primary care
Jacobson 1984	No PST investigated
Jappoup 1980	No PST investigated (a combined treatment)
Kelly 2007	Prediction of engagement with psychological treatment
Kennedy 2005	No RCT
Kina 1995	No RCT
Lapidus 2001	No PST investigated
McDonaah 2005	No PST investigated (present-centered therapy)
Mitchell 2009	Not performed in primary care
Mvnors-Wallis 2001	No RCT
Nezu 1986	Not performed in primary care
Nezu 2003	Not performed in primary care
Pathways Study 2003	Stepped care programme without randomisation at the start
Patsiokas 1985	Not performed in primary care
Paretaineean 2002	No POT
Pivera 2003	
Roberts 1995	Not performed in primary care
Robinson 2008	Prevention rather than treatment of emotional symptoms
Rovner 2007	Not performed in primary care (outpatient clinic)
Rovner 2008	Prevention rather than treatment of emotional symptoms
Rudd Rajab 1996	Not performed in primary care, no PST investigated (combined treatment with
-	PST as 1 of 3 components)
Sahler 2002	Not performed in primary care
Sahler 2005	Not performed in primary care (hospital sites)
Salkovskis 1990	Not performed in primary care, no PST investigated
Schwartz 1998	Not performed in primary care, no PST investigated (one session only)
Sherwood 2005	Not performed in primary care
Shiplev 1973	No RCT no PST investigated
Slimmer 2003	NoRCI
1978 Turning 1978	Not performed in primary care
Tari 1007	No emotional symptoms investigated
Tecoland 1080	No PST investigated (combined treatment)
Vallaa 2000	The For Investigated (complined treatment)
1000	No PCI
Van den Hout 1009	
Van den Hout 1998 Van der Klink 2003	No RCT No RCT No PST investigated (stress inoculation training.)
Van den Hout 1998 Van der Klink 2003 Van Straten 2008	No RCI No RCT No PST investigated (stress inoculation training) No PST investigated (combined treatment)
Van den Hout 1998 Van der Klink 2003 Van Straten 2008 Van Vliet 2006	No RCI No RCT No PST investigated (stress inoculation training) No PST investigated (combined treatment) No RCT
Van den Hout 1998 Van der Klink 2003 Van Straten 2008 Van Vliet 2006 Warmerdam 2008	No RCI No RCT No PST investigated (stress inoculation training) No PST investigated (combined treatment) No RCT No PST investigated (combined treatment)
Van den Hout 1998 Van der Klink 2003 Van Straten 2008 Van Vliet 2006 Warmerdam 2008 Wood 1997	No RCI No RCT No PST investigated (stress inoculation training) No PST investigated (combined treatment) No RCT No PST investigated (combined treatment) Not performed in primary care (palliative care setting)

Characteristics of studies awaiting classification

Oxman 2008

Methods	USA: RCT to compare 2 conditions					
	Assessments on 4 occasions: baseline, 4, 9 and 35 weeks.					
Participants	141 Patients with minor depression					
Interventions	T1 (n=72): PST					
	T2 (n=69): usual care					
	T1.T2: individual treatment.					
Outcomes	35-week follow-up: HRDS, MADRS, HSCL-d-20, Brief COPE, SF-36.					
	PST patients improved more quickly than usual care patients did. Patients with an avoidant coping style showed greater improvement with PST than with usual care.					
Notes	The data as currently presented in the paper are not appropriate for usage in meta-analyses. We asked the authors for more detailed outcome data.					

Characteristics of ongoing studies

Hewitt 2004

Study name	Problem-solving treatment for Primary Care Depression
Methods	RCT
Participants	300 adults (aged 18 or older). Inclusion criteria: primary care patient; HDRS Score >= 10 and minor depression at time of entry and after 4 weeks of observation. Exclusion criteria: psychosis, Obsessive Compulsive Disorder (OCD), or Post Traumatic Stress Disorder (PTSD); active substance abuse; receiving treatment for depression.
Interventions	In the first phase, participants are observed for 4 weeks to identify those most in need of depression-specific treatment. Participants are then exposed to the PST. After 4 weeks, patients who do not respond adequately to the treatment are randomly assigned to either continued PST or to usual care for 9 weeks. Participants are followed for 6 months after the study.
Outcomes	Not reported
Starting date	December 2002
Contact information	Not reported (see: http://clinicaltrials.gov/ct/show/NCT00055328)
Notes	

Walsh 2004

Study name	PST in Geriatric Depression With Executive Dysfunction
Methods	
Participants	240 elderly people (aged 60 or older). Inclusion criteria: nonpsychotic, unipolar major depression; cognitive impairment; English speaking. Exclusion criteria: high suicide risk; dementia; acute or severe medical illness; current psychotherapy.
Interventions	Problem-solving therapy versus brief supportive therapy. Patients are randomly assigned to receive 12 sessions (1 session/week for 12 weeks) of either PST or BST.
Outcomes	Following treatment, patients are followed for 6 months to determine functional and clinical outcomes. Depression scales, disability scales, and scales that measure problem solving skills are used to assess patients.
Starting date	September 2002
Contact information	Jennifer Walsh: 914-997-5914
Notes	

CCDAN Quality Rating Scale

1. Objectives	0 = objectives unclear 1 = objectives clear but main outcome not a priori 2 = objectives clear and main outcome not a priori
2. Sample size	0 = < 50 per Group 1 = 51-100 per Group 2 = > 100 per Group
3. Duration of trial and follow-up	0 = < 3 months 1 = > 3 months and < 6 months 2 = > 6 months
4. Power	0 = not reported 1 = mentioned without details 2 = details of calculation provided
5. Method of allocation	0 = unrandomised and likely to be biased 1 = partial or quasi-randomised with bias possible 2 = randomised allocation
6. Concealment	0 = not done or not reported 1 = partial concealment reported 2 = done adequately
7. Description of treatment	0 = main treatments not clearly described 1 = inadequate details of main or adjunctive treatments 2 = full details of main or adjunctive treatments
8. Blinding of subjects	0 = not done 1 = done, but no test of blind 2 = done and integrity of blind tested
9. Source of subjects, representativeness of sample	0 = source of subjects not described 1 = source of subjects but unrepresentative 2 = source of subjects plus representative sample
10. Diagnostic/inclusion criteria	0 = none 1 = diagnostic criteria or clear inclusion criteria 2 = diagnostic criteria and specification of severity
11. Record of exclusion criteria, exclusions, refusals	0 = criteria and number not reported 1 = criteria or number not reported 2 = criteria and number reported

12. Description of sample demographics	0 = little/no info (only age/sex)
	1 = basic details (marital status/ethnicity)
13 Blinding of assassor	Q = pot dopo
10. Dimaing of assessor	1 = done but no test of blind
	2 = done but no test of blind
14. Compliance /attendance for therapy	0 = not assessed
	1 = assessed for some experimental treatments
	2 = assessed for all experimental treatments
15. Details on side effects	0 = inadequate details
	1 = recorded by group but details inadequate
	2 = full side effect profiles by aroup
16. Record of number/reasons for withdrawal by group	0 = no information on withdrawals by group
	1 = withdrawals by group without reason
	2 = withdrawals and reason by group
17. Outcome measures/use of validated instruments	0 = outcomes not described clearly
	1 = some outcomes not clearly described
	2 = outcomes described and valid/reliable
 Group comparability and adjustment in analysis 	0 = no information on comparability
	1 = some information with adjustment
	2 = sufficient information with adjustment
19. Inclusion of withdrawals in analysis (ITT)	0 = not included or reported
	1 = withdrawals included by estimation of outcome
20 Barris	2 = withdrawais followed up and included in analysis
20. Results presented with data for re-analysis	U = Inadequate presentation
O1 Appropriate statistical enablesis	\geq = comprehensive
21. Appropriate statistical analysis	
	2 – mainiy appropriate
22. Conclusions justified	
ZZ. Conclusions justified	1 = partially
	$2 = v \alpha \epsilon$
23 Declaration of interests	$O = N_0$
Eo. Boolaration of Interests	$2 = Y_{PS}$
	2 100

Quality Rating Scores per study	Barrett 2001	Catalan 1991	Dowrick 2000	Kendrick 2006	Liu 2007	Lynch 1997	Lynch 2004	Mynors- Wallis 1995	Mynors- Wallis 1997	Mynors- Wallis 2000	Screuders 2007	Tezel 2006
Objectives and specification of main outcomes a priori	2	2	2	2	2	2	2	2	1 337 1	2	2	1
Sample size (number per group)	2	0	2	1	1	0	0	0	0	0	1	0
Planned duration of trial including follow up	0	2	2	2	1	0	0	1	2	2	2	0
Power calculation	2	0	2	2	0	0	2	2	0	2	2	0
Method of allocation/randomization	2	2	2	2	2	1	1	2	2	2	2	1
Concealment of allocation	2	0	2	2	2	0	0	2	2	2	2	0
Clear description of treatment (including doses of drugs) & adjunctive treatment	2	1	1	2	1	0	1	2	1	2	1	1
Blinding of subjects	0	0	0	0	0	0	0	0	0	0	0	0
Source of subjects described and representative sample recruitment	1	1	1	1	1	1	1	1	1	1	1	1
Use of diagnostic criteria (or clear specification of inclusion criteria)	2	2	1	2	2	1	2	2	2	2	2	1
Record of exclusion criteria and number of exclusions and refusals reported	2	0	2	2	2	1	2	2	0	2	2	0
Description of sample demographics	2	0	1	2	2	1	1	2	1	2	1	2
Blinding of assessor	1	1	0	0	1	1	0	1	0	1	0	0
Assessment of compliance with experimental treatments (including attendance for therapy)	2	2	1	2	2	0	0	2	2	2	2	0
Details on side-effects	1	0	0	0	0	0	0	0	0	0	0	0
Record of number and reasons for withdrawal by group (i.e. completing treatment or not)	2	2	1	1	1	1	1	2	1	2	1	0
Outcome measures described clearly or use of validated instruments	2	2	2	2	2	2	2	2	2	2	2	2
Information on comparability and adjustment for differences in analysis	2	2	2	2	2	2	1	2	1	2	2	2
Inclusion of withdrawals in analysis (ITT or endpoint)	1	2	2	2	2	0	0	2	1	2	0	2
Presentation of results with inclusion of data for re-analysis of main outcomes (eg SDs)	1	2	2	2	2	0	1	2	2	2	2	2
Appropriate statistical analysis (including correction for multiple tests where applicable)	2	2	2	2	2	2	1	2	2	2	2	1
Conclusions justified	2	2	2	2	2	0	0	2	1	2	2	1
Declaration of interests (eg source of funding)	2	2	2	2	2	2	2	0	2	2	2	0
Total CCDAN-QRS score (max 46)	37	29	34	37	34	17	20	35	26	38	33	17

References to studies

Included studies

Barrett 2000

Barrett JE, Williams JW, Jr, Oxman TE, Frank E, Katon W, Sullivan M, Hegel MT, Cornell JE, Sengupta AS. Treatment of dysthymia and minor depression in primary care: A ramdomized trial in patients aged 18 to 59 years. Journal of Family Practice 2001;50(5):405-412.

Frank E, Rucci P, Katon W, Barrett J, Williams Jr JW, Oxman T, Sullivan M, Cornell J. Correlates of remission in primary care patients treated for minor depression. General hospital psychiatry 2002;24(1):12-19.

Katon W, Russo J, Frank E, Barrett J, Williams Jr JW, Oxman T, Sullivan M, Cornell J. Predictors of nonresponse to treatment in primary care patients with dysthymia. General hospital psychiatry 2002;24(1):20-27.

Oxman TE, Barrett JE, Sengupta A, Katon W, Williams Jr FE, Hegel M. Status of minor depression or dysthymia in primary care following a randomised controlled treatment. General Hospital Psychiatry 2001;23(6):301-310.

Oxman TE, Hull JG. Social support and treatment response in older depressed primary care patients. The Journals of Gerontology - Series B Psychological Sciences and Social Sciences 2001;56(1):35-45.

Schmaling KB, Dimidjian S, Katon W, Sullivan M. Response styles among patients with minor depression and dysthymia in primary care. Journal of Abnormal Psychology 2002;111(2):350-356. Williams JW, Jr, Barrett J, Oxman T, Frank E, Katon W, Sullivan M, Cornell J, Sengupta A. Treatment of dysthymia and minor depression in primary care: A randomised controlled trial in older adults. JAMA: Journal of the American Medical Association 2000;284(12):1519-1526.

Catalan 1991

Catalan J, Gath DH, Anastasiades P, Bond SA, Day A, Hall L. Evaluation of a brief psychological treatment for emotional disorders in primary care. Psychological medicine 1991;21(4):1013-1018.

Dowrick 2000

Dowrick CD. Problem solving treatment and group psychoeducation for depression: multicentre randomised controlled trial. BMJ 2000;321(7274):1450-1454.

Kendrick 2005

Kendrick T, Simons L, Mynors-Wallis L, Gray A, Lathlean J, Pickering R, Harris S, Rivero-Arias O, Gerard K, Thompson C. Cost-effectiveness of referral for generic care or problem-solving treatment from community mental health nurses, compared with usual general practitioner care for common mental disorders. British Journal of Psychiatry 2006;189(JULY):50-59.

Kendrick TS. A trial of problem-solving by community mental health nurses for anxiety, depression and life difficulties among general practice patients. The CPN-GP study. Health technology assessment 2005;9(37):1-87.

Liu 2007

Liu S-1, Huang H-C, Yeh Z-T, Hwang L-C, Tjung J-J, Huang C-R, et al. Controlled trial of problemsolving therapy and consultation-liaison for common mental disorders in general medical settings in Taiwan. General Hospital Psychiatry 2007;29(5):402-408.

Lynch 1997

Lynch DJ, Tamburrino MB, Nagel R. Telephone counseling for patients with minor depression: Preliminary findings in a family practice setting. Journal of Family Practice 1997;44(3):293-298.

Lynch 2004

Lynch D, Tamburrino M, Nagel R, Smith MK. Telephone-based treatment for family practice patients with mild depression. Psychological reports 2004;94(3 Pt 1):785-792.

Mynors-Wallis 1995

MynorsWallis L, Gath D. Predictors of treatment outcome for major depression in primary care. Psychological medicine 1997;27(3):731-736.

MynorsWallis LM, Gath DH, LloydThomas AR, Tomlinson D. Randomised controlled trial comparing problem solving treatment with amitriptyline and placebo for major depression in primary care. British Medical Journal 1995;310(6977):441-445.

Mynors-Wallis 1997

MynorsWallis L, Davies I, Gray A, Barbour F, Gath D. A randomised controlled trial and cost analysis of problem-solving treatment for emotional disorders given by community nurses in primary care. The British Journal of Psychiatry 1997;170:113-119.

Mynors-Wallis 2000

MynorsWallis L. Does problem-solving treatment work through resolving problems? Psychological medicine 2002;32(7):1315-1319.

MynorsWallis LM Gath DH Day. Randomised controlled trial of problem solving treatment, antidepressant medication, and combined treatment for major depression in primary care. British Medical Journal 2000;320(7226):26-30. Thornett AM, MynorsWallis LM. Credibility of problem-solving therapy and medication for the treatment of depression among primary care patients. Medical science monitor : international medical journal of experimental and clinical research 2002;8(3):R193-R196.

Schreuders 2007

Schreuders B, van Marwijk H, Smit JH, Rijmen F, Stalman W, van Oppen P. Primary care patients with mental health problems: Outcome of a randomised clinical trial. British Journal of General Practice 2007;57(544):886-91.

Schreuders B, van Oppen P, van Marwijk HW, Smit JH, Stalman WA. Frequent attenders in general practice: problem solving treatment provided by nurses [ISRCTN51021015]. BMC Family Practice 2005;6:42. [DOI: 10.1186/1471-2296-6-42]

Schreuders B. Problem-solving treatment provided by nurses for patients with mental health problems in general practice. Thesis. Amsterdam, The Netherlands: Bettine Schreuders, 2007.

Tezel A. Comparison of effects of nursing care to problem solving training on levels of depressive symptoms in post partum women. Patient education and counseling 2006;63(1-2):64-73.

Excluded studies

Alexopoulos GS, Raue P, Arean P. Problem-solving therapy versus supportive therapy in geriatric major depression with executive dysfunction. American Journal of Geriatric Psychiatry 2003;11(1):46-52.

Allen SM, Shah AC, Nezu AM, Nezu CM, Ciambrone D, Hogan J, Mor V. A problem-solving approach to stress reduction among younger women with breast carcinoma: A randomised controlled trial. Cancer 2002;94(12):3089-3100. Arean PA, Perri MG, Nezu AM, Schein RL, Christopher F, Joseph TX. Comparative effectiveness of social problem-solving therapy and reminiscence therapy as treatments for depression in older adults. Journal of Consulting & Clinical Psychology 1993;61(6):1003-1010.

Atha C, Salkovskis PM, Storer D. Cognitivebehavioural problem solving in the treatment of patients attending a medical emergency department: A controlled trial. Journal of psychosomatic research 1992;36(4):299-307.

Audrain J, Rimer B, Cella D, Stefanek M, Garber J, Pennanen M, Helzlsouer K, Vogel V, Lin TH, Lerman C. The impact of a brief coping skills intervention on adherence to breast selfexamination among first-degree relatives of newly diagnosed breast cancer patients. Psychooncology 1999;8(3):220-229.

Ayuso-Mateos JL, Pereda A, Dunn G, Vazquez-Barquero JL, Casey P, Lehtinen V, Dalgard O, Wilkinson G, Dowrick C. Predictors of compliance with psychological interventions offered in the community. Psychological Medicine 2007;37(5):717-25.

Ball KBDH. Effects of cognitive training interventions with older adults: a randomised controlled trial. JAMA: Journal of the American Medical Association 2002;288(18):2271-2281.

Bambling M, King R, Raue P, Schweitzer R, Lambert W. Clinical supervision: Its influence on client-rated working alliance and client symptom reduction in the brief treatment of major depression. Psychotherapy Research 2006;16(3):317-331.

Barrett JE Williams JJ. The Treatment Effectiveness Project. A comparison of the effectiveness of paroxetine, problem-solving therapy, and placebo in the treatment of minor depression and dysthymia in primary care patients: background and research plan. General hospital psychiatry 1999;21(4):260-273.

Bennun I. Two approaches to family therapy with alcoholics: problem-solving and systemic therapy. Journal of substance abuse treatment 1985;2(1):19-26.

Berglund GB. Is the wish to participate in a cancer rehabilitation program an indicator of the need? Comparisons of participants and non-participants in a randomised study. Psycho-oncology 1997;6(1):35-46.

Blumberg Lapidus L, Shin SK, Hutton EM. An evaluation of a six-week intervention designed to facilitate coping with psychological stress. Journal of clinical psychology 2001;57(12):1381-1401.

Brouwers EP, de Bruijne MC, Terluin B, Tiemens BG, Verhaak PF. Cost-effectiveness of an activating intervention by social workers for patients with minor mental disorders on sick leave: a randomised controlled trial. European Journal of Public Health 2007;17(2):214-220.

Buwalda FM, Bouman TK, van Duijn MAL. Psychoeducation for hypochondriasis: a comparison of a cognitive-behavioural approach and a problem-solving approach. Behaviour Research and Therapy 2006;45:887-899.

Ciechanowski P, Wagner E, Schmaling K, Schwartz S, Williams B, Diehr P, Kulzer J, Gray S, Collier C, LoGerfo J. Community-integrated home-based depression treatment in older adults. A randomised controlled trial. JAMA 2004;291:1569-1577.

Collins KL. Everyday problem solving in elderly women: contributions of residence, perceived control, and age. Gerontologist 1997;37(3):293-302.

Del Pino A, Gaos M, Dorta R, Garcia M. Modification of coronary-prone behaviours in a sample of coronary patients of low socio-

5 **96 - 97**

economic status. European Journal of Psychiatry 2004;18(4):235-248.

Den Boer PCAM, Wiersma D, Vaarwerk IT, Span MM, Stant AD, Van den Bosch RJ. Cognitive self-therapy for chronic depression and anxiety: a multi-centre randomised controlled study. Psychological Medicine 2007;37(3):329-339.

Doorenbos AG. Physical functioning: effect of behavioral intervention for symptoms among individuals with cancer. Nursing Research 2006;55(3):161-171.

Elliott TR, Brossart D, Berry JW, Fine PR. Problemsolving training via videoconferencing for family caregivers of persons with spinal cord injuries: a randomised controlled trial. Behaviour Research and Therapy 2008;46(11):1220-1229.

Elliott TR, Berry JW, Grant JS. Problem-solving training for family caregivers of women with disabilities: a randomised clinical trial. Behaviour Research and Therapy 2009;47(7):548-558.

Elliott TR, Berry JW. Brief problem-solving training for family caregivers of persons with recent-onset spinal cord injuries: a randomised controlled trial. Journal of Clinical Psychology 2009;65(4):406-422.

Fitzpatrick KK, Witte TK, Schmidt NB. Randomised controlled trial of a brief problemorientation intervention for suicidal ideation. Behavior Therapy 2005;36(4):323-333.

Friedli K, King MB. Psychological treatments and their evaluation. International Review of Psychiatry 1998;10(2):123-126.

Gallo JJ, Coyne JC. The challenge of depression in late life: Bridging science and service in primary care. JAMA: Journal of the American Medical Association 2000;284(12):1570-1572. Gellis ZD, McGinty J, Horowitz A, Bruce ML, Misener E. Problem-solving therapy for late-life depression in home care: a randomised field trial. American Journal of Geriatric Psychiatry 2007;15:968-978.

Gellis ZD, McGinty J, Tierney L, Jordan C, Burton J, Misener E. Randomised controlled trial of problem-solving therapy for minor depression in home care. Research on Social Work Practice 2008;18(6):596-606.

Gensichen J. IMPACT collaborative care improves depression in elderly patients in primary care in the longer term. Evidence-Based Mental Health 2006;9(3):76.

Glasgow RE Price DW. Individualised treatment improves depression in people with depression and diabetes. Evidence-Based Mental Health 2005;8(2):40.

Grant JS. Social problem-solving partnerships with family caregivers. Rehabilitation Nursing 1999;24(6):254-260.

Grant JS Elliott TR Weaver. Telephone intervention with family caregivers of stroke survivors after rehabilitation. Stroke 2002;33(8):2060-2065.

Hussain RA, Lawrence PS. Social reinforcement of activity and problem-solving training in the treatment of depressed institutionalized elderly patients. Cognitive Therapy Research 1981;5(1):57-69.

IJff MA, Huijbregts KML, Van Marwijk HWJ, Beekman ATF, Hakkaart-van Roijen L, Rutten FF, et al. Cost-effectiveness of collaborative care including PST and an antidepressant treatment algorithm for the treatment of major depressive disorder in primary care; A randomised clinical trial. BMC Health Services Research 2007;7:34. Arean P, Hegel M, Vannoy S, Fan MY, Unutzer J. Effectiveness of problem-solving therapy for older, primary care patients with depression: results from the IMPACT project. Gerontologist 2008;48(3):311-323.

Ciechanowski PS, Russo JE, Katon WJ, Von Korff. The association of patient relationship style and outcomes in collaborative care treatment for depression in patients with diabetes. Medical care 2006;44(3):283-291.

Conn DK. Collaborative care depression management for older adults: level of comorbidity does not affect outcome. Evidence-Based Mental Health 2005;8(4):105.

Harpole LH, Williams Jr JW, Olsen MK, Stechuchak KM, Oddone E, Callahan CM, Katon WJ, Lin EH, Grypma LM, Unutzer J. Improving depression outcomes in older adults with comorbid medical illness. General hospital psychiatry 2005;27(1):4-12.

Hunkeler EM, Katon W, Tang L, Williams Jr JW, Kroenke K, Lin EHB, Harpole LH, Arean P, Levine S, Grypma LM, Hargreaves WA, Unutzer J. Long term outcomes from the IMPACT randomised trial for depressed elderly patients in primary care. British Medical Journal 2006;332(7536):259-262.

Katon W, Unutzer J, Fan M-Y, Williams Jr JW, Schoenbaum M, Lin EHB, Hunkeler EM. Costeffectiveness and net benefit of enhanced treatment of depression for older adults with diabetes and depression. Diabetes care 2006;29(2):265-270.

Katon WJ, Schoenbaum M, Fan M-Y, Callahan CM, Williams Jr J, Hunkeler E, Harpole L, Zhou X-HA, Langston C, Unutzer J. Cost-effectiveness of improving primary care treatment of late-life depression. Archives of General Psychiatry 2005;62(12):1313-1320. Lin EH, Katon W, Rutter C, Simon GE, Ludman EJ, Von Korff M, Young B, Oliver M, Ciechanowski PC, Kinder L, Walker E. Effects of enhanced depression treatment on diabetes self-care. Annals of Family Medicine 2006;4(1):46-53.

Lin EH, Katon W, Von Korff M, Tang L, Williams JW, Jr., Kroenke K, Hunkeler E, Harpole L, Hegel M, Arean P, Hoffing M, la Penna R, Langston C, Unutzer J, IMPACT I. Effect of improving depression care on pain and functional outcomes among older adults with arthritis: a randomised controlled trial.[see comment][comment]. JAMA: Journal of the American Medical Association 2003;290(18):2428-2429.

Steffens DC, Snowden M, Fan MY, Hendrie H, Katon WJ, Unutzer J, IMPACT I. Cognitive impairment and depression outcomes in the IMPACT study. American Journal of Geriatric Psychiatry 2006;14(5):401-409.

Unutzer J, Katon WJ, Fan M-Y, Schoenbaum MC, Lin EHB, Della Penna RD, Powers D. Long-term cost effects of collaborative care for late-life depression. American Journal of Managed Care 2008;14(2):95-100.

Unutzer J. Collaborative care management of late-life depression in the primary care setting: a randomised controlled trial. JAMA : the journal of the American Medical Association 2002;288(22):2836-2845.

Unutzer J. Reducing suicidal ideation in depressed older primary care patients. Journal of the American Geriatrics Society 2006;54(10):1550-6.

Unutzer JK. Improving primary care for depression in late life: the design of a multicenter randomised trial. Medical care 2001;39(8):785-799. Williams Jr JW, Katon W, Lin EHB, Noel PH, Worchel J, Cornell J, Harpole L, Fultz BA, Hunkeler E, Mika VS, Unutzer J. The effectiveness of depression care management on diabetes-related outcomes in older patients. Annals of Internal Medicine 2004;140(12):1015-1024.

Inerney-Leo A, Biesecker BB, Hadley DW, Kase RG, Giambarresi TR, Johnson E, Lerman C, Struewing JP. BRCA1/2 testing in hereditary breast and ovarian cancer families: Effectiveness of problemsolving training as a counseling intervention. American Journal of Medical Genetics 2004;130 A(3):221-227.

Jacobson NS. A component analysis of behavioral marital therapy: the relative effectiveness of behavioral exchange and problem solving training. Journal of consulting and clinical psychology 1984;52(2):295-305.

Jacobson NS, Follette VM, Follette WC, HoltzworthMunroe A, Katt JL, Schmaling KB. A component analysis of behavioral marital therapy: 1-year follow-up. Behaviour research and therapy 1985;23(5):549-555.

Jannoun L, Munby M, Catalan J, Gelder M. A home-based treatment program for agoraphobia: Replication and controlled evaluation. Behavior Therapy 1980;11(3):294-305.

Kelly BD, Casey P, Dunn G, Ayuso-Mateos JL, Dowrick C. The role of personality disorder in 'difficult to reach' patients with depression: findings from the ODIN study. European Psychiatry 2007;22(3):153-9.

Kennedy GJ. Psychotherapies and other psychosocial interventions for depression in late life: Innovation through hybridization. Primary Psychiatry 2005;12(11):19-23. King M, Davidson O, Thakore JH, Smeeth L, Watt J, Smyth MG, Mynors-Wallis L, Gath DH, Lloyd-Thomas A. Problem solving treatment for major depression in primary care [3]. British medical journal 1995;310(6989):1266-1268.

Lapidus LB, Shin S-K, Hutton EM. An evaluation of a six-week intervention designed to facilitate coping with psychological stress. Journal of clinical psychology 2001;57(12):1381-1401.

McDonagh A, Friedman M, McHugo G, Ford J, Sengupta A, Mueser K, Demment CC, Fournier D, Schnurr PP, Descamps M. Randomised trial of cognitive-behavioral therapy for chronic posttraumatic stress disorder in adult female survivors of childhood sexual abuse. Journal of Consulting & Clinical Psychology 2005;73(3):515-24.

Mitchell J, Stanimirovic R, Klein B, Vella-Brodrick D. A randomised controlled trial of a self-guided internet intervention promoting well-being. Computers in Human Behavior 2009;25(3):749-760.

Mynors-Wallis L. Problem-solving treatment in general psychiatric practice. Advances in Psychiatric Treatment 2001;7(6):417-425.

Nezu AM. Efficacy of a social problem-solving therapy approach for unipolar depression. Journal of Consulting & Clinical Psychology 1986;54(2):196-202.

Nezu AM, Nezu CM, Felgoise SH, McClure KS, Houts PS. Project Genesis: assessing the efficacy of problem-solving therapy for distressed adult cancer patients. Journal of Consulting & Clinical Psychology 2003;71(6):1036-1048.

Katon W, V. Improving primary care treatment of depression among patients with diabetes mellitus: the design of the Pathways Study. General hospital psychiatry 2003;25(3):158-168. Katon WJ, Russo JE, Von Korff M, Lin EHB, Ludman E, Ciechanowski PS. Long-term effects on medical costs of improving depression outcomes in patients with depression and diabetes. Diabetes Care 2008;31(6):1155-1159.

Katon WJ, Von Korff M, Lin EH, Simon G, Ludman E, Russo J, Ciechanowski P, Walker E, Bush T. The Pathways Study: a randomised trial of collaborative care in patients with diabetes and depression. Archives of General Psychiatry 2004;61(10):1042-1049.

Patsiokas AT, Clum GA. Effects of psychotherapeutic strategies in the treatment of suicide attempters. Psychotherapy 1985;22(2):281-290.

Payne DK Lundberg JC Brennan MF Holland. A psychosocial intervention for patients with soft tissue sarcoma. Psycho-oncology 1997;6(1):65-71.

Porsteinsson AP. Dementia and depression - A treatable condition? Geriatrics 2003;58(8 SUPPL.):8-12.

Rivera PA Shewchuk RE. Project FOCUS: using videophones to provide problem-solving training to family caregivers of persons with spinal cord injuries. Topics in Spinal Cord Injury Rehabilitation 2003;9(1):53-62.

Roberts JBGS. Problem-solving counseling or phone-call support for outpatients with chronic illness: effective for whom? Canadian Journal of Nursing Research 1995;27(3):111-137.

Robinson RG, Jorge RE, Moser DJ, Acion L, Solodkin A, Small SL, Fonzetti P, Hegel M, Arndt S. Escitalopram and problemsolving therapy for prevention of poststroke depression: a randomised controlled trial. JAMA: Journal of the American Medical Association 2008:299(20):2391-2400. Rovner BW, Casten RJ, Hegel MT, Leiby BE, Tasman WS. Preventing depression in age-related macular degeneration. Archives of General Psychiatry 2007;64(8):886-92.

Rovner BW, Casten RJ, Preventing late-life depression in age-related macular degeneration. American Journal of Geriatric Psychiatry 2008;16(6):454-459.

Rudd Rajab MH, Orman DT, Joiner T, Stulman DA, Dixon W. Effectiveness of an outpatient intervention targeting suicidal young adults: preliminary results. Journal of consulting and clinical psychology 1996;64(1):179-190.

Sahler OJ, Varni JW, Fairclough DL, Butler RW, Noll RB, Dolgin MJ, Phipps S, Copeland DR, Katz ER, Mulhern RK. Problem-solving skills training for mothers of children with newly diagnosed cancer: a randomised trial. Journal of Developmental & Behavioral Pediatrics 2002;23(2):77-86.

Sahler OJ, Fairclough DL, Phipps S, Mulhern RK, Dolgin MJ, Noll RB, Katz ER, Varni JW, Copeland DR, Butler RW. Using problem-solving skills training to reduce negative affectivity in mothers of children with newly diagnosed cancer: report of a multisite randomised trial.[see comment]. Journal of Consulting & Clinical Psychology 2005;73(2):272-283.

Salkovskis PM, Atha C, Storer D. Cognitivebehavioural problem solving in the treatment of patients who repeatedly attempt suicide. A controlled trial. The British journal of psychiatry : the journal of mental science 1990;157:871-876.

Schwartz MD, Lerman C, Audrain J, Cella D, Rimer B, Stefanek M, Garber J, Lin TH, Vogel V. The impact of a brief problem-solving training intervention for relatives of recently diagnosed breast cancer patients. Annals of Behavioral Medicine 1998;20(1):7-12. Sherwood PGGC. A cognitive behavioral intervention for symptom management in patients with advanced cancer. Oncology nursing forum 2005;32(6):1190-1198.

Shipley CR, Fazio AF. Pilot study of a treatment for psychological depression. Journal of abnormal psychology 1973;82(2):372-376.

Slimmer L. A collaborative care management programme in a primary care setting was effective for older adults with late life depression. Evidence-Based Nursing 2003;6(3):91.

Slipp S, Kressel K. Difficulties in family therapy evaluation. I. A comparison of insight vs. problem-solving approaches. II. Design critique and recommendations. Family process 1978;17(4):409-422.

Tarrier N, Beckett R, Harwood S, Baker A, Yusupoff L, Ugarteburu I. A trial of two cognitivebehavioural methods of treating drug-resistant residual psychotic symptoms in schizophrenic patients: I. Outcome. British Journal of Psychiatry 1993;162:524-32.

Teri L, Logsdon RG, Uomoto J, McCurry SM. Behavioral treatment of depression in dementia patients: a controlled clinical trial. Journals of Gerontology Series B-Psychological Sciences & Social Sciences 1997;52(4):159-166.

Toseland RW, Rossiter CM, Labrecque. The effectiveness of three group intervention strategies to support family caregivers. American Journal of Orthopsychiatry 1989;59(3):420-429.

Vallee JP. [What is the "best" treatment of depression in general practice?]. Presse Medicale 2000;29(32):1765-1766.

Van den Hout JHC, Vlaeyen JWS, Kole-Snijders AMJ, Heuts PHTG, Willen JEHL, Sillen WJT. Graded activity and problem solving therapy in subacute non-specific low back pain. Physiotherapy 1998;84(4):167.

van der Klink JJL Blonk RWB Schene AH van Dijk. Reducing long term sickness absence by an activating intervention in adjustment disorders: a cluster randomised controlled design. Occupational and environmental medicine 2003;60(6):429-437.

Van Straten A, Cuijpers P, Smits N. Effectiveness of a web-based self-help intervention for symptoms of depression, anxiety, and stress: randomised controlled trial. Journal of Medical Internet Research 2008;10(1):all.

Van Vliet D. Taking a step back can be a step forward. Hearing Journal 2006;59(2):70.

Warmerdam L, Van Straten A, Twisk J, Riper H, Cuijpers P. Internet-based treatment for adults with depressive symptoms: randomised controlled trial. Journal of Medical Internet Research 2008;10(4):all.

Wood BC MynorsWallis LM. Problem-solving therapy in palliative care. Palliative medicine 1997;11(1):49-54.

Studies awaiting classification

Oxman TE, Hegel MT, Hull JG, Dietrich AJ. Problem-solving treatment and coping styles in primary care for minor depression. Journal of Consulting & Clinical Psychology 2008;76(6):933-943.

Ongoing studies

Hewitt C, Barrett A. Problem-solving treatment for primary care depression. ClinicalTrials gov 2004.

Walsh J, Mackin S, Bistricky S, Arean PA, Alexopoulos GS, Raue P. PST in geriatric depression with executive dysfunction. ClinicalTrials gov 2004.

Additional references

Bell AC, D'Zurilla TJ. Problem-solving therapy for depression: a meta-analysis. Clinical Psychology Review 2009;29:348-353.

Bower P, Rowland N, Mellor Clark J, Heywood P, Godfrey C, Hardy R. Effectiveness and cost-effectiveness in primary care. Cochrane Database of Systematic Reviews 2001, Issue 1. Art. No.: CD001025. DOI: 10.1002/14651858. CD001025.pub2.

Ciechanowski P, Wagner, E, Schmaling K, Schwartz S, Williams B, Diehr P, Kulzer J, Gray S, Collier C, LoGerfo J. Community-integrated home-based depression treatment in older adults - a randomised controlled trial. JAMA 2004;291(13):1569-77.

Cuijpers P, van Straten A, Warmerdam L. Problem solving therapies for depression: a meta-analysis. European Psychiatry 2007;22:9-15.

den Boer P, Wiersma D, Russo S, van den BR. Paraprofessionals for anxiety and depressive disorders. Cochrane Database of Systematic Reviews 2005, Issue 2. Art. No.: CD004688. DOI: 10.1002/14651858.CD004688.pub2.

Edwards P, Clarke M, DiGuiseppi C, Pratap S, Roberts I, Wentz R. Identification of randomised controlled trials in systematic reviews: accuracy and reliability of screening records. Statistics in Medicine 2002;21:1635-40.

Egger M, Smith GD, Phillips AN. Metaanalysis: principles and procedures. BMJ 1997;315(7121):1533-7.

Fritzsche K, Armbruster U, Hartmann A, Wirsching M. Psychosocial primary care - what patients expect from their General Practitioners. A crosssectional trial. BMC Psychiatry 2002;8:2-5. Gale C, Oakley-Browne M. Generalised anxiety disorder. In: Clinical evidence. Vol. 8. London: BMJ Publishing Group, 2002.

Gath D, Mynors-Wallis L. Problem-solving treatment in primary care. In: Clark DM, Fairburn CG, editor(s). Science and practice of cognitive behaviour therapy. Oxford: Oxford University Press, 1997.

Geddes J, Butler R, Hatcher S. Depressive disorders. In: Clinical evidence. Vol. 8. London: BMJ Publishing Group, 2002:188-91.

Gureje O. Psychological disorders and symptoms in primary care. Association with disability and service use after 12 months. Social Psychiatry and Psychiatric Epidemiology 2002;37(5):220-4.

Hermens MLM, Van hout HPJ, Terluin B, adèr HJ, De haan M. The importance of equivalence trials in showing the usefulness of treatments [Het belang van equivalentieonderzoek om de waarde van behandelingen aan te tonen]. Ned Tijdschr Geneeskd 2003;147(44):2162-66.

Higgins JPT, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. BMJ 2002;327:557-60.

Huibers MJH, Beurskens AJHM, Bleijenberg G, Schayk CP van. The effectiveness of psychosocial interventions delivered by general practitioners. Cochrane Database of Systematic Reviews 2003, Issue 2. Art. No.: CD003494. DOI: 10.1002/14651858.CD003494.

Jadad AR; Moore RA; Carroll D; Jenkinson C; Reynolds DJM; Gavaghan DJ; McQuay HJ. Assessing the quality of reports of randomised clinical trials: is blinding necessary? Controlled Clinical Trials 1996;17:1-12.

Juni P, Altman DG, Egger M. Assessing the quality of randomised controlled trials. In: Egger M, Davey Smith G, Altman DG, editor(s). Systematic reviews in health care. Meta-analysis in context. London: BMJ Publishing Group, 2001:87-108.

Kaptchuk TJ, Kelley JM, Conboy LA, Davis RB, Kerr CE, Jacobson EE, Kirsch I, Schyner RN, Hyun Nam B, Nguyen LT, Park M, Rivers AL, McManus C, Kokkotou E, Drossman DA, Goldman P, Lembo AJ. Components of placebo effect: randomised controlled trial in patients with irritable bowel syndrome. BMJ 2008;336(7651):999-1003.

Kirsch I, Deacon BJ, Huedo-Medina TB, Scoboria A, Moore TJ, Johnson BT. Initial severity and antidepressant benefits: a meta-analysis of data submitted to the Food and Drug Administration. PLoS Med. 2008;5(2):e45.

Kumar S, Oakley-Browne M. Panic disorder. In: Clinical evidence. Vol. 8. London: BMJ Publishing Group, 2002:197-8.

Kunz R, Oxman AD. The unpredictability paradox: review of empirical comparisons of randomised and non-randomised trials. BMJ 1998;317:1185-90.

Lecrubier Y. The burden of depression and anxiety in general medicine. J Clin Psychiatry 2001;62(Suppl 8):4-9.

Malouff JM, Thorsteinsson EB, Schutte NS. The efficacy of problem solving therapy in reducing mental and physical health problems: a metaanalysis. Clinical Psychology Review 2007;27:46-57.

Man-Son-Hing M, Laupacis A, O'Rourke K, Molnar F, Mahon J, Chan KBY, Wells G. Determination of the clinical importance of study results - a review. J Gen Intern Med 2002;17:469-76.

Moncrieff J. Development of a quality assessment instrument for trials of treatments for depression and neurosis. Intern J Methods Psych Research 2001;10:126-33. Mynors-Wallis LM, Gath DH, Day A, Baker F. Randomised controlled trial of problem solving treatment, antidepressant medication and combined treatment for major depression in primary care. BMJ 2000;320:26-30.

Mynors-Wallis L. Problem-solving treatment for Anxiety and Depression. Oxford University Press, November 2005.

Oopik P, Aluoja A, Kalda R, Maaroos HI. Family doctors' problems and motivating factors in management of depression. BMC Family Practice 2006;7:64.

Ormel J, VonKorff M, Ustun B, Pini S, Korten A, Oldehinkel T Common mental disorders and disability across cultures. Results from the WHO Cllaborative Study on Psychological Problems in General Health Care. JAMA 1994;272(22):1741-8.

Unutzer J, Katon W, Williams JW, Callahan CM, Harpole L, Hunkeler EM, Hoffing M, Arean P, Hegel MT, Schoenbaum M, Oishi SM, Langston CA. Improving primary care for depression in late life the design of a multicenter randomised trial. Med Care 2001;39:785-799.

Van Os TWDP, Van den Brink RHS, Tiemens BG, Jenner JA, Van der Meer K, Ormel J. Communicative skills of general practitioners augment the effectiveness of guideline-based depression treatment. Journal of Affective Disorders 2005;84:43-51.

Williams JW Jr, Mulrow CD, Chiquette E, Noel PH, Aguilar C, Cornell J. A systematic review of newer pharmacotherapies for depression in adults: evidence report summary. Annals of Internal Medicine 2000;132(9):743-756.

Wittchen HU, Lieb R, Wunderlich U, Schuster P. Comorbidity in primary care: presentation and consequences. Journal of Clinical Psychiatry 1999;60 Suppl 7:29-36.

Data and analyses

1 PST versus Usual Care at 3-month follow-up

Outcome or Subgroup 1.1 Change in GHQ-28 score	Studies 2	Participants	Statistical Method Mean Difference (IV,	Effect Estimate -1.20 [-4.23, 1.83]
1.2 Change in HADS depression score	2	252	Random, 95% Cl) Mean Difference (IV,	-0.53 [-1.66, 0.59]
1.3 Change in HADS anxiety score	2	252	Random, 95% Cl) Mean Difference (IV,	0.40 [-0.66, 1.45]
1.4 Change in HADS total score	1	130	Random, 95% Cl) Mean Difference (IV,	-0.70 [-3.37, 1.97]
1.5 Change in CIS score	3	314	Random, 95% CI) Mean Difference (IV,	0.14 [-2.25, 2.53]
1.6 Change in SAS score	2	185	Random, 95% Cl) Mean Difference (IV,	-0.05 [-0.22, 0.12]
1.7 Change in HDRS score	1	129	Random, 95% Cl) Mean Difference (IV,	0.50 [-1.44, 2.44]
1.8 Change in PSE score	1	47	Random, 95% CI) Mean Difference (IV,	-3.00 [-5.64, -0.36]
1.9 Change in GHQ-12 score	1	122	Random, 95% Cl) Mean Difference (IV,	-0.70 [-2.10, 0.70]
1.10 Change in SPSI score	1	130	Random, 95% Cl) Mean Difference (IV,	3.45 [-1.15, 8.05]
1.11 Change in SF-36 Mental Health score	2	259	Random, 95% CI) Mean Difference (IV,	0.49 [-3.22, 4.21]
1.12 Change in SF-36 Physical Health score	2	259	Random, 95% CI) Mean Difference (IV,	0.30 [-3.44, 4.05]
1.13 Change in Euroqol-5D score	2	185	Mean Difference (IV,	0.07 [-0.01, 0.14]
1.14 All depression scales, changes	5	491	Std. Mean Difference	-0.09 [-0.28, 0.09]
1.15 All quality of life scales, changes	4	444	(IV, Random, 95% CI) Std. Mean Difference (IV, Random, 95% CI)	0.12 [-0.09, 0.32]

2 PST versus Usual Care at 6-month follow-up

Outcome or Subgroup 2.1 Change in GHQ-28 score	Studies 2	Participants 105	Statistical Method Mean Difference (IV,	Effect Estimate -0.47 [-5.66, 4.72]
2.2 Change in SAS score	2	180	Random, 95% CI) Mean Difference (IV, Dandem, 95% CI)	0.05 [-0.09, 0.18]
2.3 Change in CIS score	2	180	Mean Difference (IV, Random, 95% CI)	0.97 [-2.36, 4.31]
2.4 Change in HADS depression score	1	122	Mean Difference (IV, Random, 95% CI)	-0.73 [-2.38, 0.92]
2.5 Change in HADS anxiety score	1	122	Mean Difference (IV,	1.59 [0.00, 3.18]
2.6 Change in BDI score	1	218	Mean Difference (IV,	-3.09 [-5.78, -0.40]
2.7 Recoveries at BDI	1	270	Random, 95% CI) Risk Ratio (M-H,	2.54 [1.85, 3.50]
2.8 Change in GHQ-12 score	1	122	Mean Difference (IV,	-0.50 [-1.77, 0.77]
2.9 Change in PSE score	1	47	Random, 95% CI) Mean Difference (IV,	-4.70 [-8.35, -1.05]
2.10 Change in SF-36 Mental Health score	1	218	Random, 95% CI) Mean Difference (IV,	8.91 [2.82, 15.00]
2.11 Change in SF-36 Mental Role score	1	218	Random, 95% Cl) Mean Difference (IV,	11.89 [-0.02, 23.80]
2.12 Change in SF-36 Social Function score	1	218	Random, 95% CI) Mean Difference (IV,	17.47 [8.85, 26.09]
2.13 Change in Euroqol-5D score	2	180	Random, 95% Cl) Mean Difference (IV,	0.06 [-0.02, 0.13]
2.14 All depression scales, changes	4	445	Random, 95% Cl) Std. Mean Difference	-0.19 [-0.52, 0.13]
2.14.1 Emotional symptoms	3	227	(IV, Random, 95% CI) Std. Mean Difference	-0.15 [-0.63, 0.33]
2.14.2 Depressive disorders	1	218	(IV, Random, 95% CI) Std. Mean Difference	-0.31 [-0.58, -0.04]
2.15 All quality of life scales, changes	3	497	(IV, Random, 95% CI) Std. Mean Difference	0.33 [0.15, 0.51]
2.15.1 Emotional symptoms	2	180	(IV, Random, 95% CI) Std. Mean Difference	0.22 [-0.08, 0.52]
2.15.2 Depressive disorders	1	317	(IV, Random, 95% CI) Std. Mean Difference (IV, Random, 95% CI)	0.39 [0.16, 0.62]

5 104 - 105

Outcome or Subgroup 3.1 Change in HADS total score	Studies 1	Participants 127	Statistical Method Mean Difference (IV,	Effect Estimate 0.38 [-2.45, 3.21]
3.2 Change in BDI score	1	218	Mean Difference (IV,	-2.05 [-4.61, 0.51]
3.3 Recoveries at BDI	1	171	Random, 95% CI) Risk Ratio (M-H,	1.01 [0.80, 1.29]
3.4 Change in SPSI score	1	127	Mean Difference (IV,	4.51 [-0.63, 9.65]
3.5 Change in SF-36 Mental Health score	2	345	Mean Difference (IV,	2.07 [-1.92, 6.07]
3.6 Change in SF-36 Physical Health score	1	127	Mean Difference (IV,	-2.43 [-6.70, 1.84]
3.7 Change in SF-36 Mental Role score	1	218	Mean Difference (IV,	6.60 [-4.83, 18.03]
3.8 Change in SF-36 Social Function score	1	218	Random, 95% CI) Mean Difference (IV,	14.01 [5.51, 22.51]
3.9 All depression scales, changes	2	345	Std. Mean Difference	-0.11 [-0.36, 0.14]
3.9.1 Emotional symptoms	1	127	(IV, Random, 95% CI) Std. Mean Difference	0.05 [-0.30, 0.40]
3.9.2 Depressive disorders	1	218	(IV, Random, 95% CI) Std. Mean Difference	-0.21 [-0.49, 0.06]
3.10 All quality of life scales, changes	2	345	(IV, Random, 95% CI) Std. Mean Difference	0.15 [-0.06, 0.36]
3.10.1 Emotional symptoms	1	127	(IV, Random, 95% CI) Std. Mean Difference	0.05 [-0.30, 0.40]
3.10.2 Depressive disorders	1	218	(IV, Random, 95% CI) Std. Mean Difference (IV, Random, 95% CI)	0.21 [-0.06, 0.48]

3 PST versus Usual Care at 12-month follow-up

4 PST versus Medication at 3-month follow-up

Outcome or Subgroup 4.1 Change in HDRS score	Studies 3	Participants 465	Statistical Method Mean Difference (IV,	Effect Estimate 0.93 [-0.21, 2.07]
4 1.1 Minor Depression & Dysthymia, 18-59	1	126	Random, 95% Cl) Mean Difference (IV,	0.30 [-1.28, 1.88]
year-old 4.1.2 Minor depression & Dysthymia, 60-	1	215	Random, 95% Cl) Mean Difference (IV,	1.70 [0.32, 3.08]
years and older 4.1.3 Major depression	2	124	Random, 95% Cl) Mean Difference (IV,	0.36 [-2.87, 3.60]
4.2 Recovered cases (HDRS<8)	з	564	Random, 95% CI) Risk Ratio (M-H,	0.98 [0.83, 1.17]
4.2.1 Major depression	2	129	Random, 95% CI) Risk Ratio (M-H,	0.96 [0.69, 1.32]
4.2.2 Dysthymia, 18-59 year-old	1	85	Random, 95% CI) Risk Ratio (M-H,	0.92 [0.54, 1.57]
4.2.3 Minor Depression, 18-59 year-old	1	75	Random, 95% CI) Risk Ratio (M-H,	0.97 [0.60, 1.57]
4.2.4 Dysthymia, 60-year and older	1	141	Risk Ratio (M-H,	1.17 [0.80, 1.72]
4.2.5 Minor Depression, 60-year and older	1	134	Random, 95% CI) Risk Ratio (M-H,	0.91 [0.58, 1.43]
4.3 Change in BDI score	2	124	Mean Difference (IV,	-0.75 [-5.26, 3.76]
4.4 HSCL-D symptom resolution, 18-59	1	320	Mean Difference (IV,	0.05 [-0.34, 0.43]
4 4.1 week 0-2	1	160	Mean Difference (IV,	0.24 [0.07, 0.41]
4.4.2 week 2-11	1	160	Mean Difference (IV,	-0.15 [-0.33, 0.03]
4.5 Change in HSCL-D score	1	330	Mean Difference (IV, Bandom, 95% CI)	0.16 [-0.01, 0.33]
4.5.1 Minor Depression & Dysthymia, 18-59	1	123	Mean Difference (IV,	0.25 [0.05, 0.45]
4.5.2 Dysthymia, 60-year and older	1	109	Mean Difference (IV, Bandom, 95% CI)	0.25 [0.01, 0.49]
4.5.3 Minor Depression, 60-year and older	1	98	Mean Difference (IV, Bandom, 95% CI)	-0.01 [-0.23, 0.21]
4.6 Change in HSCL-D score	1	156	Mean Difference (IV, Random, 95% (I)	-0.10 [-0.32, 0.12]
4.6.1 Minor depression [Frank]	1	156	Mean Difference (IV, Random, 95% CI)	-0.10 [-0.32, 0.12]
4.7 Change in PSE score	1	56	Mean Difference (IV, Random, 95% CI)	0.30 [-4.08, 4.68]
4.8 Change in SAS score	2	124	Mean Difference (IV, Random, 95% CI)	-0.03 [-0.46, 0.41]

4.9 Change in CIS score	1	68
4.10 Change in SF-36 Mental Health score	1	329
4.10.1 Minor Depression & Dysthymia, 18-	1	124
4.10.2 Dysthymia, 60 years and older	1	111
4.10.3 Minor Depression, 60 years and	1	94
4.11 Change in SF-36 Physical Health score	1	124
4.11.1 Minor Depression & Dysthymia, 18-	1	124
4.12 All depression scales, changes	3	461
4.12.1 Major depression	2	131
4.12.2 Minor Depression & Dysthymia, 18-	1	123
4.12.3 Dysthymia, 60-year and older	1	109
4.12.4 Minor Depression, 60-year and older	1	98

2.30 [-3.01, 7.61]
-2.70 [-5.20, -0.19]
-3.90 [-8.18, 0.38]
-3.40 [-7.49, 0.69]
-0.30 [-5.02, 4.42]
-2.80 [-5.64, 0.04]
-2.80 [-5.64, 0.04]
0 21 [-0 02 0 44]
0.06[-0.39_0.51]
0.44[0.08_0.80]
0.44 [0.08, 0.80]
0.39[0.01, 0.77]
-0.02 [-0.41, 0.38]

5 PST versus Medication at 6-month follow-up

Outcome or Subgroup	Studies	Participants	Statistical Method	Effect Estimate
5.1 Recovered cases (HDRS<7) at 6 month	1	286	Risk Ratio (M-H,	0.90 [0.65, 1.26]
follow-up [Oxman: Treatment completers only] 5.1.1 Minor depression, 18 years and older	1	132	Random, 95% Cl) Risk Ratio (M-H,	0.77 [0.57, 1.02]
5.1.2 Dysthymia, 18 years and older	1	154	Random, 95% Cl) Risk Ratio (M-H, Random, 95% Cl)	1.07 [0.80, 1.44]

6 PST versus Medication at 12-month follow-up

Outcome or Subgroup 6.1 Change in BDI score	Studies	Participants 55	Statistical Method Mean Difference (IV,	Effect Estimate -0.80 [-8.06, 6.46]
6.2 Change in HDRS score	1	55	Random, 95% Cl) Mean Difference (IV,	-1.70 [-5.66, 2.26]
6.3 Recovered cases (HDRS<8)	1	75	Random, 95% Cl) Risk Ratio (M-H,	1.11 [0.76, 1.63]
6.4 Change in CIS score	1	55	Random, 95% Cl) Mean Difference (IV,	-3.60 [-9.92, 2.72]
6.5 Change in SAS score	1	55	Random, 95% Cl) Mean Difference (IV, Random, 95% Cl)	-0.20 [-0.54, 0.14]

7 PST versus Placebo at 3-month follow-up

Outcome or Subgroup 7.1 Change in HDRS score	Studies 2	Participants 418	Statistical Method Mean Difference (IV,	Effect Estimate -1.89 [-4.16, 0.38]
7.1.1 Minor Depression & Dysthymia, 18-59	1	133	Random, 95% CI) Mean Difference (IV, Bandam 05% CI)	-1.30 [-3.02, 0.42]
7.1.2 Minor Depression & Dysthymia, 60-	1	230	Mean Difference (IV,	-0.35 [-1.70, 1.00]
7.1.3 Major Depression	1	55	Mean Difference (IV,	-5.70 [-9.21, -2.19]
7.2 Recovered cases (HDRS<8)	2	505	Random, 95% CI) Risk Ratio (M-H,	1.06 [0.77, 1.46]
7.2.1 Major depression	1	60	Random, 95% CI) Risk Ratio (M-H,	2.25 [1.16, 4.36]
7.2.2 Dysthymia, 18-59 year-old	1	85	Random, 95% CI) Risk Ratio (M-H,	0.87 [0.52, 1.46]
7.2.3 Minor Depression, 18-59 year-old	1	76	Random, 95% CI) Risk Ratio (M-H,	0.90 [0.56, 1.43]
7.2.4 Dysthymia, 60-year and older	1	142	Random, 95% Cl) Risk Ratio (M-H,	1.23 [0.83, 1.83]
7.2.5 Minor Depression, 60-year and older	1	142	Random, 95% Cl) Risk Ratio (M-H,	0.77 [0.50, 1.19]
7.3 Change in BDI score	1	55	Random, 95% Cl) Mean Difference (IV,	-8.40 [-14.44, -2.36]
7.4 Change in HSCL-D score	1	351	Random, 95% Cl) Mean Difference (IV, Random, 95% Cl)	0.00 [-0.12, 0.13]

7.4.1 Minor Depression & Dysthymia, 18-59	1	131	Mean Difference (IV,
year-old 7.4.2 Minor depression, 60-years and older	1	103	Random, 95% Cl) Mean Difference (IV,
7.4.3 Dysthymia, 60-years and older	1	117	Random, 95% Cl) Mean Difference (IV,
7.5 Change in HSCL-D score	1	168	Random, 95% Cl) Mean Difference (IV,
7.5.1 Minor depression [Frank]	1	168	Random, 95% Cl) Mean Difference (IV,
7.6 HSCL-D symptom resolution, 18-59	1	322	Random, 95% Cl) Mean Difference (IV,
year-old 7.6.1 week 0-2	1	161	Random, 95% Cl) Mean Difference (IV,
7.6.2 week 2-11	1	161	Random, 95% Cl) Mean Difference (IV,
7.7 Change in PSE score	1	55	Random, 95% Cl) Mean Difference (IV.
7.8 Change in SAS score	1	55	Random, 95% Cl) Mean Difference (IV.
7.9 Change in SF-36 Mental Health score	1	350	Random, 95% Cl) Mean Difference (IV.
7 9.1 Minor Depression & Dysthymia, 18-59	1	131	Random, 95% Cl) Mean Difference (IV.
year-old 7.9.2 Dysthymia, 60 years and older	1	118	Random, 95% Cl) Mean Difference (IV.
7.9.3 Minor Depression, 60 years and older	1	101	Random, 95% Cl) Mean Difference (IV.
7.10 Change in SF-36 Physical Health score	1	131	Random, 95% Cl) Mean Difference (IV.
7 10.1 Minor Depression & Dysthymia, 18-	1	131	Random, 95% Cl) Mean Difference (IV.
59 year-old 7.11 All depression scales, changes	2	406	Random, 95% Cl) Std. Mean Difference
7.11.1 Maior Depression	1	55	(IV, Random, 95% CI) Std. Mean Difference
7.11.2 Minor Depression & Dysthymia, 18-	1	131	(IV, Random, 95% CI) Std. Mean Difference
59 year-old 7 11 3 Minor depression, 60-years and	- 1	103	(IV, Random, 95% CI) Std. Mean Difference
7.11.4 Dysthymia, 60-years and older	1	117	(IV, Random, 95% Cl) Std. Mean Difference
			(IV, Random, 95% CI)

ference (IV, 0.11 [-0.11, 0.33] -0.10 [-0.31, 0.11] 0.00 [-0.22, 0.22] -0.10 [-0.32, 0.12] -0.10 [-0.32, 0.12] 0.03 -0.30, 0.37] 0.20 [0.03, 0.37] -0.14 [-0.33, 0.05] -6.10 [-10.47, -1.73] -0.52 [-0.87, -0.17] 1.80 -2.08, 5.68 -1.60 [-5.48, 2.28] 4.88 [1.91, 7.85] 1.63 -2.45, 5.71] -2.00 [-4.64, 0.64] -2.00 [-4.64, 0.64] an Difference -0.17 [-0.53, 0.20] lom, 95% CI) an Difference -0.85 [-1.40, -0.29] lom, 95% CI) an Difference 0.17 -0.17, 0.51 lom, 95% Cl) an Difference -0.18 [-0.57, 0.21] lom, 95% Cl) an Difference 0.00 [-0.36, 0.36]

8 PST versus Placebo at 6-month follow-up

Outcome or Subgroup	Studies	Participants	Statistical Method	Effect Estimate
8.1 Recovered cases (HDRS<7) at 6 months	1	290	Risk Ratio (M-H,	1.02 [0.75, 1.39]
[Oxman: Treatment completers only]			Random, 95% Cl)	
8.1.1 Minor Depression	1	136	Risk Ratio (M-H,	0.88 [0.65, 1.19]
			Random, 95% Cl)	
8.1.2 Dysthymia	1	154	Risk Ratio (M-H,	1.20 [0.87, 1.64]
, ,			Random, 95% Cl)	

9 PST versus Other Psychological Treatment at 3-month follow-up

Studies 1	Participants 140	Statistical Method Mean Difference (IV,	Effect Estimate -0.30 [-2.16, 1.56]
1	133	Random, 95% Cl) Mean Difference (IV,	-0.01 [-1.54, 1.52]
1	133	Mean Difference (IV,	-0.31 [-1.74, 1.12]
1	62	Random, 95% CI) Mean Difference (IV,	6.30 [2.84, 9.76]
1	62	Random, 95% Cl) Risk Ratio (M-H,	0.49 [0.31, 0.76]
1	133	Random, 95% Cl) Mean Difference (IV,	0.00 [-0.14, 0.14]
2	273	Random, 95% Cl) Mean Difference (IV,	-0.83 [-3.22, 1.57]
1	133	Random, 95% Cl) Mean Difference (IV,	-0.48 [-1.83, 0.87]
1	140	Random, 95% Cl) Mean Difference (IV,	1.40 [-5.87, 8.67]
1	140	Random, 95% Cl) Mean Difference (IV,	3.30 [-4.12, 10.72]
1	133	Random, 95% Cl) Mean Difference (IV, Random, 95% Cl)	0.05 [-0.04, 0.14]
	Studies 1 1 1 1 1 1 1 1 1 1 1 1 1	Studies Participants 1 140 1 133 1 62 1 62 1 133 2 273 1 133 1 133 1 133 1 133 1 140 1 140 1 140 1 133	Studies Participants Statistical Method 1 140 Mean Difference (IV, Random, 95% CI) 1 133 Mean Difference (IV, Random, 95% CI) 1 133 Mean Difference (IV, Random, 95% CI) 1 133 Mean Difference (IV, Random, 95% CI) 1 62 Mean Difference (IV, Random, 95% CI) 1 62 Risk Ratio (M-H, Random, 95% CI) 1 133 Mean Difference (IV, Random, 95% CI) 2 273 Mean Difference (IV, Random, 95% CI) 1 133 Mean Difference (IV, Random, 95% CI) 1 140 Mean Difference (IV, Random, 95% CI) 1 140 Mean Difference (IV, Random, 95% CI) 1 140 Mean Difference (IV, Random, 95% CI) 1 133 Mean Difference (IV, Random, 95% CI) 1 140 Mean Difference (IV, Random, 95% CI) 1 133 Mean Difference (IV, Random, 95% CI)

9.12 All depression scales, changes	З	335	Std. Mean Difference 0.22 [-0.31, 0.74]
9.12.1 Emotional symptoms	2	273	(IV, Random, 95% CI) Std. Mean Difference -0.06 [-0.30, 0.17]
9.12.2 Minor Depression	1	62	(IV, Random, 95% Cl) Std. Mean Difference 0.89 [0.37, 1.42]
9.13 All quality of life scales, changes	2	273	(IV, Random, 95% Cl) Std. Mean Difference 0.12 [-0.11, 0.36]
9.13.1 Emotional symptoms	2	273	(IV, Random, 95% Cl) Std. Mean Difference 0.12 [-0.11, 0.36]
			(IV, Random, 95% CI)

10 PST versus Other Psychological Treatment at 6-month follow-up

Outcome or Subgroup 10.1 Change in HADS depression score	Studies 1	Participants 133	Statistical Method Mean Difference (IV,	Effect Estimate 0.31 [-1.15, 1.77]
10.2 Change in HADS anxiety score	1	133	Random, 95% Cl) Mean Difference (IV,	0.38 [-1.10, 1.86]
10.3 Change in BDI score	1	172	Random, 95% CI) Mean Difference (IV,	-2.48 [-5.50, 0.54]
10.4 Recoveries at BDI	1	178	Random, 95% CI) Risk Ratio (M-H,	1.08 [0.83, 1.39]
10.5 Change in SAS score	1	133	Mean Difference (IV,	0.11 [-0.04, 0.26]
10.6 Change in CIS score	1	133	Mean Difference (IV,	4.00 [0.12, 7.88]
10.7 Change in GHQ-12 score	1	133	Mean Difference (IV,	0.45 [-0.62, 1.52]
10.8 Change in SF-36 Mental Health score	1	172	Mean Difference (IV,	2.55 [-3.92, 9.02]
10.9 Change in SF-36 Mental Role score	1	172	Mean Difference (IV, Bandom, 95% CI)	3.52 [-9.53, 16.57]
10.10 Change in SF-36 Social Function	1	172	Mean Difference (IV, Random, 95% CI)	3.22 [-6.03, 12.47]
10.11 Change in Euroqol-5D score	1	133	Mean Difference (IV, Bandom, 95% CI)	0.03 [-0.06, 0.12]
10.12 All depression scales, changes	2	305	Std. Mean Difference	0.04 [-0.53, 0.62]
10.12.1 Emotional symptoms	1	133	Std. Mean Difference	0.35 [0.00, 0.69]
10.12.2 Depressive disorders	1	172	Std. Mean Difference	-0.25 [-0.55, 0.05]
10.13 All quality of life scales, changes	2	305	Std. Mean Difference	0.11 [-0.11, 0.34]
10.13.1 Emotional symptoms	1	133	Std. Mean Difference	0.11 [-0.23, 0.45]
10.13.2 Depressive disorders	1	172	Std. Mean Difference (IV, Random, 95% CI)	0.12 [-0.18, 0.42]

11 PST versus Other Psychological Treatment at 12-month follow-up

Outcome or Subgroup 11.1 Change in BDI score	Studies 1	Participants 172	Statistical Method Mean Difference (IV,	Effect Estimate -4.15 [-7.01, -1.29]
11.2 Recoveries at BDI	1	172	Random, 95% Cl) Risk Ratio (M-H, Bandam, 95% Cl)	1.17 [0.90, 1.51]
11.3 Change in SF-36 Mental Health score	1	172	Mean Difference (IV,	7.69 [1.25, 14.13]
11.4 Change in SF-36 Mental Role score	1	172	Mean Difference (IV,	13.61 [0.97, 26.25]
11.5 Change in SF-36 Social Function score	1	172	Random, 95% Cl) Mean Difference (IV, Random, 95% Cl)	6.67 [-2.48, 15.82]

12 Patient Satisfaction at 6-month follow-up

Outcome or Subgroup	Studies	Participants	Statistical Method	Effect Estimate
12.1 PST versus Usual Care	1	114	Mean Difference (IV,	6.00 [3.43, 8.57]
12.2 PST versus Nurse Care	1	125	Random, 95% Cl) Mean Difference (IV, Random, 95% Cl)	0.40 [-1.66, 2.46]
13 Days off work & Costs, PST versus Usual Care

Outcome or Subgroup	Studies	Participants	Statistical Method	Effect Estimate
13.1 Total number of days off work, 6-month	2	192	Mean Difference (IV,	-4.32 [-19.00, 10.36]
follow-up (Emotional symptoms)			Random, 95% CI)	
13.2 Total number of days off work, 9-month	1	121	Mean Difference (IV,	-9.50 [-30.38, 11.38]
follow-up (Emotional symptoms)			Random, 95% CI)	
13.3 Total costs of days off work, 6-month	1	122	Mean Difference (IV,	2093.00 [-1815.51,
follow-up (Emotional symptoms)			Random, 95% CI)	6001.51
13.4 Total costs of days off work, 9-month	1	121	Mean Difference (IV,	-1683.00 -4021.08,
follow-up (Emotional symptoms)			Random, 95% Cl)	655.08]

14 Days off work & Costs, PST versus Other Psychological Treatment, at

6-month follow-up

Outcome or Subgroup	Studies	Participants	Statistical Method	Effect Estimate
14.1 Total number of days off work past 6		113	Mean Difference (IV,	3.60 [-5.85, 13.05]
months (Emotional symptoms) 14.2 Total costs of days off work past 6 months (Emotional symptoms)	1	113	Random, 95% Cl) Mean Difference (IV, Random, 95% Cl)	2186.00 [-1893.12, 6265.12]

15 Health care use & Costs, PST versus Usual Care

Outcome or Subgroup 15.1 Total number of GP consultations.	Studies	Participants	Statistical Method Mean Difference (IV.	Effect Estimate -0.57 [-1.14, 0.00]
3-month follow-up (Emotional symptoms) 15.2 Total number of GP consultations,	2	192	Random, 95% Cl) Mean Difference (IV,	-1.08 [-2.57, 0.42]
6-month follow-up (Emotional symptoms) 15.3 Total number of GP consultations,	1	121	Random, 95% Cl) Mean Difference (IV,	-0.70 [-2.09, 0.69]
15.4 Total costs of GP consultations,	2	192	Mean Difference (IV,	-18.47 [-51.49,
6-month follow-up (Emotional symptoms) 15.5 Total costs of GP consultations,	1	121	Random, 95% Cl) Mean Difference (IV,	14.55] -14.00 [-42.40,
9-month follow-up (Emotional symptoms) 15.6 Total Medications costs, 6-month	2	192	Random, 95% Cl) Std. Mean Difference	14.40] -0.30 [-0.82, 0.21]
follow-up (Emotional symptoms) 15.7 Total Medications costs, 9-month	1	121	(IV, Random, 95% CI) Std. Mean Difference	0.06 [-0.29, 0.42]
follow-up (Emotional symptoms) 15.8 Total costs, 6-month follow-up	2	192	(IV, Random, 95% CI) Mean Difference (IV,	168.61 [-94.62.
(Emotional symptoms) 15.9 Total costs. 9-month follow-up	1	121	Random, 95% CI) Mean Difference (IV	431.83]
(Emotional symptoms)	1	122	Random, 95% CI) Mean Difference (IV	130.71]
follow-up (Emotional symptoms)	÷	101	Random, 95% CI)	450 00 [4400 40
15.11 Costs of hospital admissions, 9-month follow-up (Emotional symptoms)	1	121	Mean Difference (IV, Random, 95% Cl)	-450.00 [-1130.46, 230.46]

16 Health care use & Costs, PST versus Other Psychological Treatment

Outcome or Subgroup 16.1 Total number of GP consultations, 3-month follow-up (Emotional symptoms)	Studies 1	Participants 140	Statistical Method Mean Difference (IV, Random, 95% (1)	Effect Estimate -0.70 [-1.25, -0.15]
16.2 Total number of GP consultations,	1	113	Mean Difference (IV,	-1.22 [-2.21, -0.23]
6-month follow-up (Emotional symptoms) 16.3 Total costs of GP consultations,	1	113	Random, 95% CI) Mean Difference (IV,	-25.00 [-45.59,
6-month follow-up (Emotional symptoms) 16.4 Total medications costs, 6-month	1	113	Random, 95% Cl) Mean Difference (IV,	-4.41] 4.00 [-20.22, 28.22]
follow-up (Emotional symptoms) 16.5 Total costs, 6-month follow-up	1	113	Random, 95% Cl) Mean Difference (IV,	32.00 [-132.94.
(Emotional symptoms) 16.6 Costs of hospital admissions, 6-month follow-up (Emotional symptoms)	1	113	Random, 95% Cl) Mean Difference (IV, Random, 95% Cl)	196.94] 13.00 [-35.09, 61.09]

1. PST versus Usual Care at 3-month follow-up

Analysis 1.1. Comparison 1 PST versus Usual Care at 3-month follow-up, Outcome 1 Change in GHQ-28 score

		PST		Usı	Jal Ca	re		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% Cl
Catalan 1991	-12.8	7.22	21	-10.2	7.2	26	53.3%	-2.60 [-6.75, 1.55]	
Mynors-Wallis 1997	-4.4	7.89	34	-4.8	9.76	29	46.7%	0.40 [-4.03, 4.83]	
Total (95% CI)			55			55	100.0%	-1.20 [-4.23, 1.83]	-
Heterogeneity: Tau ² = (0.00; Cł	ni² = 0.	94, df =	1 (P =	0.33);	l² = 0%			-10 -5 0 5 10
l est for overall effect. 2	2 = 0.78	(P = t).44)						Favours PST Favours Usual Ca

Analysis 1.2. Comparison 1 PST versus Usual Care at 3-month follow-up, Outcome 2 Change in HADS depression score



Analysis 1.3. Comparison 1 PST versus Usual Care at 3-month follow-up, Outcome 3 Change in HADS anxiety score

		PST		Usi	al Ca	re		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
Kendrick 2005	-3.96	4.35	71	-4.78	4.05	51	49.5%	0.82 [-0.68, 2.32]	
Schreuders 2007	-1.45	4.17	61	-1.43	4.49	69	50.5%	-0.02 [-1.51, 1.47]	
Total (95% CI)			132			120	100.0%	0.40 [-0.66, 1.45]	-
Heterogeneity: Tau ² =	0.00; Cł	ni² = 0.	61, df =	= 1 (P =	0.44);	² = 0%	b		-4 -2 0 2 4
Test for overall effect:	Z = 0.73	(P=0	0.46)						Favours PST Favours Usual Ca

Analysis 1.5. Comparison 1 PST versus Usual Care at 3-month follow-up, Outcome 5 Change in CIS score

		PST		Us	ual Car	e		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% CI
Kendrick 2005	-10.4	11.92	71	-10.9	13.43	51	26.9%	0.50 [-4.11, 5.11]	100
Liu 2007	-9.7	9.17	63	-9.8	9.39	66	55.9%	0.10 [-3.10, 3.30]	
Mynors-Wallis 1997	-7.4	11.56	34	-7.1	11.73	29	17.2%	-0.30 [-6.07, 5.47]	
Total (95% CI)			168			146	100.0%	0.14 [-2.25, 2.53]	
Heterogeneity: Tau ² = Test for overall effect:	0.00; Cł Z = 0.11	ni² = 0.0 I (P = 0.	5, df = 91)	2 (P = 0	.98); l² :	= 0%			-4 -2 0 2 4 Favours PST Favours Usu

Analysis 1.6. Comparison 1 PST versus Usual Care at 3-month follow-up, Outcome 6 Change in SAS score

		PST		Usu	al Ca	re		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% Cl
Kendrick 2005	-0.34	0.43	71	-0.34	0.48	51	74.2%	0.00 [-0.17, 0.17]	-8-
Mynors-Wallis 1997	-0.3	0.54	34	-0.1	0.72	29	25.8%	-0.20 [-0.52, 0.12]	
Total (95% CI)			105			80	100.0%	-0.05 [-0.22, 0.12]	+
Heterogeneity: Tau ² =	0.00; Cł	ni² = 1.	19, df =	= 1 (P =	0.28);	² = 16	%		
Test for overall effect:	Z = 0.59	(P = (0.56)						Favours PST Favours Usual Ca

Analysis 1.11. Comparison 1 PST versus Usual Care at 3-month follow-up, Outcome 11 Change in SF-36 Mental Health score

		PST		Us	ual Car	е		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
Liu 2007	19.2	21.81	63	21.2	23.51	66	22.5%	-2.00 [-9.82, 5.82]	
Schreuders 2007	3.67	11.62	61	2.45	12.92	69	77.5%	1.22 [-3.00, 5.44]	
Total (95% CI)			124			135	100.0%	0.49 [-3.22, 4.21]	-
Heterogeneity: Tau ² =	0.00; Cł	ni² = 0.5	0, df =	1 (P = 0	.48); l2 :	= 0%			10 5 0 5 10
Test for overall effect:	Z = 0.26	(P=0.	79)					Fa	vours Usual Care Favours PST

Analysis 1.12. Comparison 1 PST versus Usual Care at 3-month follow-up, Outcome 12 Change in SF-36 Physical Health score

		PST		Us	ual Car	е		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
Liu 2007	13.8	22.51	63	14.4	22.34	66	23.4%	-0.60 [-8.34, 7.14]	-81
Schreuders 2007	2.84	12.06	61	2.26	12.83	69	76.6%	0.58 [-3.70, 4.86]	
Total (95% CI)			124			135	100.0%	0.30 [-3.44, 4.05]	-
Heterogeneity: Tau ² = Test for overall effect:	0.00; Ch Z = 0.16	$hi^2 = 0.0$ hi(P = 0.0)	7, df = 87)	1 (P = 0	.79); l²:	= 0%		Fav	-10 -5 0 5 10 /ours Usual Care Favours PST

Analysis 1.13. Comparison 1 PST versus Usual Care at 3-month follow-up, Outcome 13 Change in Euroqol-5D score

		PST		Usu	al Ca	е		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
Kendrick 2005	0.17	0.28	71	0.13	0.23	51	64.0%	0.04 [-0.05, 0.13]	
Mynors-Wallis 1997	0.26	0.22	34	0.14	0.27	29	36.0%	0.12 [-0.00, 0.24]	
Total (95% CI)			105			80	100.0%	0.07 [-0.01, 0.14]	-
Heterogeneity: Tau ² =	0.00; Ch	ni² = 1.	05, df =	= 1 (P =	0.30);	² = 5%		-	0.2 0.1 0 0.1 0.2
Test for overall effect:	Z = 1.79	(P=0	0.07)					Favo	urs Usual Care Favours PS

Analysis 1.14. Comparison 1 PST versus Usual Care at 3-month follow-up,

Outcome 14 All depression scales, changes

		PST		Us	ual Car	e		Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% Cl
Catalan 1991	-11.2	4.18	21	-8.2	5.05	26	9.5%	-0.63 [-1.22, -0.04]	
Kendrick 2005	-10.4	11.92	71	-10.9	13.43	51	24.4%	0.04 [-0.32, 0.40]	
Liu 2007	-9.7	9.17	63	-9.8	9.39	66	26.4%	0.01 [-0.33, 0.36]	
Mynors-Wallis 1997	-7.4	11.56	34	-7.1	11.73	29	13.3%	-0.03 [-0.52, 0.47]	
Schreuders 2007	-1.92	4.16	61	-1.26	4.38	69	26.4%	-0.15 [-0.50, 0.19]	
Total (95% CI)			250			241	100.0%	-0.09 [-0.28, 0.09]	+
Heterogeneity: Tau ² =	0.00; Ch	ni² = 4.2	3, df = -	4 (P = 0	.38); l² =	= 6%		-	1 05 0 05 1
Test for overall effect:	Z = 0.97	(P=0.3	33)						Favours PST Favours Usual Ca

Analysis 1.15. Comparison 1 PST versus Usual Care at 3-month follow-up, Outcome 15 All quality of life scales, changes

		PST		Us	ual Car	е		Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
Kendrick 2005	0.17	0.28	71	0.13	0.23	51	27.0%	0.15 [-0.21, 0.51]	
Liu 2007	19.2	21.81	63	21.2	23.51	66	29.0%	-0.09 [-0.43, 0.26]	
Mynors-Wallis 1997	0.26	0.22	34	0.14	0.27	29	14.9%	0.49 [-0.02, 0.99]	
Schreuders 2007	3.67	11.62	61	2.45	12.92	69	29.1%	0.10 [-0.25, 0.44]	
Total (95% CI)			229			215	100.0%	0.12 [-0.09, 0.32]	
Heterogeneity: Tau ² = Test for overall effect:	0.01; Ch Z = 1.13	ni² = 3.4 (P = 0.	5, df = 26)	3 (P = 0	.33); ² :	= 13%		Favo	1 -0.5 0 0.5 1 Jrs Usual Care Favours PST

2 PST versus Usual Care at 6-month follow-up

Analysis 2.1. Comparison 2 PST versus Usual Care at 6-month follow-up, Outcome 1 Change in GHQ-28 score

		PST Us				re		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
Catalan 1991	-13.1	7.53	21	-9.9	8.28	26	48.5%	-3.20 [-7.73, 1.33]	
Mynors-Wallis 1997	-6.1	7.14	32	-8.2	8.71	26	51.5%	2.10 [-2.06, 6.26]	
Total (95% CI)			53			52	100.0%	-0.47 [-5.66, 4.72]	
Heterogeneity: Tau ² = Test for overall effect:	9.12; Ch Z = 0.18	ni² = 2. (P = 0	85, df =).86)	= 1 (P =	0.09);	² = 65	%		-10 -5 0 5 10 Favours PST Favours Usual Ca

Analysis 2.2. Comparison 2 PST versus Usual Care at 6-month follow-up,

Outcome 2 Change in SAS score

		PST		Usı	al Ca	ге		Mean Difference	Mean Difference IV, Random, 95% Cl
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	l Weight	IV, Random, 95% CI	
Kendrick 2005	-0.4	0.44	71	-0.46	0.43	51	75.9%	0.06 [-0.10, 0.22]	
Mynors-Wallis 1997	-0.4	0.49	32	-0.4	0.57	26	24.1%	0.00 [-0.28, 0.28]	
Total (95% CI)			103			77	1 00.0 %	0.05 [-0.09, 0.18]	
Heterogeneity: Tau ² =	0.00; Cł	ni² = 0.	14, df =	= 1 (P =	0.71);	$ ^{2} = 0\%$		-	02.01 0.01.02
Test for overall effect:	Z = 0.66	6 (P = 0	0.51)						Favours PST Favours Usua

Analysis 2.3. Comparison 2 PST versus Usual Care at 6-month follow-up, Outcome 3 Change in CIS score

		PST		Us	ual Car	е		Mean Difference	Mean Difference			
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% CI			
Kendrick 2005	-12.6	12.3	71	-14.6	11.38	51	62.0%	2.00 [-2.24, 6.24]				
Mynors-Wallis 1997	-10.5	10.53	32	-9.8	10.39	26	38.0%	-0.70 [-6.11, 4.71]				
Total (95% CI)			103			77	100.0%	0.97 [-2.36, 4.31]				
Heterogeneity: Tau ² =	0.00; Ch	ni² = 0.5	9, df =	1 (P = 0	.44); l²	= 0%			-10 -5 0 5 10			
lest for overall effect:	Z = 0.57	(P = 0.	57)						Favours PST Favours Usual Ca			

Analysis 2.13. Comparison 2 PST versus Usual Care at 6-month follow-up, Outcome 13 Change in Euroqol-5D score

		PST		Usi	al Ca	re		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% C	I IV, Random, 95% CI
Kendrick 2005	0.18	0.29	71	0.13	0.23	51	64.9%	0.05 [-0.04, 0.14]	
Mynors-Wallis 1997	0.27	0.22	32	0.2	0.26	26	35.1%	0.07 [-0.06, 0.20]	-
Total (95% CI)			103			77	100.0%	0.06 [-0.02, 0.13]	-
Heterogeneity: Tau ² =	0.00; Cł	ni² = 0.	06, df =	= 1 (P =	0.80);	² = 0%	5		
Test for overall effect:	Z = 1.50	(P = (0.13)					F	avours Usual Care Favours PST

Analysis 2.14. Comparison 2 PST versus Usual Care at 6-month follow-up,

Outcome 14 All depression scales, changes

		PST		Us	ual Car	е		Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% CI
2.14.1 Emotional sym	ptoms								
Catalan 1991	-12.3	4.69	21	-7.6	7.94	26	17.7%	-0.69 [-1.28, -0.10]	
Kendrick 2005	-12.6	12.3	71	-14.6	11.38	51	28.3%	0.17 [-0.19, 0.53]	
Mynors-Wallis 1997 Subtotal (95% CI)	-10.5	10.53	32 1 24	-9.8	10.39	26 103	20.6% 66.7%	-0.07 [-0.58, 0.45] -0.15 [-0.63, 0.33]	-
Heterogeneity: Tau ² = Test for overall effect:	0.12; Ch Z = 0.62	i ² = 5.86 (P = 0.5	6, df = 2 63)	? (P = 0.	05); i [»] =	66%			
2.14.2 Depressive dis	orders								
Dowrick 2000 Subtotal (95% CI)	-10.63	9.83	89 89	-7.54	10.16	129 1 29	33.3% 33.3%	-0.31 [-0.58, -0.04] -0.31 [-0.58, -0.04]	
Heterogeneity: Not app Test for overall effect:	plicable Z = 2.22	(P = 0.0	13)						
Total (95% CI)			213			232	100.0%	-0.19 [-0.52, 0.13]	
Heterogeneity: Tau ² = Test for overall effect:	0.06; Ch Z = 1.16	i ² = 7.42 (P = 0.2	2, df = 3 25)	8 (P = 0.	06); l² =	60%		_	-1 -0.5 0 0.5 1 Favours PST Favours Usual C

Analysis 2.15. Comparison 2 PST versus Usual Care at 6-month follow-up, Outcome 15 All quality of life scales, changes

		PST		Us	ual Car	e	:	Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
2.15.1 Emotional syn	nptom s								and the second second
Kendrick 2005	0.18	0.29	71	0.13	0.23	51	24.9%	0.19 [-0.17, 0.55]	
Mynors-Wallis 1997	0.27	0.22	32	0.2	0.26	26	12.0%	0.29 [-0.23, 0.81]	
Subtotal (95% CI)			103			77	36.9%	0.22 [-0.08, 0.52]	-
Heterogeneity: Tau ² =	0.00; Ch	i ² = 0.1	0, df =	1 (P = 0	.75); l² :	= 0%			
Test for overall effect:	Z = 1.45	(P = 0.	15)						
2.15.2 Depressive dis	sorders								
Dowrick 2000	19.11	22.15	128	10.2	23.15	189	63.1%	0.39 [0.16, 0.62]	
Subtotal (95% CI)			128			189	63.1%	0.39 [0.16, 0.62]	-
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 3.38	(P = 0.	0007)						
Total (95% CI)			231			266	100.0%	0.33 [0.15, 0.51]	+
Heterogeneity: Tau ² =	0.00; Ch	i ² = 0.9	1. df = :	2 (P = 0	.63); l² =	= 0%		· · · · · · · · · · · · · · · · · · ·	
Test for overall effect:	Z = 3.57	(P = 0.	0004)		,,			-	1 -0.5 0 0.5 1
			.,					Favo	urs usual care Favours PST

3 PST versus Usual Care at 12-month follow-up

Analysis 3.5. Comparison 3 PST versus Usual Care at 12-month follow-up, Outcome 5 Change in SF-36 Mental Health score

		PST		Us	ual Car	е		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
Dowrick 2000	21.82	22.72	89	17	22.32	129	35.6%	4.82 [-1.27, 10.91]	
Schreuders 2007	5.94	11.79	59	5.38	11.84	68	64.4%	0.56 [-3.56, 4.69]	
Total (95% CI)			148			197	100.0%	2.07 [-1.92, 6.07]	-
Heterogeneity: Tau ² =	2.03; Ch	i² = 1.2	9, df =	1 (P = 0	.26); l² :	= 22%			10 5 0 5 10
Test for overall effect:	Z = 1.02	(P = 0.3	31)					Fav	vours Usual Care Favours PST

Analysis 3.9. Comparison 3 PST versus Usual Care at 12-month follow-up, Outcome 9 All depression scales, changes

		PST		Usu	al Ca	re		Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% Cl
3.9.1 Emotional symp	toms								
Schreuders 2007 Subtotal (95% CI)	-2.98	8.45	59 59	-3.36	7.72	68 68	40.8% 40.8%	0.05 [-0.30, 0.40] 0.05 [-0.30, 0.40]	
Heterogeneity: Not app	licable								
Test for overall effect: 2	2 = 0.26	(P = 0	.79)						
3.9.2 Depressive diso	rders								
Dowrick 2000	-11.96	9.32	89	-9.91	9.67	129	59.2%	-0.21 [-0.49, 0.06]	
Subtotal (95% CI)			89			129	59.2%	-0.21 [-0.49, 0.06]	
Heterogeneity: Not app	licable								
Test for overall effect: 2	2 = 1.55	(P = 0	.12)						
Total (95% CI)			148			197	100.0%	-0.11 [-0.36, 0.14]	
Heterogeneity: Tau ² = (0.01: Ch	i² = 1.3	4.df=	1 (P = ().25); I	2 = 269	6		t. A
Test for overall effect: 2	2 = 0.84	(P = 0	.40)	Ì					-0.5 -0.25 0 0.25 0.5 Favours PST Favours Usual Ca

Analysis 3.10. Comparison 3 PST versus Usual Care at 12-month follow-up, Outcome 10 All quality of life scales, changes

		PST		Us	ual Car	е		Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
3.10.1 Emotional syn	ptoms								
Schreuders 2007 Subtotal (95% CI)	5.94	11.79	59 59	5.38	11.84	68 68	37.6% 3 7.6%	0.05 [-0.30, 0.40] 0.05 [-0.30, 0.40]	
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 0.26	(P = 0.	79)						
3.10.2 Depressive dis	orders								
Dowrick 2000 Subtotal (95% CI)	21.82	22.72	89 89	17	22.32	129 129	62.4% 62.4%	0.21 [-0.06, 0.48] 0.21 [-0.06, 0.48]	-
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 1.55	(P = 0.	12)						
Total (95% CI)			148			197	100.0%	0.15 [-0.06, 0.36]	-
Heterogeneity: Tau ² = Test for overall effect:	0.00; Cł Z = 1.38	ni² = 0.5 ? (P = 0.	5, df = 17)	1 (P = 0	.46); l² :	= 0%		- Fav	-0.5 -0.25 0 0.25 0.5 yours Usual Care Favours PST

4 PST versus Medication at 3-month follow-up

Analysis 4.1. Comparison 4 PST versus Medication at 3-month follow-up, Outcome 1 Change in HDRS score

	PST				dicatio	n		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
4.1.1 Minor Depressi	on & Dy	sthym	ia, 18-	59 year	-old				
Barrett 2000 Subtotal (95% CI)	-8.5	4.56	64 64	-8.8	4.51	62 62	36.4% 36.4%	0.30 [-1.28, 1.85] 0.30 [-1.28, 1.88]	-
Heterogeneity: Not ap	olicable								
Test for overall effect:	Z = 0.37	(P=(0.71)						
4.1.2 Minor depression	on & Dy	sthyrr	ia, 60- <u>-</u>	years a	nd old	er			
Barrett 2000	-5.8	5.19	112	-7.5	5,15	103	43.7%	1.70 [0.32, 3.08]	-8-
Subtotal (95% CI)			112			103	43.7%	1.70 [0.32, 3.08]	•
Heterogeneity: Not app	olicable								
Test for overall effect:	Z = 2.41	(P=0	0.02)						
4.1.3 Major depressio	on								
Mynors-Wallis 1995	-12.3	6.53	29	-11	6.8	27	9.8%	-1.30 [-4.80, 2.20]	
Mynors-Wallis 2000	-12	7.8	34	-14	6.66	34	10.0%	2.00 1-1 45 5.451	
Subtotal (95% CI)			63			61	19.8%	0.36 [-2.87, 3.60]	
Heterogeneity: Tau ² =	2.31: Cł	ni² = 1.	73. df =	= 1 (P =	0.19):	$ ^2 = 42^{\circ}$	%		
Test for overall effect:	Z = 0.22	(P = 0)	0.83)		,,				
		,							
Total (95% CI)			239			226	100.0%	0.93 [-0.21, 2.07]	•
Heterogeneity Tau ² =	0.28 [,] Cł	ni² = 3	72 df=	= 3 (P =	0.29).	l ² = 199	%		+ + + + +
Test for overall effect:	7 = 1.59		111	- (1	- 0),				-4 -2 0 2 4
	- 1.00								Favours PST Favours Medicatic

Analysis 4.2. Comparison 4 PST versus Medication at 3-month follow-up, Outcome 2 Recovered cases (HDRS<8)

	PST		Medicat	tion		Risk Ratio	Risk Ratio
Study or Subgroup	Events	Total	Events	Total	Welght	M-H, Random, 95% C	M-H, Random, 95% Cl
4.2.1 Major depressio	n						
Mynors-Wallis 1995	18	30	16	31	15.3%	1.16 [0.74, 1.82]	
Mynors-Wallis 2000	20	34	24	34	24.5%	0.83 [0.58, 1.19]	=
Subtotal (95% CI)		64		65	39.9%	0.96 [0.69, 1.32]	
Total events	38		40				
Heterogeneity: Tau ² = (0.01; Chi ²	= 1.31	df = 1 (P	= 0.25); I² = 24%	1	
Test for overall effect: 2	Z = 0.27 (I	P = 0.7	8)				
4 2 2 Dysthymia 18-5	9 vear-oli	4					
Barrett 2000	16	- 13	17	12	10.8%	0 92 [0 54 1 57]	
Subtotal (95% CI)	10	43	17	42	10.8%	0.92 [0.54, 1.57]	
Total events	16		17				
Heterogeneity: Not ann	licable						
Test for overall effect: 7	7 = 0.31 (1	⊃ = 0 7i	6)				
	- 0.01 (0.1	.,				
4.2.3 Minor Depressio	m, 18-59	year-ol	d				
Barrett 2000	17	37	18	38	13.2%	0.97 [0.60, 1.57]	=
Subtotal (95% CI)		37		38	13.2%	0.97 [0.60, 1.57]	
Total events	17		18				
Heterogeneity: Not app	licable						
Test for overall effect: 2	Z = 0.12 (I	⊃ = 0.9	0)				
4.2.4 Dysthymia, 60-y	ear and o	lder					
Barrett 2000	33	72	27	69	20.7%	1 17 [0 80 1 72]	
Subtotal (95% CI)	00	72	-	69	20.7%	1.17 [0.80, 1.72]	
Total events	33		27				
Heterogeneity: Not app	licable						
Test for overall effect: 2	Z = 0.80 (I	⊃ = 0.4	2)				
4.2.5 Minor Depressio	n 60.ves	r and a	older				
Porrott 2000		a anu i cc		60	15 50/	0.04 [0.59, 4.42]	
Subtotal (95% CI)	23	66	20	68	15.5%	0.91 [0.58, 1.43]	
Total events	23		26			a.a. [a.aa, 1.4a]	
Heteroreneity: Not ann	licable		20				
Test for overall effect: 2	Z = 0.41 (1	⊃ = 0.6	8)				
Total (95% CI)		202		202	100.0%	0 00 10 02 1 171	
Total avonto	107	202	100	202	100.070	u.ao [u.oa, 1.17]	
Hotorogonoity: Tou? = (127 2.00: Chi2	- 2 24	128 df = 5 (D	- 0.00	12 - 00/		
Test for everall offerst:	5.00, CNP 7 = 0.10 (1	- 2.34	, ui = 5 (P	- 0.80), 1" = 0%		0.5 0.7 1 1.5 2
Test for overall effect? 2	⊆ 0.18 (I	- = 0.8	0)			F	avours Medication Favours PST

Analysis 4.3. Comparison 4 PST versus Medication at 3-month follow-up, Outcome 3 Change in BDI score

		PST		Me	dicatio	n		Mean Difference	Mean Difference			
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	I IV, Random, 95% CI			
Mynors-Wallis 1995	-17.5	10.84	29	-14.4	10.5	27	48.9%	-3.10 [-8.69, 2.49]				
Mynors-Wallis 2000	-16.9	11.69	34	-18.4	11.16	34	51.1%	1.50 [-3.93, 6.93]				
Total (95% CI)			63			61	100.0%	-0.75 [-5.26, 3.76]	-			
Heterogeneity: Tau ² = Test for overall effect:	2.67; CH Z = 0.33	i ² = 1.3 (P = 0.	4, df = [.] 74)	1 (P = 0	.25); l² :	= 25%			-10 -5 0 5			

Analysis 4.4. Comparison 4 PST versus Medication at 3-month follow-up, Outcome 4 HSCL-D symptom resolution, 18-59 year-old

		PST		Me	dicatio	n		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean SD Total			Weight	IV, Random, 95% CI	IV, Random, 95% CI
4.4.1 week 0-2									
Barrett 2000	-0.36	0.54	80	-0.6	0.54	80	50.4%	0.24 [0.07, 0.41]	
Subtotal (95% CI)			80			80	50.4%	0.24 [0.07, 0.41]	-
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 2.81	(P=0	0.005)						
4.4.2 week 2-11									
Barrett 2000	-0.43	0.63	80	-0.28	0.54	80	49.6%	-0.15 [-0.33, 0.03]	
Subtotal (95% CI)			80			80	49.6%	-0.15 [-0.33, 0.03]	-
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 1.62	? (P = 0	0.11)						
Total (95% CI)			160			160	100.0%	0.05 [-0.34, 0.43]	
Heterogeneity: Tau ² =	0.07; Ch	ni² = 9.	57, df =	= 1 (P =	0.002)	; l² = 9	0%		0.5 0.25 0 0.25 0.5
Test for overall effect:	Z = 0.24	(P=0	0.81)						-0.5 -0.25 U U.25 U.5 Favoure PST Favoure Medic
Test for subgroup diffe	erences:	Chi² =	9.57, c	if = 1 (F	= 0.00	02), l² =	89.5%		

Analysis 4.5. Comparison 4 PST versus Medication at 3-month follow-up, Outcome 5 Change in HSCL-D score

		Me	dicatio	n		Mean Difference	Mean Difference		
Study or Subgroup	Mean	SD T	Fotal	Mean	SD	Total	Weight	IV, Random, 95% CI	IV. Random, 95% Cl
4.5.1 Minor Depressi	on & Dyst	thymia	a, 18-5	i9 year	-old				
Barrett 2000 Subtotal (95% CI)	-0.7	0.5	62 62	-0.95	0.63	61 61	36.7% 36.7%	0.25 [0.05, 0.45] 0.25 [0.05, 0.45]	
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 2.44 (P = 0.0	01)						
4.5.2 Dysthymia, 60-y	/ear and o	older							
Barrett 2000 Subtotal (95% CI)	-0.45	0.58	59 59	-0.7	0.7	50 50	29.6% 29.6%	0.25 [0.01, 0.49] 0.25 [0.01, 0.49]	
Heterogeneity: Not ap	plicable								
Test for overall effect:		P = 0.0	04)						
4.5.3 Minor Depressi	on, 60-ye	ar and	alder						
Barrett 2000	-0.52	0.57	49	-0.51	0.53	49	33.7%	-0.01 [-0.23, 0.21]	
Subtotal (95% CI)			49			49	33.7%	-0.01 [-0.23, 0.21]	
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 0.09 (P = 0.9	93)						
Total (95% CI)			170			160	100.0%	0.16 [-0.01, 0.33]	-
Heterogeneity: Tau ² =	0.01; Chi ²	= 3.63	3, df =	2 (P =	0.16);	l² = 459	%	-	-02 0.0102
Test for overall effect:	Z = 1.86 (P = 0.0	06)						Favours PST Favours Medicati

Analysis 4.8. Comparison 4 PST versus Medication at 3-month follow-up, Outcome 8 Change in SAS score

		PST		Med	dicatio	in		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean SD Total		Weight	IV, Random, 95% CI	IV, Random, 95% CI	
Mynors-Wallis 1995	-0.83	0.64	29	-0.59	0.53	27	51.2%	-0.24 [-0.55, 0.07]	
Mynors-Wallis 2000	-0.7	0.72	34	-0.9	0.69	34	48.8%	0.20 [-0.14, 0.54]	
Total (95% CI)			63			61	100.0%	-0.03 [-0.46, 0.41]	
Heterogeneity: Tau ² = Test for overall effect:	0.07: Cł Z = 0.12	ni² = 3. ? (P = 0	60_df= 0.91)	= 1 (P =	0.06),	⊮ = 72	%	-	-0.5 -0.25 0 0.25 0.5 Favours PST Favours Medicatio

5 116 - 117

Analysis 4.10. Comparison 4 PST versus Medication at 3-month follow-up, Outcome 10 Change in SF-36 Mental Health score

		PST		Me	dicatio	n		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% C	IV, Random, 95% CI
4.10.1 Minor Depress	sion & D	ysthym	ia, 18-	59 year	-old				
Barrett 2000 Subtotal (95% CI)	9.7	10.88	62 62	13.6	13,32	62 62	34.3% 34.3%	-3.90 [-8.18, 0.38] -3.90 [-8.18, 0.38]	-
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 1.79	9 (P = 0.	07)						
4.10.2 Dysthymia, 60	years a	nd olde	r						
Barrett 2000 Subtotal (95% CI)	5.5	9.75	61 61	8,9	11.83	50 50	37.5% 37.5%	-3.40 [-7.49, 0.69] - 3.40 [-7.49, 0.69]	
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 1.63	8 (P = 0.	10)						
4.10.3 Minor Depress	sion, 60	years a	nd old	er					
Barrett 2000 Subtotal (95% CI)	6.96	12.33	48 48	7.26	11	46 46	28.2% 28.2%	-0.30 [-5.02, 4.42] -0.30 [-5.02, 4.42]	
Heterogeneity: Not ap	plicable	0 - 0	0.01						
rest for overall effect.	Z - 0.12	2 (P - U.	90)						
Total (95% CI)			171			158	100.0%	-2.70 [-5.20, -0.19]	-
Heterogeneity: Tau ² =	0.00; CI	ni² = 1.4	1, df = 3	2 (P = 0	.49); l² =	= 0%			10 5 0 5 10
Test for overall effect:	Z = 2.11	(P = 0.	03)						Eavours PST Eavours Medicati
Test for subgroup diffe	erences:	Chi ² = 1	.41, df	= 2 (P :	= 0.49),	2 = 0%			accurs for Tarours medicat

Analysis 4.12. Comparison 4 PST versus Medication at 3-month follow-up, Outcome 12 All depression scales, changes

		PST		Med	dicatio	n		Std. Mean Difference	Std. Mean Difference		
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI		
4.12.1 Major depress	ion										
Mynors-Wallis 1995	-12.3	6.53	29	-11	6.8	27	14.3%	-0.19 [-0.72, 0.33]			
Mynors-Wallis 2000	-12	7.8	39	-14	6.66	36	17.7%	0.27 [-0.18, 0.73]			
Subtotal (95% CI)			68			63	32.0%	0.06 [-0.39, 0.51]			
Heterogeneity: Tau ² =	0.04; Ch	i² = 1.	71, df =	= 1 (P =	0.19);	² = 42	%				
Test for overall effect:	Z = 0.26	(P = (0.80)								
4.12.2 Minor Depress	sion & D	ysthyi	mia, 18	-59 yea	r-old						
Barrett 2000	-0.7	0.5	62	-0.95	0.63	61	24.2%	0.44 [0.08, 0.80]			
Subtotal (95% CI)			62			61	24.2%	0.44 [0.08, 0.80]			
Heterogeneity: Not ap	plicable										
Test for overall effect:	Z = 2.40	(P = (0.02)								
4.12.3 Dysthymia, 60	-year an	d alde	er								
Barrett 2000	-0.45	0.58	59	-0.7	0.7	50	22.5%	0.39 [0.01, 0.77]			
Subtotal (95% CI)			59			50	22.5%	0.39 [0.01, 0.77]			
Heterogeneity: Not ap	plicable										
Test for overall effect:	Z = 2.01	(P = 0	0.04)								
4.12.4 Minor Depress	sion, 60-	year a	nd old	er							
Barrett 2000	-0.52	0.57	49	-0.51	0.53	49	21.3%	-0.02 [-0.41, 0.38]			
Subtotal (95% CI)			49			49	21.3%	-0.02 [-0.41, 0.38]			
Heterogeneity: Not ap	plicable										
Test for overall effect:	Z = 0.09	(P = 0	0.93)								
Total (95% CI)			238			223	100.0%	0.21 [-0.02, 0.44]	-		
Heterogeneity: Tau ² =	0.02; Ch	i² = 5.	98, df =	= 4 (P =	0.20);	² = 33	%	-			
Test for overall effect:	Z = 1.81	(P = 0)	0.07)		,,				-U.5 -U.25 U 0.25 0.5		
		-							Favours For Favours Medicali		

7 PST versus Placebo at 3-month follow-up

Analysis 7.1. Comparison 7 PST versus Placebo at 3-month follow-up, Outcome 1 Change in HDRS score

			PI	acebo			Mean Difference	Mean Difference		
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI	
7.1.1 Minor Depressi	on & Dy	sthyrr	ia, 18-	59 year	-old					
Barrett 2000 Subtotal (95% CI)	-8.5	4.56	64 64	-7.2	5.52	69 69	37.3% 37.3%	-1.30 [-3.02, 0.42] -1.30 [-3.02, 0.42]	-	
Heterogeneity: Not ap	plicable									
Test for overall effect:	Z = 1.48	(P=0	0.14)							
7.1.2 Minor Depressi	on & Dy	sthyrr	ia, 60- <u>-</u>	yearsa	nd old	ег				
Barrett 2000 Subtotal (95% CI)	-5.8	5.19	112 112	-5.45	5.25	118 11 8	40.5% 40.5%	-0.35 [-1.70, 1.00] -0.35 [-1.70, 1.00]	-	
Heterogeneity: Not ap Test for overall effect:	plicable Z = 0.51	(P=0).61)							
7.1.3 Major Depressi	on								5	
Mynors-Wallis 1995 Subtotal (95% CI)	-12.3	6.53	29 29	-6.6	6.73	26 26	22.2% 22.2%	-5.70 [-9.21, -2.19] -5.70 [-9.21, -2.19]	-	
Heterogeneity: Not ap Test for overall effect:	plicable Z = 3.18	(P=0	0.001)							
Total (95% CI)			205			213	100.0%	-1.89 [-4.16, 0.38]	-	
Heterogeneity: Tau ² =	2.84; Ch	i² = 7.	82, df =	= 2 (P =	0.02);	² = 74	%		10 5 0 5	
Test for overall effect:	Z = 1.63	(P=(0.10)						Favours PST Favours Place	

Analysis 7.2. Comparison 7 PST versus Placebo at 3-month follow-up, Outcome 2 Recovered cases (HDRS<8)

	PST		Placel	00		Risk Ratio	Risk Ratio		
Study or Subgroup	Events	Total	Events	Total	Weight	M-H, Random, 95% Cl	M-H, Random, 95% Cl		
7.2.1 Major depression	n								
Mynors-Wallis 1995	18	30	8	30	14.3%	2.25 [1.16, 4.36]			
Subtotal (95% CI)		30		30	14.3%	2.25 [1.16, 4.36]			
Total events	18		8						
Heterogeneity: Not app	licable								
Test for overall effect: 2	z = 2.40 (F	P = 0.02	2)						
7.2.2 Dysthymia, 18-5	9 year-ol	d					1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		
Barrett 2000	16	43	18	42	18.7%	0.87 [0.52, 1.46]			
Subtotal (95% CI)		43		42	18.7%	0.87 [0.52, 1.46]			
Total events	16		18						
Heterogeneity: Not app	licable								
Test for overall effect: Z	z = 0.53 (F	□ = 0.60	0)						
7.2.3 Minor Depressio	n. 18-59 v	vear-ol	d						
Barrett 2000	17	37	20	39	20.9%	0.90 [0.56, 1.43]			
Subtotal (95% CI)		37	20	39	20.9%	0.90 [0.56, 1.43]	-		
Total events	17		20						
Heterogeneity: Not app	licable								
Test for overall effect: 2	z = 0.46 (F	□ = 0.64	4)						
7.2.4 Dysthymia, 60-y	ear and o	lder							
Barrett 2000	33	72	26	70	23.9%	1.23 [0.83, 1.83]			
Subtotal (95% CI)		72		70	23.9%	1.23 [0.83, 1.83]			
Total events	33		26						
Heterogeneity: Not app	licable								
Test for overall effect: 2	z = 1.04 (F	⊃ = 0.30	0)						
7.2.5 Minor Depressio	n, 60-yea	ar and o	der						
Barrett 2000	23	72	29	70	22.1%	0.77 [0.50, 1.19]			
Subtotal (95% CI)		72		70	22.1%	0.77 [0.50, 1.19]	-		
Total events	23		29						
Heterogeneity: Not app	licable								
Test for overall effect: 2	z = 1.17 (F	□ = 0.24	4)						
Total (95% CI)		254		251	100.0%	1.06 [0.77, 1.46]	+		
Total events	107		101						
Heterogeneity: Tau ² = ().07; Chi ²	= 8.61.	df = 4 (P	= 0.07	'); l² = 549	6	to to to to to		
Test for overall effect: Z	z = 0.37 (F	⊃ = 0.7 ⁺	1)		,,		0.2 0.5 1 2 5		
	0		·				FAVOUIS PIACEDO FAVOUIS PST		

5 118 - 119

Analysis 7.4. Comparison 7 PST versus Placebo at 3-month follow-up, Outcome 4 Change in HSCL-D score

		пст			acebo			Mean Difference	Mean Difference
Study or Subgroup	Mean	en	Total	Moan	en en	Total	Moight	IV Bandom 95% C	Wear Difference
7.4.4 Mines Departure		30		Riean	30	TOLAI	weigni	TV, Random, 95% C	1 1V, Randolli, 95% Cl
7.4.1 Minor Depressio	un a∉ ⊔y	sinym	ia, 10-5	be year	-010				
Barrett 2000	-0.7	0.5	62	-0.81	0.75	69	33.5%	0.11 [-0.11, 0.33]	
Subtotal (95% CI)			62			69	33.5%	0.11 [-0.11, 0.33]	
Heterogeneity: Not app	olicable								
Test for overall effect:	Z = 1.00	(P=0	.32)						
7.40.00	~~								
7.4.2 Minor depressio	оп, ви-у	ears a		er					and the second sec
Barrett 2000	-0.52	0.57	49	-0.42	0.54	54	33.9%	-0.10 [-0.31, 0.11]	
Subtotal (95% CI)			49			54	33.9%	-0.10 [-0.31, 0.11]	
Heterogeneity: Not app	olicable								
Test for overall effect:	Z = 0.91	(P=0	.36)						
7.4.3 Dysthymia, 60-y	/ears an	d olde	r						
Barrett 2000	-0.45	0.58	59	-0.45	0.63	58	32.5%	0.00 [-0.22, 0.22]	
Subtotal (95% CI)			59			58	32.5%	0.00 [-0.22, 0.22]	
Heterogeneity: Not ap	olicable								
Test for overall effect:	Z = 0.00	(P = 1	.00)						
Total (95% CI)			170			181	100.0%	0.00 [-0.12, 0.13]	-
Heterogeneity: Tau? =			22 df-	2/P-	0 40\-	12 - 094			+ + + + +
Telefogenelly. Tau	0.00, CI	u = 1.0	52, UI -	- 2 (F -	0.40),	= 0%			-0.5 -0.25 0 0.25 0.5
lest for overall effect:	∠ = 0.05	(P = U)	1.90)						Favours PST Favours Placebo

Analysis 7.6. Comparison 7 PST versus Placebo at 3-month follow-up, Outcome 6 HSCL-D symptom resolution, 18-59 year-old



Analysis 7.9. Comparison 7 PST versus Placebo at 3-month follow-up, Outcome 9 Change in SF-36 Mental Health score

		PST			Placebo			Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
7.9.1 Minor Depressi	on & Dy	sthymi	a, 18-5	9 year-o	bld				
Barrett 2000 Subtotal (95% CI)	9.7	10.88	62 62	11.3	11.78	69 69	32.0% 32.0%	-1.60 [-5.48, 2.28] -1.60 [-5.48, 2.28]	
Heterogeneity: Not ap	olicable								
Test for overall effect:	Z = 0.81	(P = 0.	42)						
7.9.2 Dysthymia, 60 y	rears an	d alder							
Barrett 2000 Subtotal (95% CI)	5.5	9.75	61 61	0.62	6.48	57 57	36.9% 36.9%	4.88 [1.91, 7.85] 4.88 [1.91, 7.85]	-
Heterogeneity: Not ap	olicable								
Test for overall effect:	Z = 3.22	! (P = 0.	001)						
7.9.3 Minor Depressi	on, 60 y	ears an	d olde	r					
Barrett 2000	6.96	12.33	48	5.33	7.83	53	31.0%	1.63 [-2.45, 5.71]	
Subtotal (95% CI)			48			53	31.0%	1.63 [-2.45, 5.71]	
Heterogeneity: Not app	olicable								
Test for overall effect:	Z = 0.78	(P = 0.	43)						
Total (95% CI)	0 24. 01	2 = 6 0	171	2 (D = 0	02) 12	179	100.0%	1.80 [-2.08, 5.68]	-
Test for overall effect:	0.31; U 7 = 0.01	(P = 0.07	8, at = . 36)	2 (P = 0	.03); 14 =	= / 1 %			-10 -5 0 5 10
Test for subgroup diffe	rences:	Chi ² = 6	5.88, df	= 2 (P :	= 0.03),	l² = 70.	9%		Favours PST Favours Placebo

Analysis 7.11. Comparison 7 PST versus Placebo at 3-month follow-up, Outcome 11 All depression scales, changes

		PST		Placebo				Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
7.11.1 Major Depress	sion								
Mynors-Wallis 1995 Subtotal (95% CI)	-12.3	6.53	29 29	-6.6	6.73	26 26	19.8% 19.8%	-0.85 [-1.40, -0.29] -0.85 [-1.40, -0.29]	
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 3.00) (P = (0.003)						
7.11.2 Minor Depress	sion & D	ysthy	nia, 18	-59 yea	r-old				
Barrett 2000 Subtotal (95% CI)	-0.7	0.5	62	-0.81	0.75	69	27.6%	0.17 [-0.17, 0.51]	
Subioral (95% CI)	- lia - h la		02			08	21.0/0	0.17 [-0.17, 0.51]	
Test for events Not ap		(D = (1 2 2 1						
rest for overall effect.	2 - 0.97	(P – (1.33)						
7.11.3 Minor depress	sion, 60-	years	and ol	der					
Barrett 2000	-0.52	0.57	49	-0.42	0.54	54	25.8%	-0.18 [-0.57, 0.21]	
Subtotal (95% CI)			49			54	25.8%	-0.18 [-0.57, 0.21]	-
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 0.91	(P = ().37)						
7.11.4 Dysthymia, 60	-vears a	nd ala	ler						
Barrett 2000	-0.45	0.58	59	-0.45	0.63	58	26.8%	0.00 [-0.36, 0.36]	_ <u>+</u>
Subtotal (95% CI)			59			58	26.8%	0.00 [-0.36, 0.36]	-
Heterogeneity: Not ap	plicable								
Test for overall effect:	Z = 0.00) (P = 1	.00)						
Total (95% CI)			199			207	100.0%	-0.17 [-0.53, 0.20]	-
Heterogeneity: Tau ² =	0.09; Cł	1 ² = 9.	82. df =	= 3 (P =	0.02):	2 = 69	%		1 1. 1. 1
Test for overall effect:	Z = 0.90) (P = (0.37)	- (,	- /,				-1 -0.5 0 0.5 1
									Favours PST Favours Placebo

9 PST versus Other Psychological Treatment at 3-month follow-up

Analysis 9.7. Comparison 9 PST versus Other Psychological Treatment at 3-month follow-up, Outcome 7 Change in CIS score

		PST		Other psy	/chol treat	ment			Mean Difference				
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% C	1	IV,	Random, 9	95% CI	
Kendrick 2005	-10.4	11.92	71	-10.1	12.15	62	34.1%	-0.30 [-4.40, 3.80]			- 10		
Liu 2007	-9.7	9.17	63	-8.6	8.46	77	65.9%	-1.10 [-4.05, 1.85]		_	-10		
Total (95% CI)			134			139	100.0%	-0.83 [-3.22, 1.57]			-		
Heterogeneity: Tau ² = Test for overall effect:	0.00; Ch Z = 0.68	ni² = 0.1 (P = 0.	0, df = 1 50)	I (P = 0.76);	I ² = 0%				-10	-5 Favour		5	1C psychol

Analysis 9.12. Comparison 9 PST versus Other Psychological Treatment at 3-month follow-up, Outcome 12 All depression scales, changes

	PST		Other psy	chol treat	ment	9	Std. Mean Difference	Std. Mean Difference
Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
ptoms								
-3.98	4.79	71	-3.97	4.24	62	35.2%	-0.00 [-0.34, 0.34]	
-9.7	9.17	63	-8.6	8.46	77	35.5%	-0.12 [-0.46, 0.21]	
		134			139	70.7%	-0.06 [-0.30, 0.17]	-
0.00;Ch	i² = 0.3	25, df=	1 (P = 0.61)	; I² = 0%				
2 = 0.53	(P = 0	.59)						
оп								
-2.6	7.04	32	-8.9	6.86	30	29.3%	0.89 [0.37, 1.42]	
		32			30	29.3%	0.89 [0.37, 1.42]	
licable								
2 = 3.35	(P = 0	.0008)						
		166			169	100.0%	0.22 [-0.31, 0.74]	
0.17; Ch	i ² = 10	.92, df	= 2 (P = 0.00	14); l ² = 82	%		-	1 15 1 15 1
2 = 0.81	(P = 0	.42)						Eavours PST Eavours Other psychol 1
	Mean ptoms -3.98 -9.7 0.00; Ch = 0.53 on -2.6 licable = 3.35 0.17; Ch = 0.81	PST <u>S</u> phoms <u>S</u> -3.38 4.79 -9.7 9.17 100: Chi ² = 0.3 (P = 0 0 on -2.6 7.04 licable := 3.35 (P = 0 1.17; Chi ² = 10 := 0.81 (P = 0	PST Mean SD Total ptoms -	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{tabular}{ c c c c c } \hline PST & Other psychol treatment \\ \hline Mean & SD & Total & Mean & SD & Total \\ \hline point \\ \hline -3.98 & 4.79 & 71 & -3.97 & 4.24 & 62 \\ -9.7 & 9.17 & 63 & -8.6 & 8.46 & 77 \\ 134 & -8.6 & 8.46 & 77 \\ 134 & -9.6 & -9.6 & -19.6 \\ \hline -9.6 & -9.5 & -9.5 & -10.6 & -10.6 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 30 \\ -2.6 & 7.04 & 32 & -8.9 & 6.86 & 30 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 30 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 30 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 30 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 30 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 30 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 30 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 30 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 10 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 10 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 10 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 10 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 10 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 10 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 10 \\ \hline -2.6 & 7.04 & 32 & -8.9 & 6.86 & 10 \\ \hline -2.6 & 7.04 & -9.0008 \\ \hline -2.6 & -$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	PST Other psychol treatment Std. Mean Difference Mean SD Total Weight IV. Random, 95%, CI -3.98 A.79 71 -3.97 4.24 62 35.2% -0.00 [-0.34, 0.34] -9.7 9.17 63 -8.6 8.46 77 35.5% -0.05 [-0.30, 0.17] 1.00; Ch ^{ip} = 0.25, df = 1 (P = 0.61); P = 0% = = -0.06 [-0.30, 0.17] = 0.53 (P = 0.50) = = = -0.06 [-0.30, 0.17] on - - 32 -8.8 6.86 30 29.3% 0.89 [0.37, 1.42] clobe - - 32 -8.8 6.86 30 29.3% 0.89 [0.37, 1.42] ilcable - - - 166 169 100.0% 0.22 [-0.31, 0.74] 1.17; Ch ^{ip} = 10.92; df = 2 (P = 0.004); P = 82% - 169 100.0% 0.22 [-0.31, 0.74] -

Analysis 9.13. Comparison 9 PST versus Other Psychological Treatment at 3-month follow-up, Outcome 13 All quality of life scales, changes

		PST		Otherpsy	ychol treat	ment	1	Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
9.13.1 Emotional sym	ptoms								
Kendrick 2005	0.17	0.28	71	0.12	0.25	62	48,8%	0.19 [-0.15, 0.53]	
Liu 2007	19.2	21.81	63	17.8	21.86	77	51.2%	0.06 [-0.27, 0.40]	
Subtotal (95% CI)			134			139	100.0%	0.12 [-0.11, 0.36]	
Heterogeneity: Tau ² =	0.00; CI	ni² = 0.2	5,df=1	(P = 0.61);	² = 0%				
Test for overall effect:	Z = 1.02	(P=0.	31)						
Total (95% CI)			134			139	100.0%	0.12 [-0.11, 0.36]	-
Heterogeneity: Tau ² =	0.00; CI	ni² = 0.2	5,df=1	(P = 0.61);	l² = 0%			÷	
Test for overall effect:	Z = 1.02	(P=0.	31)					Favours Oth	ter psychol treatment Favours PST

10 PST versus Other Psychological Treatment at 6-month follow-up

Analysis 10.12. Comparison 10 PST versus Other Psychological Treatment at 6-month follow-up, Outcome 12 All depression scales, changes



Analysis 10.13. Comparison 10 PST versus Other Psychological Treatment at 6-month follow-up, Outcome 13 All quality of life scales, changes

		PST		Other psychol treatment		9	Std. Mean Difference	Std. Mean Difference				
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI		IV, Randon	1, 95% CI	
10.13.1 Emotional sy	mptoms	:										
Kendrick 2005 Subtotal (95% CI)	0.18	0.29	71 71	0.15	0.25	62 62	43.5% 43.5%	0.11 [-0.23, 0.45] 0.11 [-0.23, 0.45]				-
Heterogeneity: Not ap	plicable											
Test for overall effect:	Z = 0.63	(P = 0.	53)									
10.13.2 Depressive d	isorders	6										
Dowrick 2000 Subtotal (95% CI)	19.11	22.15	89 89	16.56	21.13	83 83	56.5% 56.5%	0.12 [-0.18, 0.42] 0.12 [-0.18, 0.42]			-	-
Heterogeneity:Notap Test for overall effect:	plicable Z = 0.77	(P = 0.	44)									
Total (95% CI)			160			145	100.0%	0.11 [-0.11, 0.34]		-	-	
Heterogeneity: Tau ² =	0.00; Ch	ni² = 0.01),df=1	(P = 0.97);	l² = 0%				0.5	0.05	0.05	
Test for overall effect:	Z = 0.99	(P = 0.	32)					Favours	-u.o Otherpsyci	-u.25 U holtreatment F	0.25 Favours PST	0.5

13 Days off work & Costs, PST versus Usual Care

Analysis 13.1. Comparison 13 Days off work & Costs, PST versus Usual Care, Outcome 1 Total number of days off work, 6-month follow-up (Emotional symptoms)

		PST		Us	ual Car	е		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
Kendrick 2005	13.8	27.6	51	10.7	18.18	71	50.4%	3.10 [-5.58, 11.78]	
Mynors-Wallis 1997	4.35	10.35	40	16.23	23.82	30	49.6%	-11.88 [-20.99, -2.77]	
Total (95% CI)			91			101	100.0%	-4.32 [-19.00, 10.36]	-
Heterogeneity: Tau ² =	91.61; C	hi² = 5.4	45, df =	: 1 (P =	0.02); l ^a	= 82%		-	
Test for overall effect:	Z = 0.58	(P = 0.	56)						Favours PST Favours Usual Ca

15 Health care use & Costs, PST versus Usual Care

Analysis 15.2. Comparison 15 Health care use & Costs, PST versus Usual Care, Outcome 2 Total number of GP consultations, 6-month follow-up (Emotional symptoms)



Analysis 15.4. Comparison 15 Health care use & Costs, PST versus Usual Care, Outcome 4 Total costs of GP consultations, 6-month follow-up (Emotional symptoms)

	1	PST	Usual Care			е	Mean Difference		Mean Difference				
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI		IV, Ran	dom,	95% CI	
Kendrick 2005	56	44	51	91	76	71	51.0%	-35.00 [-56.41, -13.59]		-			
Mynors-Wallis 1997	63.9	52	40	65.2	46.9	30	49.0%	-1.30 [-24.57, 21.97]		_	-	_	
Total (95% CI)			91			101	100.0%	-18.47 [-51.49, 14.55]	-		-		
Heterogeneity: Tau ² = 437.73; Chi ² = 4.36, df = 1 (P = 0.04); l ² = 77% Test for overall effect: Z = 1.10 (P = 0.27)								-50 Favo	-25 ours PS	0 T Fa	25 vours U	50 sual Ca	

Analysis 15.6. Comparison 15 Health care use & Costs, PST versus Usual Care, Outcome 6 Total Medications costs, 6-month follow-up (Emotional symptoms)



Analysis 15.8. Comparison 15 Health care use & Costs, PST versus Usual Care, Outcome 8 Total costs, 6-month follow-up (Emotional symptoms)

		PST		Usu	al Ca	е		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
Kendrick 2005	631	501	51	316	327	71	45.7%	315.00 [157.86, 472.14]	-8
Mynors-Wallis 1997	132	55.3	40	86.6	54.9	30	54.3%	45.40 [19.33, 71.47]	
Total (95% CI)			91			101	100.0%	168.61 [-94.62, 431.83]	
$\begin{array}{l} \text{Total (35.6 cf)} & \text{Total (35.6 cf)} \\ \text{Heterogeneity: Tau' = 33039.81; Chi' = 11.01, df = 1 (P = 0.0009); P = 91\% \\ \text{Test for overall effect: } Z = 1.26 (P = 0.21) \end{array}$								-500 -250 0 250 500 Favours PST Favours Usual Car	

Appendices

1 CCDAN-CTR search strategy

August 2009: 1"Cochrane Depression, Anxiety and Neurosis Group".qc. 2. problem solving.mp. [mp=title, original title, abstract, mesh headings, heading words, keyword] 3. problem solving therapy.mp. [mp=title, original title, abstract, mesh headings, heading words, keyword] 4. problem solving treatment.mp. [mp=title, original title, abstract, mesh headings, heading words, keyword] 5. pst\$.mp. [mp=title, original title, abstract, mesh headings, heading words, keyword 6.2 or 3 or 4 or 5 7.1 and 6

2 MEDLINE search strategy

1950-August 2009:

- 1. randomised controlled trial.pt.
- 2. controlled clinical trial.pt.
- 3. randomised controlled trials/
- 4. random allocation/
- 5. Double-Blind Method/
- 6. single blind method/
- 7. clinical trial.pt.
- 8. clinical trials/
- 9. (clin\$ adj25 trial\$).ti,ab.

10. ((singl\$ or doubl\$ or trebl\$ or tripl\$) adj25 (blind\$ or mask\$)).ti,ab.

- 11. placebos/
- 12. placebo\$.ti.
- 13. placebo\$.ab.
- 14. random\$.ti.
- 15. random\$.ab.

- 16. research design/
- 17. comparative study/
- 18. evaluation studies/
- 19. follow up studies/
- 20. prospective studies/
- 21. control\$.ti.
- 22. control\$.ab.
- 23. prospectiv\$.ti.
- 24. prospectiv\$.ab.
- 25. volunteer\$.ti.
- 26. volunteer\$.ab.
- 27. 1 or 2 or 3 or 4 or 5 or 5 or 7 or 8
- or 9 or 10 or 11 or 12 or 13 or 14 or
- $15 \mbox{ or } 16 \mbox{ or } 17 \mbox{ or } 18 \mbox{ or } 19 \mbox{ or } 20 \mbox{ or } 21$
- or 22 or 23 or 24 or 25 or 26
- 28. (animals not human).sh.
- 29. 27 not 28
- 30. problem solving.mp. and treatment outcome/
- 31. problem solving therapy.mp. [mp=title, original title, abstract, name of substance, mesh subject heading] 32. problem solving treatment.mp. [mp=title, original title, abstract, name of substance, mesh subject heading] 33. pst\$.mp. [mp=title, original title, abstract, name of substance, mesh subject heading]
- 34.30 or 31 or 32 or 33
- 35. exp "behaviour and behaviour mechanisms"/ or exp "psychological phenomena and processes"/ or exp mental disorders/ or exp "behavioural disciplines and activities"/ 36. 35 and 34 and 29

3 EMBASE search strategy

- 1980-August 2009:
- 1.Randomised Controlled Trial/
- 2. Clinical Trial/
- 3. Randomization/
- 4. Double Blind Procedure/
- 5. Single Blind Procedure/
- 6. Major Clinical Study/
- 7. Controlled Study/

- 8. Multicenter Study/ 9. Placebo/ 10. placebo\$.ti. 11. placebo\$.ab. 12 random\$ti 13. random\$.ab. 14. Methodoloav/ 15. Evidence Based Medicine/ 16. Comparative Study/ 17. exp "evaluation and follow up"/ 18. Follow Up/ 19. Prospective Study/ 20. control\$.ti. 21. control\$.ab. 22. prospectiv\$.ti. 23. prospectiv\$.ab. 24. volunteer\$.ti. 25. volunteer\$.ab. 26. ((sinal\$ or doubl\$ or trebl\$ or tripl\$) adj (blind\$ or mask\$ or dummy)).ti,ab. 27. (clinic\$ adj (trial\$ or study or studies\$)).ti.ab. 28. 1 or 2 or 3 or 4 or 5 or 5 or 7 or 8 or 9 or 10 or 11 or 12 or 13 or 14 or 15 or 16 or 17 or 18 or 19 or 20 or 21 or 22 or 23 or 24 or 25 or 26 or 27 29. human.de. 30. nonhuman.de. 31.29 and 30 32.30 not 31 33 28 not 32 34. (problem solving.mp.) and (treatment outcome/) 35. problem solving therapy.mp. [mp=title, abstract, subject headings, drug trade name, original title, device manufacturer, drug manufacturer name] 36. problem solving treatment.mp. [mp=title, abstract, subject headings, drug trade name, original title, device manufacturer, drug manufacturer name] 37. pst\$.mp. [mp=title, abstract, subject headings, drug trade name,
- original title, device manufacturer, drug manufacturer namel
- 38. 34 or 35 or 36 or 37

39. (exp mental function/) or (exp
"psychological and psychosocial
phenomena"/) or (exp mental disease/)
or (exp "psychological and psychiatric
procedures, techniques and concepts"/)
40. 39 and 38 and 33

4 CINAHL search strategy

1982-August 2009:

- 1. (MH "Random Assignment")
- 2. (MH "Double-Blind Studies")
- 3. (MH "Single-Blind Studies")
- 4. PT clinical trial
- 5. (MH "Clinical Trials+")
- 6. TI clin* W25 trial*
- 7. AB clin* W25 trial*
- 8. (singl* or doubl* or tripl* or trebl*)

AND (blind* or mask* or dummy*)

- 9. (MH "Placebos")
- 10. Tl placebo*
- 11. AB placebo*
- 12. Tl random*
- 13. AB random*
- 14. (MH "Comparative Studies")
- 15. (MH "Evaluation Research+")
- 16. (MH "Prospective Studies+")
- 17. TI control*
- 18. AB control*
- 19. Tl prospectiv*
- 20. AB prospectiv*
- 21. Tl volunteer*
- 22. AB volunteer*
- 23. 1 or 2 or 3 or 4 or 5 or 6 or 7 or 8
- or 9 or 10 or 11 or 12 or 13 or 14 or
- 15 or 16 or 17 or 18 or 19 or 20 or 21 or 22
- 24. pst*
- 25. (MH "Problem Solving+")
- 26. problem solving therapy
- 27. problem solving treatment
- 28. 24 or 25 or 26 or 27
- 29. (MH "Mental Disorders+")
- 30. (MH "Behavior and Behavior

Mechanisms+")

31. (MH "Psychological Processes and

Principles+") 32. 29 or 30 or 31 33. 32 and 28 and 23

5 PsycINFO search strategy

1806-August 2009:

problem AND solving (Limiters: Population Group: Human; Methodology: Treatment Outcome/ Randomised Clinical Trial)

6 126 - 127

6 Effectiveness of problem-solving treatment by general practice registrars for patients with emotional symptoms: a controlled trial

Lieke J.A. Hassink-Franke Evelyn M. van Weel-Baumgarten Eric Wierda Maike W.M. Engelen Mechtild M.L. Beek Hans H.J. Bor Henk J.M. van den Hoogen Peter L.B.J. Lucassen Chris van Weel

Submitted for publication

Presented at: NAPCRG Annual Meeting 2009, Montreal, Canada

6 128 - 129

Abstract

Introduction

In general practice many patients present with emotional symptoms. Both patients and physicians desire effective non-pharmacological treatments. To study the effectiveness of problemsolving treatment (PST) delivered by trained general practice (GP) registrars for patients with emotional symptoms.

Methods

In a controlled clinical trial we compared the effectiveness of PST versus usual care for patients with emotional symptoms. Dutch GP registrars provided PST or usual care, according to their own preference. Patients were included if they (a) had presented for three or more consultations with emotional symptoms in the past 6 months; and (b) scored four or more on the 12-item General Health Questionnaire. Outcomes at 3 and 9 month follow-up were standard measures of depression, anxiety and quality of life.

Results

Thirty-eight GP registrars provided PST and included 98 patients; 43 provided usual care and included 104 patients. PST patients improved significantly more than usual care patients: at 9-month follow-up recovery rates for somatoform disorder and anxiety were higher in the PST group (OR 6.50, p=0.01 respectively OR 11.25, p=0.03). PST patients had improved significantly more on the domains social functioning, role limitation due to emotional problems and general health perception.

Discussion

Patients with emotional symptoms improved significantly more after PST delivered by motivated GP registrars than after usual care by GP registrars. Further research, with randomization of interested registrars or interested GPs, is needed.

Introduction

In general practice many patients have emotional symptoms and/or psychosocial problems.^{1,2} Most patients are treated adequately, but in a minority of cases a pattern of recurrent or chronic symptoms develops with a negative impact on guality of life³ and frequent consultations.⁴ This makes diagnosis and treatment of emotional symptoms an important task in general practice. General practicioners (GPs) often prescribe medication, usually benzodiazepines or antidepressants² but medication is not always appropriate. It has important side effects, 5,6 patient adherence is low7 and the effectiveness of antidepressants is being disputed.⁸ Alternative approaches have to be considered. This looks attractive as most patients prefer non-pharmacological treatments.⁹ Counseling is nearly always part of the treatment in general practice¹⁰ and has the potential to strengthen patients' self-management. However, its content often varies and evidence for its longterm effectiveness is weak.¹¹ Problem-solving treatment (PST) might be an attractive option because of its

structured approach with a focus on patient-empowerment.¹² PST is a brief psychological intervention suitable for primary care, focusing on how to deal with everyday problems. PST is effective in anxiety and depression, especially in major depression.¹²⁻¹⁴and there are indications that it is effective for unexplained physical symptoms¹⁵ and in palliative care.¹⁶ A recent Cochrane review recommended further research on the effectiveness of PST in patients with emotional symptoms. irrespective of whether these fulfil the criteria for DSM-IV disorders.¹⁷ Concurrently, GPs and GP registrars have expressed the need for an effective psychological treatment they can deliver themselves to manage patients with emotional symptoms.18-²⁰ Training GP registrars in PST could meet GPs' need in an early career stage. A pilot study with 11 GP registrars showed that registrars can be trained successfully in PST but the authors recommended further investigation with a larger sample of registrars and evaluation of patient outcomes.²¹ We aimed to study the effectiveness of PST delivered by trained GP registrars

Figure 1 Problem-solving treatment (PST)

A brief psychological treatment with 7 stages:

- 1. Explanation and rationale
- 2. Clarification and definition of the problems
- 3. Establishing achievable goals
- 4. Generating solutions
- 5. Selecting preferred solution
- 6. Implementing solution
- Evaluation of progress

6 130 - 131

for patients with recurrent or chronic emotional symptoms.

Methods

Design

We compared, in a pragmatic controlled clinical trial, the effectiveness of PST versus usual care for patients with emotional symptoms. PST and usual care were applied by GP registrars. This design promotes external validity, which means that it increases the applicability of a trial's results to situations other than the trial situation.²²

Setting

The study took place in a Dutch three-year GP residency programme. From 2003 to 2005 the residency programme scheduled the participation of all third-year registrars (81) in this study as part of the core programme. Registrars participated in two groups, PST and 'usual care'. Initially, we assigned registrars randomly to PST or usual care. We had to change this selection as registrars who were uncomfortable with PST did not include any patients. We allowed the next year group (2004-2005) to choose the strategy they were most comfortable with: PST (including training) or usual care. Ethical approval was obtained according to local protocols.

Recruitment and selection criteria We asked registrars to recruit adult patients who presented emotional symptoms, during their regular clinical work in their training practice (September 2003 to April 2006). We asked each registrar to recruit 4-6 patients because, from a logistical perspective, this was regarded as the

maximum feasible number within one year of residency. We defined emotional symptoms as sub threshold as well as formal disorders of depressed mood. anxiety or stress, and psychosocial problems. Patients were included in the study, if they (a) had presented emotional symptoms during three or more consultations in the past 6 months: and (b) had a score of four or more on the 12-item General Health Ouestionnaire (GHO-12).23 Exclusion criteria were (a) severe physical disease; (b) severe mental morbidity (organic psychiatric disorder, substance misuse, active suicidal ideas); (c) current or recent (past year) psychiatric or psychological treatment or cognitive behavioural therapy; (d) insufficient mastery of Dutch language. Registrars received support of a research assistant in the selection of suitable patients. All participating patients signed informed consent.

Treatment and training

PST is a brief psychological treatment, derived from cognitive behavioural therapy, teaching patients how to use their own skills to cope with everyday life problems in a systematic way. It is assumed that symptoms reduce if control over problems is (re)gained.¹² PST comprises seven stages (Figure 1). The treatment consists of four to six consultations over a period of approximately 8-12 weeks with a duration of no more than 30 minutes, except for the first session which may last 60 minutes.

The registrars were trained by experienced PST trainers in a twoday course, followed by supervised treatment and feedback meetings. Trainers assessed the quality of PST through registrars' PST work sheets. Details about the feasibility of this training programme during residency were published before.²⁴ The exact nature of 'usual care' was retrieved from patient records after the trial. Both treatment groups were allowed to prescribe medication.

Follow-up and outcomes

Primary outcomes were the proportion of patients who remitted, the reduction of symptoms, and improvement of quality of life. We used the Primary Health Questionnaire (PHQ) assessing the presence of five DSM-IV disorders,²⁵ the Hospital Anxiety

and Depression Scale (HADS)²⁶. the 36-item MOS short form (SF-36)²⁷ and the 5-dimension Eurogol measuring quality of life (EQ-5D)²⁸, and the social problem-solving (skills) inventory-revised measuring problemsolving skills (SPSI-R)²⁹. Secondary outcomes were: patient satisfaction (a self developed questionnaire based on the Consultation Satisfaction Questionnaire³⁰ with 9 items measuring satisfaction with the doctor and 7 items measuring satisfaction with the treatment); number of disability days (TiC-P)³¹; and health care utilization. Health care utilization data were

Figure 2 Flow chart PST-trial



collected from the patients' records: data on referrals and medication, and numbers of contacts with the GP. Higher scale scores indicated better patient outcome, except for the PHQ and HADS with lower scores indicating better scores.

Participants received self-completing questionnaires at baseline (TO), after treatment (at 3 months, T1) and at 9month follow-up (T2). Record data of the six-month period before treatment were compared to data of the six-month period after treatment.

Sample size

We aimed to detect a clinically relevant difference of 30% between interventions with the primary outcome measure PHQ. To provide a power of 80% at a two-sided 5% level of significance, we needed 42 patients with full data in each group.

Analysis of effectiveness We conducted statistical analyses using SPSS 16.0, according to the intention-to-treat principle. We analysed all cases with data at baseline and data at T1 and/or T2. Subsequently, we repeated the analysis

Table 1 Baseline characteristics

Patient characteristics	PST (n=98)	Usual care (n=104)	p-value
Mean age , years (SD)	40.3 (13.3)	46.0 (16.3)	0.01
Female, No. (%)	79/98 (81%)	63/104 (61%)	< 0.001
Married or living with partner, No. (%)	62/92 (67%)	51/79 (65%)	0.98
Ethnicity, white, No. (%)	83/91 (91%)	95/98 (97%)	0.09
Paid employment, No. (%)	61/92 (66%)	51/86 (59%)	0.33
Clinical characteristics	(n=92)	(n=86)	
GHQ-12, mean (SD)	9.30 (2.37)	7.58 (2.56)	<0.001
PHQ somatoform disorder, No. (%)	37 (40%)	31 (36%)	0.57
PHQ major depressive syndrome, No. (%)	37 (40%)	25 (29%)	0.12
PHQ other anxiety syndrome, No. (%)	32 (35%)	18 (21%)	0.04
HADS depression score, mean (SD)	9.26 (4.25)	8.09 (4.60)	0.05
HADS anxiety score, mean (SD)	10.41 (3.28)	9.14 (4.81)	0.08
SF-36 social functioning, mean (SD)	47.64 (19.20)	51.49 (23.85)	0.29
SF-36 mental health, mean (SD)	43.51 (13.56)	49.18 (20.28)	0.05
SF-36 role limitation due to	24.20 (30.05)	46.48 (41.59)	<0.001
emotional problems, mean (SD)			
SF-36 general health perception, mean (SD)	55.12 (19.45)	55.68 (19.94)	0.87
EQ-5D score, mean (SD)	0.69 (0.16)	0.71 (0.19)	0.60
SPSI-R total score, mean (SD)	9.07 (2.64)	9.91 (2.72)	0.04

with missing data being imputed by the last observations carried forward (LOCF) principle.

We compared differences within the treatment groups with McNemar tests and paired t-tests to assess changes over time. In order to investigate the effect of the intervention, we used univariate general linear models and binary logistic regression using gender, age and baseline values as covariates to correct for baseline differences between treatment groups. We separately analysed the effect of treatment at T1 and T2. The effect of the intervention was the difference in outcome between the PST group and the usual care group (level of significance p < 0.05).

Results

Recruitment & follow-up Thirty-eight registrars (28 women) provided PST and 43 (29 women) provided usual care. They included 202 patients: 98 in the PST group and 104 in the usual care group (Figure 2). Patients in the PST group were significantly younger and more often female, had at baseline significantly higher symptom severity and

Table 2Numbers of cases at Patient Health Questionnaire (PHQ) at 3-month follow-up (T1) and9-month follow-up (T2) and binary logistic regression for differences in effects between PST andUsual care at T1 and T2 compared to baseline (with gender, age and baseline values as covariates).

No of PST	Cases (%) Usual Care	Odds Ratio (95% CI) for achievement of remission	
37/92 (40.2%)	31/86 (36.0%)	ref	
6/63 (9.5%)	20/65 (30.8%)	6.52 (1.94 to 21.91)*	
4/60 (6.7%)	16/63 (25.4%)	6.50 (1.74 to 24.31)*	
37/92 (40.2%)	25/86 (29.1%)	ref	
8/63 (12.7%)	11/65 (16.9%)	1.90 (0.61 to 5.92)*	
5/60 (8.3%)	4/63 (6.3%)	0.62 (0.14 to 2.76)*	
32/92(34.8%)	18/86 (20.9%)	ref	
6/63 (9.5%)	8/65 (12.3%)	2.02 (0.56 to 7.31)*	
1/60 (1.7%)	7/63 (11.1%)	11.25 (1.21 to 104.26)*	
	No of PST 37/92 (40.2%) 6/63 (9.5%) 4/60 (6.7%) 37/92 (40.2%) 8/63 (12.7%) 5/60 (8.3%) 32/92 (34.8%) 6/63 (9.5%) 1/60 (1.7%)	No of Cases (%) PST Usual Care 37/92 (40.2%) 31/86 (36.0%) 6/63 (9.5%) 20/65 (30.8%) 4/60 (6.7%) 16/63 (25.4%) 37/92 (40.2%) 25/86 (29.1%) 8/63 (12.7%) 11/65 (16.9%) 5/60 (8.3%) 4/63 (6.3%) 32/92 (34.8%) 18/86 (20.9%) 6/63 (9.5%) 8/65 (12.3%) 1/60 (1.7%) 7/63 (11.1%)	

PST = problem-solving treatment; CI = confidence interval; ref = reference group; * adjusted for gender, age and baseline values

6 134 - 135

significantly worse SPSI-scores than patients in the control group (Table 1). Overall, 128 (63%) participants returned follow-up questionnaires at T1 and 123 (61%) at T2. Patients lost to follow-up did not differ significantly from those who completed the study with regard to age, gender, PST or usual care, or baseline values. Recorded reasons for loss to follow-up were: no more interest/time (n=7), psychosocial problems (n=6), aggravation of psychological symptoms (n=5), physical complaints (n=4), being moved (n=2), language (n=1). Medical records were retrieved for 96 PST patients and 99 control group patients (1 patient died and 6 patients moved).

Clinical outcome and quality of life Both treatment groups improved significantly over time. Tables 2 and 3 show the results at T1 and T2 compared to TO. From the PHQ we analysed the three most prevalent disorders: major depression (n=62), somatoform disorder (n=68), and other anxiety syndrome (n=50) (Table 1). The PST group showed significantly better recovery rates for somatoform disorder at T1 and T2 and for anxiety at T2. but not for major depression (Table 2). The HADS depression score improved significantly more in the PST group than in the usual care group at T1; the HADS anxiety score did so at T1 and T2. In the PST group general health perception improved significantly more at T1 and T2. In table 3 we present the three SF-36 domains most relevant to a mental health oriented intervention: social function and mental role limitation improved significantly more in the PST group at T2; mental health improved in both groups but did not differ significantly between groups at

T1 nor T2. The EO-5D scale improved more in the PST group at T1. Regarding problem-solving skills, the SPSI-R total scores did not show significant differences in change between groups. When analyzing only those PST patients who attended 4 or more sessions - the number recommended for effectiveness - the differences were more pronounced and the HADS depression score differed significantly in favour of PST at T2 too (p=0.020). There were no significant differences between patients recruited during the first year and the second year when the randomization was released. After imputing missing data by LOCF both quality of life and psychological symptom scores differed significantly in favour of PST.

Health care utilization & disability days The numbers of patients being referred or using psychotropic medication during treatment did not differ significantly between groups. The changes in consultation rate and numbers of patients being referred or using psychotropic medication in the six months before versus the six months after treatment were not significantly different from usual care but all in favour of PST.

Absence of work did not differ significantly (TiC-P). PST patients experienced significantly less difficulties during unpaid activities than usual care patients (T1 B-coefficient -0.90, p<0.001; T2 B-coefficient -0.46, p=0.04).

Patient satisfaction

No significant differences in patient satisfaction were found. At T1, PST patients scored 25.7 (SD 5.0) and usual care patients 24.8 (SD 5.8) on the 45-point 'satisfaction with physician' scale and 20.5 (SD 4.3) and 19.4 (SD 4.6) respectively on the 35point 'satisfaction with treatment' scale.

Treatment received

Patients in the intervention group received on average 4.3 PST sessions (range 1-7), including the consultation of study inclusion accompanied by the intake of PST. Fifty-three patients completed treatment (≥ 4 sessions) with on average 5.3 sessions (range 4-7). The mean number of consultations in addition to PST sessions during the 3-month treatment period was 0.7 (range 0-5). Based on work sheets of PST sessions the PST supervisor reported good quality performance of PST.

In the usual care group, the average number of consultations – for any reason – during the 3-month treatment period was 3.3 (range 1-12). Most registrars used counseling in most consultations, but content and duration were not described in the records.

Discussion

Both PST and usual care patients with chronic or recurrent emotional symptoms improved significantly over time. However, patients who were treated by GP registrars providing PST, had significant better outcomes than patients who were treated by registrars providing usual care. PST patients reported significantly fewer symptoms of depression and anxiety. and a significantly higher general health perception than usual care patients did, both short- and long-term. PST patients had a significantly higher chance of recovery from somatoform disorder short-term. Long-term this was also the case for anxiety. Long-term they

reported significantly better scores of social function and role limitation due to emotional problems, corresponding with a relevantly better subjective function in daily life. This fits in with earlier research that found significantly better improvements of quality of life after PST than after usual care.14 The PHQ did not show significant differences in recoveries of major depressive disorder but significant differences in favour of PST were found on the HADS concerning the severity of anxiety and depressive symptoms. Thus, PST did not diminish the number of *cases* of depression but reduced symptom severity. Strikingly, SPSI scores did not differ significantly between aroups. This could be due to the fact that actual problem-solving performance is not necessarily a function of cognitive- behavioural skills in generating solutions.32

Strengths and limitations of the study Notwithstanding these positive findings, this study has limitations. Firstly, the lack of randomization of registrars providing PST. The registrars who were initially randomised to PST and were uncomfortable with it. did not recruit patients not allowing any comparison. We realize that changing the selection of registrars by offering the choice between PST and usual care may have resulted in potentially overestimating the impact of PST, because PST registrars were probably more motivated to deliver mental health care than their usual care colleagues. However, there were no differences in outcome between patients of registrars randomly allocated to PST and patients of registrars who made a choice for it. Also, registrars in the usual care group were not necessarily unmotivated for

6 136 - 137

		Means	score (SD)	Mean difference (95% CI) between PST and
Outcome#		PST*	Usual care*	B-coefficient
Hospital Anxiety and De	epression Scal	e		
Depression score				
	ТО	9.26 (4.25)	8.09 (4.60)	ref
	T1	4.97 (4.12)	6.73 (4.40)	-1.88 (-3.11 to -0.64)∽
	T2	4.55 (4.45)	5.47 (3.65)	-1.23 (-2.49 to 0.02)~
Anxiety score				
	ТО	10.41 (3.28)	9.14 (4.81)	ref
	T1	6.98 (3.94)	8.33 (4.80)	-2.17 (-3.44 to -0.90)~
	T2	6.65 (3.25)	7.05 (4.93)	-1.33 (-2.50 to -0.15)~
SF-36				
Social functioning				
-	ТО	47.64 (19.20)	51.49 (23.85)	ref
	T1	65.97 (23.42)	57.33 (25.73)	5.81 (-3.73 to 15.36)~
	T2	73.23 (18.30)	62.80 (23.45)	9.83 (1.27 to 18.39)~
Mental health				
	ТО	43.51 (13.56)	49.18 (20.28)	ref
	T1	64.25 (18.16)	61.36 (21.37)	3.69 (-3.13 to 10.51)~
	T2	68.00 (17.06)	66.26 (18.64)	4.49 (-2.13 to 11.12)~
Role limitation due to e	emotional probl	ems		
	ТО	24.20 (30.05)	46.48 (41.59)	ref
	T1	63.89 (42.85)	56.67 (44.80)	9.58 (-8.44 to 27.60)~
	T2	80.30 (36.17)	70.29 (42.30)	17.18 (0.69 to 33.67)~
General health percep	tion			
	ТО	55.12 (19.45)	55.68 (19.94)	ref
	T1	64.69 (22.99)	57.94 (19.98)	8.00 (1.81 to 14.20)~
	T2	70.39 (19.48)	61.48 (20.27)	10.48 (3.66 to 17.30)~
EuroQol-5D				
	ТО	0.69 (0.16)	0.71 (0.19)	ref
	T1	0.81 (0.17)	0.71 (0.16)	0.09 (0.03 to 0.14)~
	T2	0.82 (0.18)	0.79 (0.14)	0.02 (-0.03 to 0.07)~
Social Problem Solving	Inventory - rev	ised		
	ТО	9.07 (2.64)	9.91 (2.72)	ref
	T1	9.98 (2.68)	10.57 (2.49)	-0.49 (-1.20 to 0.21)~
	T2	10.13 (2.58)	10.86 (2.52)	-0.13 (-0.87 to 0.60)~

Table 3. Mean scores on main outcome scales at 3-month follow-up (T1) and 9-month follow-up (T2) and ANCOVA for differences in effects over time between PST and Usual care (with gender, age and baseline values as covariates).

PST = problem-solving treatment; SD = standard deviation; Cl = confidence interval; #HADS=21-point scales; SF-36=100-point scales; EQ-5D=1-point scale; SPSI-R=20-point scale; *PST group: at T0 n=92; at T1 n=63; at T2 n=60. Usual care group: at T0 n=86; at T1 n=65; at T2 n=63.; ~Adjusted for gender, age and baseline values

mental health care: they mentioned 'time investment' most often as a reason for not participating in PST training (78% of registrars). They received in their programme training of the prevailing Dutch College guidelines of depression and anxiety.^{33,34} Furthermore, patient satisfaction with their GP registrar treatment was not significantly different between treatment groups.

The second limitation was the selection of patients, with registrars in the intervention group selecting patients mainly themselves. Registrars in the usual care group partly did so, but to reach the numbers planned in advance, had to be assisted by a research assistant. PST registrars enrolled patients they thought would benefit from PST. Although this reflected daily practice in the sense that GPs offer only treatments to patients when they expect a positive effect on their health status, it resulted in a biased selection of patients: all patients met the eligibility criteria of 'emotional symptoms', a consultation rate of three or more, and a GHQ-12 score of four or more but PST patients were more often female, younger and had more severe psychological symptoms. Patients in the PST group might have been more suitable, and more motivated for treatment whereas patients selected for the usual care group were not specifically motivated. This limitation also might have overestimated treatment effects of PST. Although we corrected for the baseline differences in our analyses and still found significant advantages of PST above

UC, this limitation - together with the lack of randomization of registrars compromises the internal validity. It therefore remains unclear whether the effects were the results of a specific PST techniques, b. motivated registrars, c. more open attitudes of PST patients towards treatment, or d. a treatment like PST as a vehicle for registrars to incorporate non-specific skills - such as empathy, warmth and the doctor-patient relationship - better into their consultations with patients with emotional problems. The last option might be realistic, because a recent focus group study showed that registrars expressed that they implemented many new skills during PST.³⁵ For instance, they mentioned to appreciate the patient-centered and patient-empowering character, including the activation of patients to implement their own solutions in daily life. Patient-empowering skills are increasingly valued as important within primary care, especially in mental health care.³⁶ Therefore, we think that a treatment like PST might be a practical vehicle for registrars to incorporate non-specific treatment skills more manifestly in their patient contacts.

A strength of our study is that, to our knowledge, this was the first study with PST being provided by physicians of the patients' *own* general practice. All other PST studies involved PST therapists who were unknown to the patient whereas usual care was delivered by the patient's own GP. This probably overestimated usual care effects in earlier studies because the doctor-patient relationship influences patient outcome importantly.³⁷ Another strength is that this study was one of the very few studies with GP registrars providing a specific psychological treatment for emotional symptoms, including measurement of patient outcomes. An earlier American pilot study only measured feasibility and recommended measurement of patient outcomes.²¹ Recent Chinese research with registrars providing PST voluntarily did not show significant benefit of PST over placebo group intervention. These registrars, however, only provided 3 sessions of PST.³⁸ Unfortunately, the difficulties we encountered in this study suggest that involving registrars in a randomised trial was not optimal. because their motivation was essential in providing a psychological treatment.

Recommendations for further research We recommend a trial with randomization of registrars who are interested to provide PST. In this trial, measurement of motivation must be part of the design. Furthermore, we suggest investigating the effectiveness of PST when provided by the patient's own GP, because effects build upon the more longstanding relation with the patient.

Conclusion

Patients with emotional symptoms improved significantly more after PST delivered by motivated GP registrars than after usual care by GP registrars. Further research, with randomization of interested registrars or interested GPs, is needed.

Acknowledgments

We thank ZonMw, the Netherlands organization for health research and development, which funded our study, and we kindly thank all patients and GP registrars who participated.

References

- King M, Nazareth I, Levy G, Walker C, Morris R, Weich S, et al. Prevalence of common mental disorders in general practice attendees across Europe. Br J Psychiatry 2008 May;192(5):362-7.
- 2 van Rijswijk E, Borghuis M, van de Lisdonk E, Zitman F, van Weel C. Treatment of mental health problems in general practice: a survey of psychotropics prescribed and other treatments provided. Int J Clin Pharmacol Ther 2007 Jan;45(1):23-9.
- 3 Wittchen HU, Jacobi F. Size and burden of mental disorders in Europe--a critical review and appraisal of 27 studies. Eur Neuropsychopharmacol 2005 Aug; 15(4):357-76.
- 4 Gulbrandsen P, Fugelli P, Hjortdahl P. Psychosocial problems presented by patients with somatic reasons for encounter: tip of the iceberg? Fam Pract 1998 Feb;15(1):1-8.
- 5 Lader MH. Limitations on the use of benzodiazepines in anxiety and insomnia: are they justified? Eur Neuropsychopharmacol 1999 Dec;9 Suppl 6:S399-S405.
- 6 Williams JW, Jr., Mulrow CD, Chiquette E, Noel PH, Aguilar C, Cornell J. A systematic review of newer pharmacotherapies for depression in adults: evidence report summary. Ann Intern Med 2000 May 2;132(9):743-56.
- 7 van Geffen EC, Gardarsdottir H, van Hulten R, van Dijk L, Egberts AC, Heerdink ER. Initiation of antidepressant therapy: do patients follow the GP's prescription? Br J Gen Pract 2009 Feb;59(559):81-7.
- 8 Kirsch I, Deacon BJ, Huedo-Medina TB, Scoboria A, Moore TJ, Johnson BT. Initial severity and antidepressant benefits: a meta-analysis of data submitted to the Food and Drug Administration. PLoS Med 2008 Feb;5(2):e45.
- 9 Fritzsche K, Armbruster U, Hartmann A, Wirsching M. Psychosocial primary care what patients expect from their General Practitioners A cross-sectional trial. BMC Psychiatry 2002 May 8;2(1):5.

care physicians of patients who disclose psychosocial problems. J Fam Pract 1999 Sep;48(9):698-705.

- 11 Bower P, Rowland N, Mellor C, Heywood P, Godfrey C, Hardy R. Effectiveness and cost effectiveness of counseling in primary care. Cochrane Database Syst Rev 2002;(1):CD001025.
- 12 Mynors-Wallis LM, Gath DH, Day A, Baker F. Randomised controlled trial of problem solving treatment, antidepressant medication, and combined treatment for major depression in primary care. BMJ 2000 Jan 1;320(7226):26-30.
- 13 Mynors-Wallis LM, Gath DH, Lloyd-Thomas AR, Tomlinson D. Randomised controlled trial comparing problem solving treatment with amitriptyline and placebo for major depression in primary care. BMJ 1995 Feb 18;310(6977):441-5.
- 14 Dowrick C, Dunn G, Ayuso-Mateos JL, Dalgard OS, Page H, Lehtinen V, et al. Problem solving treatment and group psychoeducation for depression: multicentre randomised controlled trial. Outcomes of Depression International Network (ODIN) Group. BMJ 2000 Dec 9;321(7274):1450-4.
- 15 Wilkinson P, Mynors-Wallis L. Problemsolving therapy in the treatment of unexplained physical symptoms in primary care: a preliminary study. J Psychosom Res 1994 Aug;38(6):591-8.
- 16 Wood BC, Mynors-Wallis LM. Problem-solving therapy in palliative care. Palliat Med 1997 Jan;11(1):49-54.
- 17 Hassink-Franke LJA, Costa J, Lucassen PLBJ, Akkermans RP, Mynors-Wallis LM, van Weel C, et al. Problem solving treatment for emotional problems in primary care. Cochrane Database of Systematic Reviews 2010;Submitted.
- 18 Kerwick S, Jones R, Mann A, Goldberg D. Mental health care training priorities in general practice. Br J Gen Pract 1997 Apr;47(417):225-7.
- 19 Sahhar D, O'Connor D. How well do Australian medical schools prepare general practitioners

10 Robinson JW, Roter DL. Counseling by primary

6 140 - 141

to care for patients with mental disorders? Australas Psychiatry 2004 Mar, 12(1):26-30.

- 20 Ratcliffe J, Gask L, Creed F, Lewis B. Psychiatric training for family doctors: what do GP registrars want and can a brief course provide this? Med Educ 1999 Jun;33(6):434-8.
- 21 Hegel MT, Dietrich AJ, Seville JL, Jordan CB. Training residents in problem-solving treatment of depression: a pilot feasibility and impact study. Fam Med 2004 Mar;36(3):204-8.
- 22 Treweek S, Zwarenstein M. Making trials matter: pragmatic and explanatory trials and the problem of applicability. Trials 2009;10:37.
- 23 Koeter MWJ, Ormel J. General Health Questionnaire, Nederlandse bewerking: Handleiding. Lisse: Swets, Test Services; 1991.
- 24 Franke LJ, van Weel-Baumgarten EM, Lucassen PL, Beek MM, Mynors-Wallis L, van Weel C. Feasibility of training in problem-solving treatment for general practice registrars. Eur J Gen Pract 2007;13(4):243-5.
- 25 Spitzer RL, Kroenke K, Williams JB. Validation and utility of a self-report version of PRIME-MD: the PHQ primary care study. Primary Care Evaluation of Mental Disorders. Patient Health Questionnaire. JAMA 1999 Nov 10;282(18):1737-44.
- 26 Bjelland I, Dahl AA, Haug TT, Neckelmann D. The validity of the Hospital Anxiety and Depression Scale. An updated literature review. J Psychosom Res 2002 Feb;52(2):69-77.
- 27 Aaronson NK, Muller M, Cohen PD, Essink-Bot ML, Fekkes M, Sanderman R, et al. Translation, validation, and norming of the Dutch language version of the SF-36 Health Survey in community and chronic disease populations. J Clin Epidemiol 1998 Nov;51(11):1055-68.
- 28 EuroQol--a new facility for the measurement of health-related quality of life. The EuroQol Group. Health Policy 1990 Dec;16(3):199-208.
- 29 D'Zurilla TJ, Chang EC, Nottingham EJ, Faccini L. Social problem-solving deficits and hopelessness, depression, and suicidal risk in college students and psychiatric inpatients. J Clin Psychol 1998 Dec;54(8):1091-107.

- 30 Baker R. Development of a questionnaire to assess patients' satisfaction with consultations in general practice. Br J Gen Pract 1990 Dec;40(341):487-90.
- 31 Hakkaart-van Roijen L., van Straten A., Donker M., Tiemens B. Manual Trimbos/iMTA questionnaire for Costs associated with Psychiatric illness (TiC-P). Rotterdam; 2002.
- 32 Shewchuk RM, Johnson MO, Elliott TR. Selfappraised social problem solving abilities, emotional reactions and actual problem solving performance. Behav Res Ther 2000 Jul;38(7):727-40.
- 33 Terluin B, Grol MH, Pijnenborg L, Goudswaard AN. [Summary of the practice guideline 'Anxiety disorders' (first revision) from the Dutch College of General Practitioners]. Ned Tijdschr Geneeskd 2005 May 28;149(22):1211-5.
- 34 Romeijnders AC, van Marwijk HW, Goudswaard AN. [Summary of the practice guideline 'Depressive disorder' (first revision) from the Dutch College of General Practitioners]. Ned Tijdschr Geneeskd 2005 Mar 5;149(10):523-7.
- 35 Hassink-Franke LJA, olde Hartman TC, Beek MM, van Weel C, Lucassen PLBJ, van Weel-Baumgarten EM. Problem-solving treatment in general practice residency: a focus group study of registrars' views. Patient Educ Couns, in press.
- 36 Laugharne R, Priebe S. Trust, choice and power in mental health: a literature review. Soc Psychiatry Psychiatr Epidemiol 2006 Nov;41(11):843-52.
- 37 Kaptchuk TJ, Kelley JM, Conboy LA, Davis RB, Kerr CE, Jacobson EE, et al. Components of placebo effect: randomised controlled trial in patients with irritable bowel syndrome. BMJ 2008 May 3;336(7651):999-1003.
- 38 Lam CL, Fong DY, Chin WY, Lee PW, Lam ET, Lo YY Brief problem-solving treatment in primary care (pst-pc) was not more effective than placebo for elderly patients screened positive of psychological problems. Int J Geriatr Psychiatry 2009 Dec 23.

7 142 - 143

7 General discussion

The aim of this thesis was to investigate the feasibility of Problemsolving treatment (PST) during general practice (GP) residency and the effectiveness of PST, when provided by GP registrars, for patients with emotional symptoms in primary care. In this final chapter, the results of the foregoing chapters will be discussed in relation to each other and the literature. Also, some methodological issues will be considered. And finally, suggestions for further research and for general practice will be made, followed by a main conclusion. Throughout this thesis we referred to the same category of patients with the terms emotional symptoms, emotional problems and mental health problems.

Regarding the feasibility of PST during residency, we found that PST training is feasible. Our observational and questionnaire study showed that training in PST during residency is feasible because GP registrars participated in the two-day training programme and subsequent supervision, and their evaluation of the training was positive. Furthermore, they provided PST as taught in the protocol and patients were willing to receive PST of the registrars (Chapter 2). Besides being feasible. PST training was also helpful during residency, as was shown in the focus group study (Chapter 3). Registrars thought that PST training provided them with a practical tool in the management of emotional problems. In daily practice, however, they would prefer implementing elements rather than the entire treatment, because they thought the entire treatment costs too much time and is not their task. An important barrier in

the experiences of the registrars was their obligatory participation in this study. This prompted our interest in the views of registrars about participation in research in general. A questionnaire survey on this topic showed that registrars are interested in participating in research, especially when they can learn a new skill, but participation should not be compulsory, and registrars prefer to choose their own research subjects (Chapter 4).

Regarding the effectiveness of PST for patients with emotional problems in primary care, the Cochrane review indicated that PST was more effective than control treatments for major depression in primary care (Chapter 5). In major depression, PST patients had a statistically significant and clinically important better SF-36 social function score and a higher recovery rate at the BDI (2.54, 95% CI 1.85 to 3.50) than usual care patients at 6-month follow-up. Also, PST patients had statistically significant and clinically important higher recovery rates at the HDRS than patients in the placebo medication group at 3-month followup (RR 2.25, 95% CI 1.16 to 4.36). For emotional problems other than maior depression, however, there was insufficient evidence to show statistically significant differences between PST and usual care, (placebo) medication, or other psychological treatments. Further research of PST for these problems was recommended. The controlled clinical trial described in this thesis provided indications that PST by motivated registrars might be more favourable for patients with emotional symptoms than usual care by registrars (Chapter 6). At 9-month

7 144 - 145
follow-up PHQ recovery rates for somatoform disorder and anxiety were higher in the PST group (OR 6.50, p=0.01 respectively OR 11.25, p=0.03). Depressive symptoms did not improve significant differently between treatment groups. PST patients had improved significantly more on the 100-point SF-36 domains social functioning, role limitation due to emotional problems and general health perception (B-coefficient of the mean difference respectively 9.83, p=0.03; 17.18, p=0.04; 10.48, p<0.001). However, due to a considerably compromised validity of the trial, it remains unclear whether the effect in this trial can be attributed to the intervention. This will be described and discussed below.

In the performance of the studies described in this thesis we faced several themes and problems. We categorised these into four main themes: A. Strengths and weaknesses; B. Specific and non-specific effects; C. General considerations of research of psychological interventions; D. Realization of training during GP residency.

A. Strengths and weaknesses

There were a number of strengths in the studies described in this thesis.

 Overall, it is an important strength that we used mixed methods by performing both quantitative and qualitative studies. Using a mixed methods design gives the broadest view possible,¹ in this case on the feasibility of PST training during GP residency and the use of PST in daily practice. On the one side we quantitatively analysed data on participation in the training, recruited patients and time investment. And on the other side we qualitatively analysed registrars' views on PST during residency. In this way we came to both objective and subjective findings which informed us about the facts, barriers and enablers of PST during residency and PST in everyday practice. For instance, we found that registrars appreciated specific elements of PST but that they thought the entire treatment cost too much time. Furthermore. qualitative research has the advantage of generating unexpected insights. We, for instance, got to know registrars' views on their role in mental health care management.

- The performance of a systematic review and meta-analysis by means of a Cochrane review is also a strong point in this thesis. Cochrane reviews are generally regarded as the strongest level of evidence. Although more reviews on PST have been performed,^{2:4} these did not specifically focus on PST in primary care. In our review, we included all kinds of emotional problems rather than only depression and we focused on primary care studies.
- Also, performing a controlled trial with a large number of registrars (n=81) who provided either PST or usual care is a strength. A trial with this number of registrars had not been done before in PST research nor in studies of other psychological treatments.
- Furthermore, the trial described in this thesis was unique because it was the first with physicians of the patients' own practice providing PST. Therapists in all other PST studies so far were unfamiliar to the patient, in

the sense that PST was headed over to a provider outside of the practice.

- Another strength of the trial was the highly pragmatic character which reflected daily practice. For instance, we included the broad range and/or mixture of emotional symptoms which characterises general practice as GPs often see mixed symptoms rather than specific, full blown DSM-IV disorders.⁵ Most mental health research, however, is undertaken on people experiencing a 'pure' form of a psychiatric disorder, for instance major depression. Varieties or co-morbid symptoms are often deliberately excluded from studies. diminishing the external validity. especially in general practice.6

This, however, also brings us to some methodological limitations of the trial.

- The first limitation was the release of randomisation of registrars. We started our effectiveness study as a randomised controlled trial with the registrars being randomised in groups. However, as described in Chapter 6, registrars who did not feel comfortable with or motivated for PST, did not recruit patients. We therefore had to change the design in order to allow any comparison between PST and usual care. This led to selection of registrars participating in the PST group. Probably, these registrars are more motivated to deliver mental health care than their non-PST colleagues who provided care as usual. This higher motivation intensifies nonspecific elements of the treatment. Therefore, this might have caused an overestimation of the effects of PST and an underestimation of

the effects of usual care. The use of a formal instrument to measure the level of motivation of registrars towards mental health care or their attitudes towards psychosocial issues would have helped us to reveal the significance of this difference. Unfortunately, however, we did not measure motivation.

- A second methodological limitation in the trial was the selection of patients. The registrars selected patients who all met the inclusion criteria of: 1. being diagnosed with emotional symptoms; 2. a score of 4 or more at the 12-item General Health Ouestionnaire: 3. having had 3 or more consultations in the past half year. Next to that. registrars considered patients to be appropriate for the type of treatment they offered, which reflects daily practice: doctors offer treatments with the highest chance of success for the individual patient. This is important as patient and physician choice determine outcome importantly.7 From this pragmatic perspective, the external validity of the trial was therefore high. However, there were differences in the selection of patients between intervention and control group. Registrars in the intervention group selected patients themselves; registrars in the usual care group partly did so, but to reach the numbers planned in advance, had to be assisted by a research assistant. This has resulted in a biased selection of patients with the supposed result that patients who were selected in the intervention group were more suitable and motivated for the treatment whereas patients selected

7 146 - 147

for the usual care group were not specifically motivated. This limitation also results in more influence of nonspecific elements in the treatment and, in this study, favours the intervention group.

We conclude that the selection bias of both registrars and patients compromised the validity of the trial considerably enhancing the nonspecific elements of the treatment. thus causing an overestimation of the intervention as compared with usual care. Therefore, we cannot be sure whether the positive trial findings were the result of a. specific PST techniques, b. non-specific effects because PST registrars were probably more motivated, c. more open attitudes of PST patients towards active treatment, or d. a treatment like PST as a valuable vehicle for registrars to incorporate non-specific skills better into their consultations with patients with emotional problems. The last option might be realistic. because the focus group study showed that registrars thought that they implemented many new skills during PST. Registrars mentioned to appreciate the concretising and structuring character, the patientcentered and patient empowering character, including the activation of patients to think of and implement their own solutions in daily life, with patients being responsible for their own solutions. Patient empowering skills are increasingly valued as important within primary care, especially in mental health care.⁸ Therefore, we think that a treatment like PST might be a practical vehicle for registrars to incorporate non-specific treatment skills more manifestly in

their patient contacts.

B. Specific and non-specific effects

There are several factors influencing patient outcome in psychological treatments. Lambert and Barely dave a clear overview of these factors: extratherapeutic factors such as spontaneous remission and social support; expectancy effects such as the placebo effect; specific therapy techniques: and non-specific factors such as empathy, warmth, and the therapeutic alliance.⁹ They state, based on data from a large number of studies, that the specific therapy techniques account for only 15% to treatment outcome, whereas the non-specific factors within the doctorpatient relationship account for 30%, the expectancy effects for 15% and the extratherapeutic change for 40%. With regard to the positive findings in our review that PST is more effective than control treatments in patients with major depression, it is not clear whether these favourable effects are based on specific PST ingredients and/ or on non-specific ingredients - with PST as the vehicle - because these non-specific ingredients have not been described or studied in detail in any trial included in the review. The fact that the review did not show effectiveness of PST for emotional problems might have been related to an overestimation of the control treatments in the review, mostly usual care. Usual care was always provided by patients' own GPs, whereas PST was provided by therapists unfamiliar to patients. So, usual care patients had the advantage of longer-standing relationships with their GPs, which usually is related to better patient

outcome. In this light, it is conceivable that GPs applying PST to their own patients would have better results. Furthermore, it was not described whether GPs who delivered usual care might have applied some psychological skills too. It has been shown that therapists in general tend to use several skills interchangeably.¹⁰ To rule out differences in relationships - and other non-specific ingredients - between treatment groups, we need further research with equally optimised conditions: a trial comparing the effectiveness of PST to the effectiveness of control treatments. with all treatments being provided by the patient's own GP.

C. General considerations of research of psychological interventions

RCTs in psychotherapy research have been critiqued already for a long time, since RCTs in this field have some important limitations.^{7,11} These limitations concern aspects of internal and external validity. Concerning internal validity, RCTs assessing the value of psychotherapy are inherently less valid than RCTs assessing the value of medication, because of the impossibility to adequately blind RCTs in psychotherapy.

Concerning external validity, there are even more limitations. Firstly, in general an RCT assesses efficacy – whether the treatment works in a controlled environment – not whether it works in the real world (effectiveness). For instance, RCTs use randomization of patients rather than a naturalistic treatment selection. RCTs use homogeneous patient populations – due to strict diagnostic inclusion criteria – with an artificial treatment adherence, rather than 'real' patients who do not adhere always optimally. Also, proper comparison groups are sometimes lacking as waiting list controls might be disappointed and therefore do worse than intervention subjects.

Secondly, RCTs are mostly based on the biomedical model rather than on the contextual model. The biomedical model interprets the treatment as a mixture of specific ingredients that are critical to the outcome of therapy. It is more important whether the ingredient is received by the patient than who delivers the ingredient. Therefore, in RCTs treatments are manualised and controlled highly, which is at the cost of deviating from usual practice.11 This has been critiqued widely,¹² since many opponents comment the specific treatment effects and rather believe in generic effects such as hope and comfort provided by psychotherapy. The contextual model considers nonspecific factors essential in treatment outcome. This model regards the person of the therapist as critical because the model recognises that there will be variability in the manner in which treatments are delivered. meaning that the skills of therapists and their belief in treatment will vary.¹³ The therapist's motivation. attitude and belief are related to the therapist's communication¹⁴ and to patient outcome: the greater the therapist's belief in treatment's efficacy, the better the outcome.¹⁵ Furthermore, non-specific factors such as the doctor's empathy, warmth and the doctor-patient relationship, including the therapeutic alliance (i.e. the collaborative and affective bond between the patient and the therapist and their ability to agree

7 148 - 149

on treatment goals and tasks) are importantly related to patient outcome.^{9,16} Kaptchuk et al. for instance showed that non-specific effects of placebo acupuncture in irritable bowel syndrome produced significant outcomes, with the doctorpatient relationship as the strongest component.¹⁷ This highlights the importance of the development and maintenance of the therapeutic relationship as this is a primary curative component of treatment and provides the context in which specific techniques exert their influence.9 And also, it stresses the importance of measuring this important factor in outcomes research We would therefore advocate to

measure some of these important *non*specific treatment factors in RCTs, because these are usually not tested in traditional RCTs of psychological treatments.

D. Realization of training during GP residency

We performed the studies presented in this thesis within a local residency programme for two main reasons. Firstly, we intended to provide young doctors with a tool for the management of emotional problems already in an early stage of their career. Secondly, we expected GP registrars to be an appropriate group of doctors for the study, because we could insert the training and study in the existing training situation. The registrars, however, were not as accessible as we had expected them to be. Instead, we met a lot of resistance in the course of our research. We took the opportunity to learn from this situation and adjusted the design of the trial in order to complete it.

This also gave us the opportunity to learn more about the exact residency training situation and the combination of this with implementation of research and PST training. We identified three main themes within the setting of our research project: 1. Research during residency; 2. Registrars' attitude towards PST training; 3. Registrars' perspective on mental health management.

Research during residency 1. We performed this study in the registrars' third - and last - year of residency. This usually is a very busy year with many topics and intensive tasks to be finished by the registrars. Because participation in the study was obligatory and perceived as 'extra' ballast, many registrars had difficulty to find a positive motivation in an already full year to participate. The resistance towards participation they showed instead, did not only count for the PST group but also for many of the registrars in the usual care group. When we adjusted the design of the study and registrars received the choice of delivering either PST or usual care. the majority was neutral or positive about participation. Especially those registrars who chose for providing PST were satisfied, because they were trained and supervised in a treatment they felt a need or interest for. This was underpinned by the results of the questionnaire study about participation in research in general: registrars mentioned to be interested in research participation but they emphasised that they like to choose the research topic themselves. This suggests that a residency research programme that offers several

options would be attractive. We think that researchers and residency programmes should cooperate in such a way that they can offer a research programme that will be attractive for registrars. But this study may illustrate that this cooperation is not easily realised, even in a setting where the residency programme is an integral part of a university department with an established research programme. To bridge the worlds of education and research. it is important that the teachers are involved in research as well as the researchers to be involved in training. Enthusiastic faculty are necessary to serve as a role model and to integrate research into residency training. 18-20 This might create helpful conditions for registrars to participate without reservations in research during their residency.²¹ A more positive attitude towards research would fit in with the enthusiasm that most registrars show towards the evidence-based guidelines of the Dutch College of General Practitioners.²² Finally, registrars might be motivated by colleague peers who are actively involved in research. From this perspective it is highly encouraging that the group of registrars in the Netherlands who combine their residency training with research training is growing.²³

2. Registrars' attitude towards PST training

We thought that the registrars would be open for PST training. The training, however, did not only teach new knowledge but also requested a change of behaviour of professionals as they had to implement new communication skills. It can be discussed how changeable the clinical behaviour of these

professionals still is, especially since voung Dutch doctors have been receiving intense communication skills training at medical school for some time now.²⁴ New skills training should connect to these higher levels of communications skills at the start and it should meet registrars' needs. But when do registrars or GPs perceive the need to learn a psychological treatment like PST? This might be after their residency, when they have been practising for some time and are getting to know both their patients and the workload of mental health problems better.²⁵ On the other side, in the focus group study with registrars who obligatorily participated, we saw a difference between individual registrars: some had a particular need or interest to learn a treatment like PST, others hadn't. We therefore recommended to implement generic problem solving skills in an early stage of the GP residency programme, and to offer more specific skills in a later stage to those who opt for this.

Another interesting focus group finding with regard to using practical skills was that many registrars implemented PST in a way they felt at ease with. From an educational perspective this is encouraging: we offered all registrars the same training, but each individual registrar familiarises him/herself in another – *his/her own* – way with it. From a research perspective, however, this loss of uniformity might be disappointing because, as outlined before, psychotherapy research generally aims to provide treatments in a protocolised way.¹⁰

7 150 - 151

3. Registrars' perspective on mental health care

From the focus group study it appeared that registrars thought it is their task to diagnose mental health problems adequately, to offer patients a listening ear and to offer the patient some treatment. Offering a specific full psychological treatment, however, was not regarded by them as their task, not even when it concerned a brief treatment such as PST, which in our study took on average 1.7 consultations per patient more than usual care did. In the light of the desire of patients to receive nonpharmacological treatment from their own GP it is an important question what makes registrars somewhat defensive against this. Could this be related to their level of experience: are they still mainly biomedically oriented? Does it request more life and practice experience to discover the workload of mental health in primary care and the accompanying challenge of managing this adequately? In contrast, a Canadian study showed that most GPs are very interested in the detection and treatment of mental health problems.²⁶ The attitude of the registrars could also be in line with Hall's question whether a majority of GP registrars tailor their practice to a narrower spectrum of care rather than the wide spectrum of total, personal, context-bound and continuous health care.²⁷ Hall calls for discussion whether or not GP registrars hold firm to the definition of a GP as described in many national guidelines.

Recommendations for clinical practice

As described above, we found that PST was favourable for major

depressed patients, and that there is no evidence that PST is better as compared with usual care for emotional problems. However, thus far. no studies have focussed on PST within the context of the GPpatient relation. This is an important deficiency of the evidence base, as this would in all probability strengthen the 'a-specific' treatment effects. PST might be a vehicle to incorporate these non-specific factors more prominently into patient contacts. Possibly PST training provides doctors with a structure and optimism for treatment of mental health problems. We encourage the registrars who participated in the PST training, and meanwhile graduated as a GP, to continue using their skills - whether specific or not - for their patients with emotional problems.

Based on our findings with motivated registrars and despite currently lacking evidence of effectiveness, we invite interested GPs to take part in a PST training. It can meet their need for a practical tool in the non-pharmacological treatment of emotional problems and we expect it to have positive effects for their patients, especially since they already have longer lasting relationships with their patients. In the Netherlands, a two-day PST training for GPs has been organized since 2007 by the Dutch College of General Practitioners. An important logistical reason for registrars' limited motivation to perform a psychological treatment could be the current lack of financial reimbursements for providing psychological treatments combined with its time investment. In order to facilitate the graduated registrars and other GPs to provide a treatment like

PST, there should be a compensation for their time investment. The Health Care Insurance Board has given a positive advice to health insurances to offer a financial remuneration for GPs to provide PST, but unfortunately this advice has not been followed up yet. Another way of coping with the time investment of treatments like PST is to schedule the consulting hours differently, for instance by arranging dedicated sessions or by creating a brief session of 5-minute consultations for simple somatic reasons to compensate for the longer psychological treatment sessions. Some GPs might, like the registrars, implement only elements of PST rather than the entire treatment so that sessions have a maximum of 20 minutes each, which does fit into normal consulting hours. Although this sounds attractive and an official protocol for Brief Problem Solving (BPS) exists,²⁸ evidence for effectiveness of BPS is currently lacking.

The use of mental health nurses working in the practice could be helpful. Dutch research of PST provided by nurses did not show additional effects above usual GP care for patients with emotional symptoms.²⁹ whereas British research suggests that for major depressed patients the effectiveness of PST by nurses is as good as the effectiveness of PST by research GPs.³⁰ Combining this knowledge with the findings of our review, we could advocate it when GPs ask mental health nurses to provide PST for patients with major depression. Other studies investigated PST as part of a stepped-care programme and showed effectiveness.^{31,32} In the Cochrane

review we did not conclude anything about these programmes, as these studies were not included. Steppedcare programmes take different patient needs into account, depending on the characteristics of their mental health problems and their personal and social circumstance. An advantage of these programmes is their selfcorrecting character, meaning that another technique is tried if one technique is not working (sufficiently). This follows daily practice in a realistic way. However, most stepped-care programmes are protocolised whereas doctors in some cases would prefer to start with another option than the programme protocol prescribes.

Further research for patients with emotional symptoms

Both patients and GPs desire effective psychological treatments by the GPs themselves. We, therefore, need evidence on the effectiveness of PST by the patient's own GP. PST has not yet been investigated when provided by the patient's own GP. Also, as outlined above, the review did not provide us with definitive answers about the exact mechanism of effectiveness of PST as we do not know the non-specific factors in detail. Furthermore, there was an inter treatment group imbalance in the doctor-patient relationships with usual care being delivered by patients' own GPs. To rule out differences in relationships - and other non-specific ingredients - between treatment groups, we need an RCT in which all treatment groups have the advantage of their own GP delivering the treatment. We therefore recommend a trial with PST by the patient's own GP compared to a control group with

7 152 - 153

time and attention, and a control group delivering usual care both also by the patient's own GP. We suggest to recruit GPs who are interested to provide PST and then randomise them into one of the three groups, with the non-PST GPs being offered PST training after the trial.

Inclusion rather than exclusion of non-specific treatment factors is essential in order to know more about specific versus non-specific factors. We therefore recommend baseline measurement of a, the GP-patient relationship, including the GP-patient alliance; b. the GP's general beliefs in psychosocial aspects of patient care; c. the GP's allegiance (i.e. the GP's belief in the treatment's efficacy) in PST or control treatment; and d. personality characteristics and personal and social circumstances of patients recruited by the doctors. Many instruments exist for measuring the doctor-patient relationship and the doctor-patient alliance.¹⁶ The Physician Belief Scale can measure GPs' beliefs in psychosocial aspects³³ as could the Depression Attitude Questionnaire in depression research.³⁴ Allegiance can easily be rated for instance on a 5-point scale.¹⁵ By measuring these non-specific factors, we might also be able to reveal some components that produce the power of usual care. Quality of life, including daily functioning of patients, should be one of the main treatment outcomes. rather than focusing only on specific symptom reduction. Also, it can be useful to evaluate the symptoms that patients themselves regard as most hindering. Follow-up should be at least one year in order to know more about the long-term effects. The need for more research is underpinned by the

latest NICE-guideline on depression not recommending PST because of its limited evidence,³⁵ and a Dutch mental health summary describing it as 'promising'.³⁶

The review concluded that PST was not more effective than control treatments for emotional problems. In order to know whether PST is as effective as control treatments. equivalence trials are needed.³⁷ lf PST turns out to be as effective, it could be an attractive alternative for patients and doctors when for instance medication is not an option. A recent trial in Hong Kong used an adapted, 3-session form of PST but did not show favourable effects of PST above placebo treatment for depressed patients of 60 years and older. However, this was not Brief Problem Solving (BPS) which fits into normal consultations rather than being applied in distinct sessions. The developers of BPS have described it as a promising tool in the treatment of depression and anxiety.²⁸ However, we do not yet know exactly whether it works through its specific treatment techniques or - as a vehicle - through optimisation of non-specific elements within consultations, and if so, which non-specific elements these specifically are. If these questions are being answered and it turns out to be effective, BPS might suit GPs better than PST in their time-constricted schedules.

Recommendations for residency programmes

We recommend residency programmes to provide their registrars with psychological skills training, for instance PST training, because it fulfils their need for a practical tool in the management of patients with emotional problems. As outlined in Chapter 3 we recommend to teach the general skills in the beginning of the residency programme, whereas the more specific problem solving skills can be taught in a later stage when registrars perceive a greater need for these skills. Since 2008 we have been training our registrars in this way. Evaluations have been positive. Emphasizing the importance and helpfulness of many generic skills taught in the training - rather than training them a specific psychological treatment - might have reduced their resistance against the training. It is important to motivate registrars positively to use their problem solving skills, because when delivered in a motivated way these skills have positive effects on patient outcome. Registrars could possibly become more motivated through discussions with their role models - faculty and teaching GPs - about mental health care and providing psychological interventions. Residency programmes should facilitate reflective thinking and debate on mental health management in general as it concerns so many patients within primary care. As Gask set out, we need to challenge the potentially negative attitudes that doctors develop towards treating mental health problems, perhaps as a result of exposure to 'medical' models of mental illness in early training that do not fit with their later experiences.³⁸ Furthermore, residency programmes should emphasise the importance of relationship and other common factors much more as these enhance patient outcome far more than the focus on specific techniques.⁹ In fact, this should yet happen in medical school,

as for many GPs their practice is considerably influenced by the training they received as medical students.³⁸ In the current teaching, there is attention for thoughts and feelings of patients and how these should be addressed in the consultation. But there is scarce. attention for non-specific factors that relate to the doctor. Students and registrars should be stimulated to be aware of and to experiment more with non-specific factors such as warmth. empathy, comfort, hope, relationship. alliance, et cetera rather than focus on collecting all biomedical information of the patient. Registrars could for instance watch consultation videos of each other or of their teaching GP and label the non-specific aspects that might contribute to the quality of the doctor-patient relationship and to treatment outcome.

We also recommend residency programmes to offer registrars attractive research programmes as some registrars are very resistant to the idea of research,³⁹ and many registrars felt frustrated and resentful at being forced to participate in research.⁴⁰ This was underpinned by the results from the focus group study in this thesis. Registrars thought that participation in research is interesting. but they should be offered choice, time and resources. An American residency programme recently introduced a new system requiring that each registrar accumulates 'scholarly activity points'. This point system allowed registrars to pursue their own interests rather than being told what type of scholarly activity to participate in. It has transformed the atmosphere within the residency into a 'culture of inquiry'.⁴⁰ Research networks,

7 154 - 155

research departments and residency training programmes must collaborate to realise to most optimal programme. Next to that it is important to have energetic and motivated faculty involved in research to act as role models, to discuss research in front of. and with, registrars regularly, and to generate an atmosphere conducive to research.^{39,41} The goal is to increase registrars' interest in and likelihood of conducting or participating in research. This is important because research in primary care is essential since evidence to underpin this care cannot be based on hospital-based research alone.42

Further research in residency

We recommend to study whether the current, adapted PST training programme during residency meets the needs of registrars adequately, which specific and non-specific skills they actually pick up, and how they keep using these skills in daily practice. Regarding the needs of registrars a questionnaire survey or another focus group study could be performed. Video recordings could reveal registrars' use of specific and non-specific treatment skills in everyday practice. This is essential as the development and maintenance of the GP-patient relationship provides the context in which specific techniques exert their influence.

We recommend residency programmes to offer an attractive research programme in order to increase research appreciation and interest. It needs further research whether offering such an attractive research programme leads to more actively involved GPs in research after their residency training. Prospective studies with educational outcomes will need to be done to conclude the best methods of teaching GP registrar research knowledge, attitudes, and skills.

Conclusion

Training in problem-solving treatment (PST) during GP residency was feasible. Registrars thought that PST training was interesting and helpful, but they thought that implementing a full psychological treatment is not their task and costs too much time in daily practice. They especially appreciated specific elements of the treatment. All said they would use these elements in daily practice.

In a Cochrane review we found that PST was more favourable than control treatments for patients with a major depression but not for patients with emotional problems other than major depression. In a pragmatic controlled trial we found indications that PST by motivated registrars might have positive effects on primary care patients with emotional problems. However, due to limited internal validity, we are not sure whether these positive effects are the result of specific or non-specific treatment factors. We assume that non-specific factors probably are most promising in the treatment of emotional problems. PST could be a vehicle to incorporate these non-specific factors more prominently into patient contacts. We encourage the - meanwhile graduated - registrars who participated in the PST training to continue using their skills for their patients with emotional problems. We recommend residency programmes to teach the non-specific factors more explicitly, for instance through PST training. We furthermore

suggest residency programmes to offer registrars attractive research programmes, in order to increase registrars' interest in research

References

- Borkan JM. Mixed methods studies: a foundation for primary care research. Ann Fam Med 2004 Jan;2(1):4-6.
- 2 Malouff JM, Thorsteinsson EB, Schutte NS. The efficacy of problem solving therapy in reducing mental and physical health problems: a meta-analysis. Clin Psychol Rev 2007 Jan;27(1):46-57.
- 3 Bell AC, D'Zurilla TJ. Problem-solving therapy for depression: a meta-analysis. Clin Psychol Rev 2009 Jun;29(4):348-53.
- 4 Cuijpers P, van SA, Warmerdam L. Problem solving therapies for depression: a meta-analysis. Eur Psychiatry 2007 Jan;22(1):9-15.
- 5 Nease DE, Jr., Volk RJ, Cass AR. Investigation of a severity-based classification of mood and anxiety symptoms in primary care patients. J Am Board Fam Pract 1999 Jan;12(1):21-31.
- 6 Gunn J, Diggens J, Hegarty K, Blashki G. A systematic review of complex system interventions designed to increase recovery from depression in primary care. BMC Health Serv Res 2006;6:88.
- 7 Black N. Why we need observational studies to evaluate the effectiveness of health care. BMJ 1996 May 11;312(7040):1215-8.
- 8 Laugharne R, Priebe S. Trust, choice and power in mental health: a literature review.
 Soc Psychiatry Psychiatr Epidemiol 2006 Nov;41(11):843-52.
- 9 Lambert MJ, Barley DE. Research summary on the therapeutic relationship and psychotherapy outcome. Psychotherapy 2001;38(4):357-61.
- 10 Ablon JS, Jones EE. Validity of controlled clinical trials of psychotherapy: findings from the NIMH Treatment of Depression Collaborative Research Program. Am J Psychiatry 2002 May:159(5):775-83.

- 11 Hodgson R, Bushe C, Hunter R. Measurement of long-term outcomes in observational and randomised controlled trials. Br J Psychiatry Suppl 2007 Aug;50:s78-s84.
- 12 Rifkin A. Randomised controlled trials and psychotherapy research. Am J Psychiatry 2007 Jan;164(1):7-8.
- 13 Wampold BE. Therapist Effects: An Ignored but Critical Factor. The Great Psychotherapy Debate - Models, Methods, and Findings. Mahwah, New Jersey: Lawrence Erlbaum Associates, Inc.; 2001. p. 184-202.
- 14 Levinson W, Roter D. Physicians' psychosocial beliefs correlate with their patient communication skills. J Gen Intern Med 1995 Jul;10(7):375-9.
- 15 Wampold BE. Allegiance and Adherence: Further Evidence for the Contextual Model. In: Wampold BE, editor. The Great Psychotherapy Debate - Models, Methods, and Findings. Mahwah, New Jersey: Lawrence Erlbaum Associates, Inc.; 2001. p. 159-83.
- 16 Martin DJ, Garske JP, Davis MK. Relation of the therapeutic alliance with outcome and other variables: a meta-analytic review. J Consult Clin Psychol 2000 Jun;68(3):438-50.
- 17 Kaptchuk TJ, Kelley JM, Conboy LA, Davis RB, Kerr CE, Jacobson EE, et al. Components of placebo effect: randomised controlled trial in patients with irritable bowel syndrome. BMJ 2008 May 3;336(7651):999-1003.
- 18 Temte JL, Hunter PH, Beasley JW. Factors associated with research interest and activity during family practice residency. Fam Med 1994 Feb;26(2):93-7.
- 19 Carek PJ, Araujo D, Nalin PM. Scholarly activity and residency training: seeking strategic partnerships. Ann Fam Med 2005 Nov;3(6):560-1.
- 20 Dehaven MJ, Wilson GR, O'Connor-Kettlestrings P. Creating a research culture: what we can learn from residencies that are successful in research. Fam Med 1998 Jul;30(7):501-7.
- 21 Neale AV. A national survey of research requirements for family practice residents and faculty. Fam Med 2002 Apr;34(4):262-7.

7 156 - 157

- 22 Wiersma D, Geijer RRM. Practice Guidelines 2010 from the Dutch College of General Practitioners. Houten: Bohn Stafleu van Loghum; 2009.
- 23 olde Hartman TC, Poels PJ, Licht-Strunk E, van WC. Combining vocational and research training. Aust Fam Physician 2008 Jun;37(6):486-8.
- 24 van DJ, Kerkhofs E, van Knippenberg-Van Den Berg BW, van Den Hout HA, Scherpbier AJ, van d, V. Longitudinal and concentrated communication skills programmes: two dutch medical schools compared. Adv Health Sci Educ Theory Pract 2002;7(1):29-40.
- 25 Zantinge EM, Verhaak PF, Kerssens JJ, Bensing JM. The workload of GPs: consultations of patients with psychological and somatic problems compared. Br J Gen Pract 2005 Aug;55(517):609-14.
- 26 Clatney L, Macdonald H, Shah SM. Mental health care in the primary care setting: family physicians' perspectives. Can Fam Physician 2008 Jun;54(6):884-9.
- 27 Hall K, Abercrombie S, Carr S, Gravel J, Jr., Hoekzema G, Kozakowski S, et al. Family medicine: what are we graduating? Ann Fam Med 2009 Nov;7(6):567-8.
- 28 van Weel-Baumgarten E, Jane-Liopis E, Mynors-Wallis L, Anderson P. Prevention of mental illness in primary care. The IMHPA manual and the general practitioners' role. Eur J Gen Pract 2005 Sep;11(3-4):92-3.
- 29 Schreuders B, van Marwijk H, Smit J, Rijmen F, Stalman W, van Oppen P. Primary care patients with mental health problems: outcome of a randomised clinical trial. Br J Gen Pract 2007 Nov;57(544):886-91.
- 30 Mynors-Wallis LM, Gath DH, Day A, Baker F. Randomised controlled trial of problem solving treatment, antidepressant medication, and combined treatment for major depression in primary care. BMJ 2000 Jan 1;320(7226):26-30.
- 31 van't Veer-Tazelaar PJ, van Marwijk HW, van OP, van Hout HP, van der Horst HE, Cuijpers P, et al. Stepped-care prevention of anxi-

ety and depression in late life: a randomised controlled trial. Arch Gen Psychiatry 2009 Mar;66(3):297-304.

- 32 Hunkeler EM, Katon W, Tang L, Williams JW, Jr., Kroenke K, Lin EH, et al. Long term outcomes from the IMPACT randomised trial for depressed elderly patients in primary care. BMJ 2006 Feb 4;332(7536):259-63.
- 33 Ashworth CD, Williamson P, Montano D. A scale to measure physician beliefs about psychosocial aspects of patient care. Soc Sci Med 1984;19(11):1235-8.
- 34 Dowrick C, Gask L, Perry R, Dixon C, Usherwood T. Do general practitioners' attitudes towards depression predict their clinical behaviour? Psychol Med 2000 Mar;30(2):413-9.
- 35 National Institute for Health and Clinical Excellence. Depression: the treatment and management of depression in adults. 2009.
- 36 Zwaanswijk M, Verhaak PFM. Effectieve kortdurende interventies voor psychische problemen - Een kennissynthese over hun toepasbaarheid in de huisartsenvoorziening. Utrecht: Nivel; 2009.
- 37 Hermens ML, van Hout HP, Terluin B, Ader HJ, de HM. [The importance of equivalence trials in showing the usefulness of treatments]. Ned Tijdschr Geneeskd 2003 Nov 1;147(44):2162-6.
- 38 Gask L, Dixon C, May C, Dowrick C. Qualitative study of an educational intervention for GPs in the assessment and management of depression. Br J Gen Pract 2005 Nov;55(520):854-9.
- 39 Seehusen DA, Weaver SP. Resident research in family medicine: where are we now? Fam Med 2009 Oct;41(9):663-8.
- 40 Seehusen DA, Asplund CA, Friedman M. A point system for resident scholarly activity. Fam Med 2009 Jul;41(7):467-9.
- 41 Carek PJ, Mainous AG, III. The state of resident research in family medicine: small but growing. Ann Fam Med 2008 Jan;6 Suppl 1:S2-S4.
- 42 Mant D, Del Mar C, Glasziou P, Knottnerus A, Wallace P, van Weel C. The state of primary-care research. Lancet 2004 Sep 11;364(9438):1004-6.

Summary

This thesis focused on problem-solving treatment (PST), a brief psychological treatment for patients with mental health problems in primary care. The general objectives were to investigate 1. the feasibility of PST training during general practice (GP) residency, and 2. the effectiveness of PST for patients with emotional symptoms in primary care.

Chapter 1. General introduction

In this chapter the rationale, aims and outline of this thesis are explained. Mental health problems are common, with one in four adults suffering from symptoms of depression, anxiety, sleeping problems, irritability, stress, or psychosocial problems. Quality of life is influenced negatively and many patients visit their general practitioner (GP) frequently. Therefore, mental health problems account for a substantial proportion of the GP's workload.

Both patients and physicians desire effective non-pharmacological treatments, especially because medication is not always the best option in the treatment of mental health problems. GP registrars often express the need for more expertise of psychological treatment skills for patients with mental health problems. Problem-solving treatment (PST) could be an attractive option. PST is a brief psychological treatment teaching patients how to use their own skills to cope with problems. PST has been shown to be effective for patients with depression, but the effectiveness for the broader range of mental health problems – typical for primary care – had not been investigated before. Training in PST was introduced in a Dutch GP residency programme to meet the registrars' need. We aimed to assess 1. the feasibility of PST training during GP residency, and 2. the effectiveness of PST for patients with emotional symptoms in primary care. In this thesis we referred to the same category of patients with the terms emotional symptoms, emotional problems and mental health problems.

Part I. Feasibility of problem-solving treatment during general practice residency

Chapter 2. Feasibility of training in problem-solving treatment for geeral practice registrars

In this chapter the feasibility of PST training during residency is described. For this purpose we used data from an observational study of 20 GP registrars who received a two-day PST training and subsequent supervision in the treatment of patients with mental health problems. The registrars evaluated the training positively. All registrars subsequently used PST, treated 52 patients, and received supervision. Registrars appreciated PST for its structure and the active role of patients during treatment. The conclusion was that training GP registrars in PST is feasible.

Chapter 3. Problem-solving treatment in general practice residency: a focus group study of registrars' views

In this chapter the results of a qualitative focus group study are presented. We aimed to explore GP registrars' views on PST training during residency and on the actual use of PST in general practice. We interviewed 18 Dutch registrars - divided over 4 groups - who had been trained in PST during residency. Registrars expressed that PST training during residency was feasible, interesting and helpful, but they found that it took too much time in everyday practice and that it was not a GP's task. All registrars, however, said they would use specific elements in a variety of consultations, for instance concretising problems, brainstorming about practical solutions, and activating patients. We recommended residency programmes to offer training in PST or another psychological treatment with comparable elements.

Chapter 4. General practice registrars and research - Attitudes toward participation

Primary care research is important and early exposure of GPs to research is recommended to increase family medicine research capacity. The participation of GP registrars in the study described in this thesis enabled us to determine their actual participation and to explore their views on participation in research in general. This chapter presents the results of an observational study of 67 registrars' participation, and a questionnaire survey of their opinions of research and participation in research in daily practice. The registrars recruited 208 patients. They were interested in participating in research, especially when they can learn a new skill. Obligatory participation, lack of time and difficulties with patient recruitment were important barriers to participation. Registrars reported that participation should not be compulsory, and they preferred to choose their own research subjects. Assuming a desire for research to be part of the culture of family medicine. we suggested that researchers and residency programmes offer research in such a way that registrars will find it attractive to participate in it.

Part II. Effectiveness of problemsolving treatment for patients with emotional symptoms in primary care

Chapter 5. Problem-solving treatment for emotional problems in primary care – a Cochrane review

Many patients with emotional problems prefer psychological treatments delivered within the primary care setting. Evidence for psychological treatments, however, comes largely from studies in secondary care settings where patients are referred to a specialist. PST has been shown to be effective in patients with depression but its effectiveness has not been proven for the broader range of emotional problems seen in primary care. Chapter 5 contains the results of a systematic review according to the principles of the Cochrane Collaboration This review aimed to assess the effectiveness of PST versus usual care, medication or other psychological treatments in adult patients with emotional problems in primary care. We included 12 studies with 2261 participants. Four studies were of good quality, five studies of moderate quality and three studies of low quality. Three studies were in favour of PST, three were in favour of control treatment and six did not show significant differences between PST and control treatments.

Overall, PST was not different from usual care, antidepressants, placebo medication or other psychological treatments. For patients with major depression, however, two good quality studies showed that PST was more effective than usual care, placebo medication and group psychoeducation: PST patients had a significantly important better SF-36 social function score and a higher recovery rate at the BDI (2.54, 95% Cl 1.85 to 3.50) than usual care patients at 6-month follow-up. Also. PST patients had significantly higher recovery rates at the HDRS than patients in the placebo medication group at 3-month follow-up (RR 2.25, 95% CI 1.16 to 4.36). For emotional problems other than major depression, however, there was insufficient evidence to show

significant differences between PST and usual care, (placebo) medication, or other psychological treatments. We therefore recommended further research of PST for these problems. Costs and health care use did not show major differences between PST and control treatments. Patients were more satisfied after PST or generic care from a mental health nurse than after usual GP care or after antidepressant medication.

Chapter 6. Effectiveness of problem-solving treatment by general practice registrars for patients with emotional symptoms

This chapter describes the results of a pragmatic controlled trial. We aimed to study the effectiveness of PST delivered by trained GP registrars for patients with mental health problems. We compared the effectiveness of PST versus usual care for patients with mental health problems. GP registrars provided PST or usual care, according to their own preference. Patients were included if they (a) had presented for three or more visits with emotional symptoms in the past 6 months; and (b) scored four or more on the 12-item General Health Ouestionnaire. Outcomes at 3 and 9 month follow-up were standard measures of depression, anxiety and quality of life. Thirty-eight GP registrars provided PST and included 98 patients: 43 provided usual care and included 104 patients. PST

160 - 161

patients improved significantly more than usual care patients: at 9-month follow-up PHO recovery rates for somatoform disorder and anxiety were higher in the PST group (OR 6.50, p=0.01 respectively OR 11.25. p=0.03). Depressive symptoms did not improve significantly different between treatment groups. PST patients had improved significantly more on the 100-point SF-36 domains social functioning, role limitation due to emotional problems and general health perception (B-coefficient of the mean difference respectively 9.83, p=0.03; 17.18, p=0.04; 10.48, p<0.001). Although these results provided indications that PST by motivated registrars might be more favourable for patients with mental health problems than usual care by registrars, it remained - due to a compromised validity of the trial unclear whether the effect in this trial could be attributed to the intervention.

Chapter 7. General discussion

The final chapter considers the results described in this thesis together with some methodological issues, and ends with suggestions for further research and general practice.

Due to limited internal validity of the trial described in this thesis, it was not sure whether the positive trial effects were the result of specific or non-specific treatment factors within the doctor-patient relationship. Nonspecific factors account for a greater part to psychological treatment outcome than specific factors, but are scarcely studied in trials. We assumed that non-specific factors probably were most promising in the treatment of emotional problems. PST could be a vehicle to incorporate these non-specific factors more prominently into patient contacts. We therefore encourage the registrars who participated in the PST training, and meanwhile graduated as a GP, to continue using their skills – whether specific or not.

The most important recommendations for further research were: a. An RCT with equally optimised doctor-patient relationships - and other non-specific factors - in all treatment groups, comparing the effectiveness of PST to the effectiveness of control treatments, with all treatments being delivered by the patient's own GP: b. Measurement of nonspecific treatment factors in trials of psychological treatments. The most important recommendations for residency programmes were: a. An attractive research programme in order to increase registrars' interest in research; b. Implementation of training in PST or a psychological treatment with comparable elements; c. More explicit teaching of non-specific treatment factors.

Samenvatting

Dit proefschrift beschrijft een onderzoek naar problem-solving treatment (PST). PST is een korte psychologische behandeling voor patiënten met psychische klachten in de huisartsenpraktijk. Doel was om te onderzoeken of: 1. PST training haalbaar is tijdens de huisartsopleiding; 2. PST effectief is voor patiënten met psychische klachten in de huisartsenpraktijk. Deze Nederlandse samenvatting is bestemd voor lezers die niet over een medische achtergrond beschikken. Lezers die graag meer details lezen, verwijs ik naar de voorgaande hoofdstukken.

Hoofdstuk 1. Algemene inleiding

In dit hoofdstuk worden de achtergrond, doelen en opbouw van dit proefschrift toegelicht. Psychische klachten komen veel voor. Ongeveer een op de vier volwassenen heeft last van symptomen van depressie, angst, slaapproblemen, stress, prikkelbaarheid, of psychosociale problemen. De kwaliteit van leven wordt hierdoor vaak negatief beïnvloed en veel patiënten met deze problematiek bezoeken hun huisarts frequent. Daarmee zorgen psychische klachten voor een substantieel deel van de werklast van de huisarts. Zowel patiënten als artsen wensen effectieve niet-medicamenteuze behandelingen, vooral omdat medicatie niet altijd de beste optie is in de behandeling van psychische klachten. Artsen in opleiding tot huisarts geven vaak aan dat zij behoefte hebben aan meer deskundigheid op het gebied van psychologische behandelvaardigheden voor patiënten met psychische klachten. Problem-solving treatment (PST)

zou hiervoor een aantrekkelijke optie kunnen zijn. PST is een korte psycholoaische behandeling die patiënten leert hoe ze hun eigen vaardigheden kunnen aebruiken om met problemen om te gaan. PST is effectief gebleken voor patiënten met een depressie, maar de effectiviteit voor de bredere range aan psychische klachten – die juist zo typisch voor de huisartsenpraktijk is - was nog niet eerder onderzocht. Om tegemoet te komen aan de behoefte van artsen in opleiding tot huisarts introduceerden we. de onderzoekers van dit project, PST training in de Nijmeegse huisartsopleiding. We wilden het volgende beoordelen: 1. De haalbaarheid van PST training tijdens de huisartsopleiding; en 2. De effectiviteit van PST voor patiënten met psychische klachten in de huisartsenpraktijk.

Deel I. Haalbaarheid van problemsolving treatment tijdens de huisartsopleiding

Hoofdstuk 2. Haalbaarheid van PST training voor artsen in opleiding tot huisarts

Dit hoofdstuk beschrijft de haalbaarheid van PST training tijdens de huisartsopleiding. We observeerden 20 artsen in opleiding tot huisarts die deelnamen aan een tweedaagse PST training en aansluitend supervisie kregen op hun behandeling van patiënten met psychische klachten. De artsen beoordeelden de training positief. Alle artsen in opleiding tot huisarts pasten PST vervolgens toe, ze behandelden 52 patiënten, en ontvingen hierop supervisie. Ze waardeerden PST vanwege de structuur en de actieve rol van patiënten tijdens de behandeling. De conclusie was dat PST training in de huisartsopleiding haalbaar is.

Hoofdstuk 3. PST tijdens de huisartsopleiding: een focusgroepstudie van de opvattingen van artsen in opleiding tot huisarts

In dit hoofdstuk worden de resultaten van een focusgroepstudie gepresenteerd. Doel was om te onderzoeken wat de meningen van artsen in opleiding tot huisarts waren over PST training tijdens de huisartsopleiding en wat hun meningen waren over het gebruik van PST in de huisartsenpraktijk. We interviewden 18 Nederlandse artsen in opleiding tot huisarts - verdeeld over 4 groepen - die getraind waren in PST tijdens hun huisartsopleiding. De artsen in opleiding gaven aan dat PST training tijdens de huisartsopleiding haalbaar, interessant en nuttig was, maar ze vonden dat het teveel tijd kostte in de dagelijkse praktijkvoering en dat het toepassen van een psychologische behandeling niet de taak van een huisarts is. Ze zeiden echter allen dat ze specifieke elementen uit de behandeling zouden toepassen in een variëteit aan consulten. Ze noemden als waardevolle elementen met name het concretiseren van problemen, het brainstormen over praktische oplossingen, en het activeren van patiënten. We adviseerden huisartsopleidingen om training in PST, of een andere psychologische behandeling met vergelijkbare elementen, aan te bieden aan hun artsen in opleiding.

Hoofdstuk 4. Artsen in opleiding tot huisarts en onderzoek – Opvattingen over deelname aan onderzoek

Huisartsgeneeskundig wetenschappelijk onderzoek is belangrijk en vroege blootstelling van huisartsen aan onderzoek is aanbevolen zodat de onderzoekscapaciteit toeneemt. De in dit proefschrift beschreven deelname van artsen in opleiding tot huisarts aan onderzoek gaf ons de mogelijkheid om hun daadwerkelijke onderzoeksdeelname vast te stellen en om hun opvattingen in kaart te brengen over deelname aan onderzoek in het algemeen. Dit hoofdstuk beschriift enerziids de resultaten van een observationele studie van de deelname van 67 artsen in opleiding tot huisarts, en anderzijds de resultaten van een vragenlijststudie naar hun opvattingen over onderzoek en over deelname aan onderzoek in de dagelijkse praktijk. De 67 artsen in opleiding sloten in totaal 208 patiënten in het onderzoek in. Ze gaven aan geïnteresseerd te zijn in deelname aan onderzoek, vooral wanneer ze een nieuwe vaardigheid kunnen leren. Verplichte deelname, tijdgebrek en moeilijkheden met patiëntenwerving waren volgens hen belangrijke barrières in deelname. Artsen in opleiding tot huisarts gaven aan dat onderzoeksdeelname niet verplicht zou moeten zijn, en ze gaven er de voorkeur aan om hun eigen onderzoeksonderwerp te kiezen. Er van uitgaande dat onderzoek een wenselijk deel uitmaakt van de huisartsgeneeskunde, suggereerden we dat onderzoekers en huisartsopleidingen onderzoek zodanig aanbieden dat artsen in opleiding tot huisarts het aantrekkelijk

vinden om eraan deel te nemen.

Deel II. Effectiviteit van problemsolving treatment voor patiënten met psychische klachten in de huisartsenpraktijk

Hoofdstuk 5. Problem-solving treatment voor psychische klachten in de eerste lijn – een Cochrane review

Veel patiënten met psychische klachten geven de voorkeur aan psychologische behandeling binnen de huisartsenpraktijk. Onderzoek naar psychologische behandelingen vond tot op heden echter vooral plaats in de tweede lijn, zoals de psychiatrie, waar patiënten verwezen worden naar een specialist. In eerder onderzoek bleek PST effectief bij patiënten met depressie maar het is nog onduidelijk in hoeverre PST ook effectief is voor de bredere range aan psychische klachten die juist zoveel aezien wordt in de huisartsenpraktijk. Hoofdstuk 5 omvat de resultaten van een systematisch literatuuroverzicht volgens de principes van de Cochrane Collaboration, een internationaal netwerk dat helpt om op basis van gestructureerde literatuuroverzichten besluiten te nemen omtrent gezondheidsvraagstukken. Dit literatuuroverzicht had als doel om de effectiviteit te beoordelen van PST ten opzichte van gebruikelijke zorg, medicatie of andere psychologische behandelingen bij volwassen patiënten met psychische klachten in de huisartsenpraktijk. In ons overzicht namen we 12 studies op met in totaal 2261 deelnemende patiënten. Vier studies waren van goede kwaliteit, vijf studies van gemiddelde kwaliteit en drie studies van lage kwaliteit. De uitkomsten van drie studies waren ten gunste van PST, drie ten gunste van de controlebehandeling en zes studies lieten geen significante

verschillen zien tussen PST en andere behandelingen.

Overall waren de effecten van PST niet anders dan die van gebruikelijke zorg. antidepressiva, placebo medicatie, of andere psychologische behandelingen. Voor patiënten met een ernstige depressie echter, toonden twee studies van goede kwaliteit aan dat PST effectiever was dan gebruikelijke zorg, placebo medicatie of groepseducatie. Echter, voor andere psychische klachten dan een ernstige depressie vonden we onvoldoende bewijs dat PST andere effecten heeft dan gebruikelijke zorg, (placebo) medicatie, of andere psychologische behandelingen. Daarom adviseerden we verder onderzoek van PST voor deze klachten. We vonden geen noemenswaardige verschillen tussen PST en andere behandelingen voor wat betreft kosten en gezondheidszorgconsumptie. Wel was duidelijk dat patiënten meer tevreden waren na PST of na zorg van een GGz-verpleegkundige dan na gebruikelijke zorg door de huisarts of na gebruik van antidepressiva.

Hoofdstuk 6. Effectiviteit van problem-solving treatment door artsen in opleiding tot huisarts voor patiënten met psychische klachten

Dit hoofdstuk beschrijft de resultaten van een patiëntenonderzoek. We beoogden de effectiviteit te bestuderen van PST, gegeven door getrainde artsen in opleiding tot huisarts, aan patiënten met psychische klachten. We vergeleken de effectiviteit van PST met die van gebruikelijke zorg voor patiënten met psychische klachten in de huisartsenpraktijk. Artsen in opleiding tot huisarts leverden gedurende het gehele onderzoek ofwel PST ofwel gebruikelijke zorg, afhankelijk van hun eigen voorkeur. Patiënten werden in het onderzoek ingesloten als zij (a)

164 - 165

driemaal of vaker met psychische klachten bij de huisarts waren geweest in het afgelopen half jaar; en (b) een score van vier of hoger hadden op de 12-item General Health Ouestionnaire (een zelfrapportagelijst voor het meten van psychische klachten). Na 3 en 9 maanden vulden de deelnemende patiënten vragenlijsten in met vragen over depressie, angst en kwaliteit van leven. Achtendertig artsen in opleiding tot huisarts leverden PST en sloten 98 patiënten in het onderzoek in; 43 artsen in opleiding tot huisarts leverden gebruikelijke zorg en sloten 104 patiënten in. PST-patiënten verbeterden significant meer dan controlepatiënten die gebruikelijke zorg ontvingen: PSTpatiënten rapporteerden significant minder symptomen van depressie en angst en een betere kwaliteit van leven dan controlepatiënten, zowel na 3 als na 9 maanden. Hoewel deze resultaten er op wijzen dat PST door artsen in opleiding tot huisarts waarschijnlijk beter is voor patiënten met psychische klachten dan gebruikelijke zorg door artsen in opleiding tot huisarts, blijft het - ten gevolge van een beperkte validiteit van het onderzoek- onduidelijk of het positieve effect in dit onderzoek toegeschreven kan worden aan PST.

Hoofdstuk 7. Algemene discussie

Het laatste hoofdstuk plaatst de resultaten van dit proefschrift in een perspectief en bespreekt enkele methodologische kwesties. Tenslotte geeft het suggesties voor verder onderzoek en voor de huisartsenpraktijk en huisartsopleiding. Doordat de interne validiteit van het patiëntenonderzoek beschreven in dit proefschrift beperkt is, is het niet geheel duidelijk of de positieve effecten het resultaat zijn van specifieke of non-specifieke behandelfactoren binnen de arts-patiëntrelatie. Nonspecifieke factoren (zoals empathie, warmte, hoop geven) dragen voor een groter deel bij aan het effect van een psychologische behandeling dan de specifieke factoren. Maar deze nonspecifieke factoren worden nauwelijks bestudeerd in wetenschappelijk onderzoek. Wij veronderstellen dat nonspecifieke factoren waarschijnlijk het meest belovend waren in de behandeling van psychische klachten. PST zou een vehikel kunnen zijn om deze nonspecifieke factoren meer prominent te incorporeren in patiëntcontacten. Daarom moedigden wij artsen in opleidingen tot huisarts, die deelnamen aan de PST training en inmiddels gecertificeerd huisarts zijn, aan om hun vaardigheden te blijven gebruiken - al dan niet specifiek. De belangrijkste aanbevelingen voor verder onderzoek waren:

- Een gerandomiseerd onderzoek waarbij de arts-patiëntrelaties – en andere non-specifieke factoren – in alle behandelgroepen even optimaal zijn. Dit onderzoek vergelijkt dan de effectiviteit van PST met de effectiviteit van andere behandelingen, waarbij alle behandelingen worden gegeven door de eigen huisarts van de patiënt;
- Het meten van non-specifieke behandelfactoren in onderzoeken van psychologische behandelingen.

De belangrijkste aanbevelingen voor huisartsopleidingen waren:

- Een aantrekkelijk onderzoeksprogramma om de interesse voor onderzoek bij artsen in opleiding tot huisarts te vergroten;
- Implementatie van training in PST of een psychologische behandeling met vergelijkbare elementen;
- c. Explicieter onderwijs van nonspecifieke behandelfactoren.

Acknowledgments / Dankwoord

Met dit laatste hoofdstuk zal een AIOTHO-traject dat in 2003 begon als een lange weg dan ineens klaar zijn. De voldoening is groot. Tegelijk is het jammer dat een einde komt aan een schitterende tijd van verdieping en ontwikkeling, waarbij onderzoek prachtig teamwork bleek. Ik ben dankbaar dat ik door de medewerking, hulp en inspiratie van anderen zoveel heb mogen leren. Daarom wil ik hierbij dan ook velen bijzonder hartelijk danken.

Promotor Chris van Weel, jij hebt me alle kans gegeven om het schrijven van dit proefschrift te volbrengen en me te ontwikkelen als onderzoeker. Hoewel ik zelf vaak neidde vooral te zien wat er aan mijn onderzoek schortte, wist jij iedere keer opnieuw de krachten te benoemen. Daarmee verraste en enthousiasmeerde je me. Omgekeerd ben ik er trots op dat kwalitatief onderzoek inmiddels concurreert met cricket... je corrigeerde een artikel tijdens een cricketwedstrijd te Singapore! Chris, ik heb genoten en geleerd van de helikopterblik waarmee jij alles in breder perspectief plaatst, heerlijk. Zeer bedankt voor je goede begeleiding.

Copromotor Evelyn van Weel-Baumgarten, je gedrevenheid in zaken is ongekend: je doet veel tegelijk en alles even grondig. Ik kon je perfectionisme waarderen want je keek kritisch en gedetailleerd naar ieder stuk van mijn hand. En je snelheid is ongekend: op welk moment van de dag ik je ook mailde, ik kreeg altijd binnen 24 uur een reactie! Ik leerde van je correcte stijl van communiceren en van je onderwijstalent. Doordat je me in veel zaken betrok, gaf je me de kans om veel meer te leren dan alleen dat wat het PST-onderzoek betrof. Daarbij heb ik je persoonlijke belangstelling door de jaren heen gewaardeerd. Hartelijk dank voor je begeleiding.

Peter Lucassen, mijn andere copromotor, ik heb veel geleerd van jouw methodologische kijk en heldere manier van schrijven. Je leerde me om onbevooroordeeld bij de onderzoeksfeiten te blijven. Daarnaast waren de discussies aan 'onze' koffietafel altiid era vruchtbaar: door jouw veelzijdigheid ging ik over meer, al dan niet huisartsgeneeskundige, zaken beter nadenken en weet ik nu wat potazzi mej skrewsaus is. Je warme persoonlijkheid maakt het tot een genot om al die jaren kamergenootjes te zijn geweest, overigens niet alleen in Nijmegen maar ook in Orlando! Ennuh, 'Failure is no option' liikt nu bewaarheid te worden...

Mechtild Beek, mijn derde copromotor, jij was vanuit de huisartsopleiding betrokken bij ons onderzoek, waarmee je een duidelijk andere inbreng had. Ik genoot van je kritische houding: je behartigde de belangen van de huisartsen in opleiding en was bovendien een onderzoeker die de realiteit voor ogen hield waarmee dit onderzoek zou kunnen slagen. Ook in het schrijfproces wees je me op heel andere, veelal opleidingsgerelateerde aspecten. Dank voor je heldere inbreng.

Ik heb het geluk gehad om met nog enkele andere mede-onderzoekers en co-auteurs samen te werken. Henk van den Hoogen, Hans Bor en Reinier Akkermans, hartelijk dank voor de methodologische en statistische inbreng. Door jullie heldere, geduldige uitleg werden de analyses stukken leuker! Daarnaast was het natuurlijk leuk om onze fietservaringen te delen. Laurence Mynors-Wallis, founder of PST for primary care, thank you for your co authorship and helping us with British humour! - in the training of the GP registrars. I'd also like to thank Yo Davies for the kind help in the training of the registrars.

Joe Costa, you helped me wonderfully with the enormous job of the Cochrane review, thanks a lot! Will you come back to Holland during the next European Championship?

Lest best, co-auteur en vooral onderzoeksmaatje Tim olde Hartman. Wat heb ik veel aan je gehad: je heldere, kritische kijk op zaken, je onderzoekstalent, je voortvarendheid en je oneindige optimisme, heerlijk! Bovendien... *leu oet Twente, leuke leu.* Het is dan ook geweldig dat jij mijn paranimf bent.

Tony Avery, thank you for your warm welcome in your practice and your supervision during my research elective in Nottingham. I enjoyed the combination of research and patient care a lot: it guided my choice for the combined research and residency training programme. Your wife Chris and you gave me a wonderful time in Nottingham, thank you so much!

Henriëtte van de Horst (hoogleraar huisartsgeneeskunde VUMC Amsterdam), jij bent natuurlijk hét voorbeeld voor een vrouwelijke AIOTHO. In Singapore hoorde ik je verhalen dan ook graag aan. Super dat jij mijn proefschrift beoordeelde en een van de opponenten bent. Bedankt voor je commentaar! Ook Anne Speckens (hoogleraar psychiatrie UMC St Radboud Nijmegen) en Giel Hutschemaekers (hoogleraar sociale wetenschappen Radboud Universiteit Niimegen). bedankt voor jullie inspanningen ten behoeve van de beoordeling van mijn proefschrift.

Tony Kendrick, thanks for the interesting and enjoyable meetings during Wonca and NAPCRG. I feel honored having you as one of my opponents during the public defence. I presume you'll be the Dancing Star at the party afterwards...

Vervolgens ben ik veel dank verschuldigd aan de 81 huisartsen in opleiding die bij dit onderzoek betrokken waren. Zonder jullie zou dit onderzoek simpelweg niet mogelijk geweest zijn! Hoewel het laatste opleidingsjaar al zo druk was en jullie verplicht werden tot onderzoeksdeelname, werkten jullie respectvol mee aan 'mijn' onderzoek. Ik heb dit niet vanzelfsprekend gevonden en dank jullie dan ook hartelijk voor de medewerking!

In het verlengde hiervan bedank ik de opleiders van alle deelnemende praktijken voor hun goedkeuring en

168 - 169

waar nodig stimulans voor het onderzoeksproject. Daarbij betrek ik ook graag de praktijkassistentes die onze onderzoeksassistentes vriendelijk welkom heetten en wegwijs maakten in de praktijk waardoor de selectie van patiënten gerealiseerd kon worden. Daarnaast dank ik de groepsbegeleiders van de huisartsopleiding in Nijmegen: jullie kanaliseerden weerstand bij de huisartsen in opleiding en bleven scherp kijken naar de praktische haalbaarheid van onze onderzoekswensen. 'De dames van het secretariaat' zijn onmisbaar: jullie handelen alle praktische regelzaken even adequaat en vlug af, fantastisch! Angèle, Elly, Franka, Karlijn, Marion, Michelle en Vanessa, bedankt

Ben Bottema, dank voor je instemming met de uitvoering van dit project binnen de Nijmeegse huisartsopleiding.

Onderzoek is onmogelijk zonder goede onderzoeksondersteuning. Twanny Jeijsman-Rouwhorst, jij was er vanaf het eerste uur bij. Jouw voortvarendheid en alertheid zijn prachtig en bijzonder waardevol! Superbedankt. Carla Walk, Annelies Pellegrino, Maartje Loeffen en Margriet Straver, hartelijk dank voor de prettige samenwerking en jullie goede selectie van patiënten in de praktijken. Jan van Dorenmalen, dank voor je nauwkeurige constructie van de database voor de onderzoeksgegevens. Onderzoekstagiaires Pim Klarenbeek, Eric Wierda, Maike Engelen, veel dank voor jullie adequate werkwijze en prettige samenwerking bij het verzamelen van dossiergegevens uit de vele praktijken. Onderzoeksstagiaire Thijs Kommers, bedankt voor je adequate hulp bij de vragenlijstanalyse. Marie-José Metz en Karel van Spaendonck, hartelijk bedankt voor het deskundig

en onbevooroordeeld leiden van de focusgroepinterviews.

Onze Amsterdamse (VUMC) PST-collega's Bettine Schreuders, Patricia van Oppen en Harm van Marwijk bedank ik voor de discussies bij aanvang van mijn promotietraject. Bettine, borrelen we op de goede afloop?

Leuke collega's zijn onmisbaar voor zowel het werkplezier als het opdoen van allerlei praktische tips en ideeën. Ik bedank mijn (ex-)collega-AIOTHO's Erik Bischoff, Wouter van Dijk, Rhona Eveleigh, Floris van de Laar, Els Licht, Hilde Luijks, Lotte van de Nieuwenhof, Erwin Klein Woolthuis, Patrick Poels, Evelien Termeer, Annemarie Uijen, Caroline van Wayenburg, en Mark van der Wel voor alle gezelligheid, kritische discussies en praktische tips. Rhona, bedankt voor je Engelse tips. Floris, fijn dat jij RevMan wel snapte als ik dat niet deed. Lotte, bedankt voor al je praktische en zorgzame tips. Annemarie, wat fijn dat wij nu - in navolging op onze Singaporese hotelkamer - kamergenoten zijn! Caroline, wanneer fietsen we de Amstel Gold Race weer? En pseudo-AIOTHO Hiske van Ravesteijn, geweldig om met jou te fantaseren over de ideale dokter, dikke buiken en kinderen.

Rob Baltussen, Lisette van de Bemt, Linelle Deunk, Nathalie Donders, Wim Gorgels, Annemarie de Graaff, André Haverkort, Dorothé Jackson, Anneke Meijsen, Dragana Milic, Lea Peters, Tilly Pouwels, Eric van Rijswijk, Caroline Roos, Tjard Schermer, Marianne Schoevers, Petra Verdonk, Saskia Zandstra en alle andere Eerstelijnsgeneeskunde (ex-)collega's, bedankt voor jullie belangstelling en gezelligheid op de gang, tijdens de lunch of op congres (lange leve de family rooms, Saskia!) als ook voor alle hulp bij print- of RefMan-problemen (Nathalie!). En Caroline, hopelijk gaan we nog eens naar Loreena McKennitt...

Dan zijn er velen die niet direct iets met onderzoek te maken hadden maar het mij wel mogelijk maakten mijn onderzoek uit te voeren. Ten eerste Huisartsenpraktijk St Annastraat en Huisartsencentrum Zetten: Nico Vlaar en Paul van der Weerd, ontzettend bedankt voor de ruimte die jullie mij gaven voor mijn onderzoek. Maar vooral veel dank voor jullie inspiratie tot het worden van een goede huisarts. Daar gaat het uiteindelijk immers om, de beste dokter voor onze patiënten ziin! Natuurliik bedank ik ook alle andere dokters en alle assistentes uit deze praktijken. Jullie hadden telkens veel begrip en belangstelling voor hetgeen waarmee 'Lieke altijd druk was'.

Ten tweede, essentieel in 'het leven van een wetenschapper', de mensen dichtbij die voor ontspanning, relativering en soms kritische vragen zorgen. Lieve Femke en Mieke, exhuisgenootjes alias 'zusjes', met jullie is het simpelweg altijd dikke pret. Miek, je moederlijke adviezen – en heerlijke soepjes! – waardeer ik zeer. En Femmie, hoe jammer ik het ook vind dat je niet lijfelijk aanwezig bent, in gedachten ben je gewoon ook mijn paranimf!

Hopeliik gaat de verdediging zoals onze Vierdaagse: onderweg af en toe zweten maar lachend de eindstreep halen, en dan... bierrr. Brechtie, querida amiga, in El Salvador ontdekten we samen wat er in het leven het meest toe doet. Het is heerlijk dat we die dingen sindsdien zo intens met elkaar delen. Met jou bijpraten is weldadiger dan welk warm bad ook! Lieve Wendy en Niels, avondjes met jullie staan garant voor geanimeerde discussies, soms met een 'huisartsenfront' versus een 'specialistenfront'. En Wens, nuchtere vriendin van het eerste uur, wanneer laten we ons weer vertroetelen?

Lieve Irmgard, jouw perspectief als niet-medicus was altijd erg waardevol. Hopelijk zullen we – voortaan met kinderen erbij! – nog vaak bijkletsen. Lieve Fenna en Pieter Lindenbergh, mijn 'oudste' vrienden, nadat ik Pieters proefschrift op de plank heb gezet, kunnen jullie nu het mijne op jullie plank zetten. Bedankt voor alle oprechte belangstelling en medeleven door de jaren heen.

Hardloopmaatje Elise, bedankt voor al onze tochtjes door de Ooijpolder en - toen dat op een lager pitje kwam te staan – je talent als oppas-nanny. Lieve Nienke en Bart, jullie belangstelling als ervaringsdeskundigen was erg plezierig, ik hoop op nog veel leuke ontmoetingen met jullie. Ook mijn andere vrienden, familieleden en buren dank ik hartelijk voor hun interesse in de voortgang van mijn onderzoek dan wel boekje.

Beste Gerhard, Marian, Esther, Dennis, Ruben, Jasper, Marloes en Robert, leuk zoals jullie altijd deelden in de vreugde als ik een artikel publiceerde. Daarnaast zijn jullie een geweldige 'oppascentrale'! Hartelijk dank voor de hulp, belangstelling en gezelligheid.

Lieve Lude en Gert, topbroers, ik ben apetrots op jullie! Ik waardeer jullie enorme optimisme, enthousiasme en betrokkenheid. Daarnaast heb ik het aan jullie perfectionisme en werklust te danken dat dit boekje zo prachtig is geworden, bedankt! Gert, het is super dat jij mijn paranimf bent. Je doet toch wel je gele sneakers aan? Lieve Madelien, met jou erbij is het alleen maar nog gezelliger! Bedankt voor je hulp bij de review en onze leuke gesprekken.

Lieve mama en papa, jullie hebben me altijd enorm gestimuleerd om me te ontwikkelen tot wie ik nu ben. Jullie optimistische levenskijk en 'niet zeuren maar aanpakken' bleken daarbij waardevol. Jullie waren altijd intens betrokken en hielpen waar mogelijk, van verhuizen en oppassen tot de samenvatting van dit proefschrift. Jullie grote enthousiasme voor jullie idealen en activiteiten is een prachtig voorbeeld voor me! Mam, pap, bedankt voor jullie onvoorwaardelijke steun en vertrouwen. Ik weet dat jullie er altijd voor me zijn.

Lieve Jaap en Huub, jullie zijn mijn grootste trots! Wat ontzettend gezellig dat jullie in ons midden zijn gekomen. Betere afleiding had ik me niet kunnen voorstellen: lekker knuffelen en wandelen, samen de kipjes voeren en eindeloos boekjes lezen. Ik hoop dat jullie altijd zulke vrolijke mannetjes blijven en dat we jullie nog heel lang in ons leven mogen volgen.

Liefste Rutger, sportieve collega van me, met een beetje mazzel verlies ook ik dan nu mijn -s. Toch is de mooiste uitkomst van mijn aiotho-traject: jij aan mijn zij! Met je droge humor relativeer je ieder obstakel, en je grote ijver stimuleert mij om de schouders eronder te zetten. Maar wat nu? Rome is bereikt, de kippen scharrelen rond, en mijn boekje is af! Zullen we even PST-en voor een nieuw doel? Lief, ik hou van je en heb zin in de komende 60 jaar.

List of publications

Franke L, Avery AJ, Groom L, Horsfield P. Is there a role for computerized decision support for drug dosing in general practice? A questionnaire survey. Journal of Clinical Pharmacology and Therapeutics 2000;25:373-7.

olde Hartman TC, **Franke LJ**, Borghuis MS, Lucassen PL. Becoming a research-oriented family physician. Family Medicine 2005;37:686-7.

Hillen-de Bruijn K, **Franke L**. Ingezonden brief n.a.v. Matige klachten van recidiverende keelontstekingen of hypertrofie van de neusen keelamandelen: adenotonsillectomie niet effectiever dan een expectatief beleid; een gerandomiseerde interventiestudie. Nederlands Tijdschrift voor Geneeskunde 2006; 150;342.

Franke L. Prikken in dikke billen. Huisarts&Wetenschap, juni 2006. Journaalstukje.

Hassink RJ, **Franke LJA**. Tijd voor een rookvrije horeca! De argumenten stapelen zich op. Nederlands Tijdschrift voor Geneeskunde 2007;151(8):488-91.

Van Weel-Baumgarten EM, **Franke LJA**. Pillen en Praten. Hoe doe je dat ook alweer? Bijblijven, 2007[1]. Franke L. Geen postpartumdepressie bij nieuwe vaders. Huisarts&Wetenschap, maart 2007. Journaalstukje.

Franke L, Van Weel-Baumgarten E, Lucassen P, Beek M, Mynors-Wallis L,Van Weel C. Feasibility of training in problem-solving treatment for General Practice registrars. European Journal of General Practice 2007;13(4):243-5.

Franke LJA, olde Hartman TC. Towards a research-oriented culture in primary care. British Medical Journal 2007 Rapid response to Sullivan F et al. Primary care research networks in the United Kingdom. British Medical Journal 2007;334:1093-4.

Franke LJA, Kommers T, Van Weel-Baumgarten EM, Lucassen PLBJ, Beek M, Van den Hoogen HH, Van Weel C. General practice registrars and research – attitudes toward participation. Australian Family Physician 2008;37(4):276-9.

olde Hartman TC, **Franke L**, Lucassen P, van Weel-Baumgarten E. Commentary in response to Escobar JI et al. Effectiveness of a time-limited cognitive behavior therapy-type intervention among primary care patients with medically unexplained symptoms by Escobar et al. Annals of Family Medicine 2007;5:328-35. Olde Hartman T, **Hassink-Franke L**, Dowrick C, Fortes S, Lam C, van der Horst H, Lucassen P, van Weel-Baumgarten E. Medically unexplained symptoms in family medicine: defining a research agenda. Proceedings from WONCA 2007. Family Practice 2008;25(4):266-71.

Uijen AA, Hassink-Franke LJA Blood pressure measurement in hemiparetic patients: at which arm? Family Medicine 2008;40(8):540.

Hartman TO, van Rijswijk E, van Ravesteijn H, **Hassink-Franke L**, Bor H, van Weel-Baumgarten E, Lucassen P. Mental health problems and the presentation of minor illnesses: data from a 30-year follow-up in general practice. European Journal of General Practice 2008;14 Suppl 1:38-43.

olde Hartman TC, **Hassink-Franke LJA**, Lucassen PLBJ, Van Spaendonck KPM, Van Weel C. Explanation and relations. How do general practitioners deal with patients with persistent medically unexplained symptoms: a focus group study. BMC Family Practice 2009;24;10:68.

Hassink-Franke LJA, Hassink RJ. Acupunctuur bij lage rugpijn... pak een tandenstoker? Nederlands Tijdschrift voor Geneeskunde 2009;153:A850. van Weel-Baumgarten E, Lucassen P, **Hassink-Franke L**, Schers H. A different way of looking at depression. International Journal of Clinical Practice, *in press*.

Hassink-Franke L, Costa J, Lucassen P, Akkermans R, Mynors-Wallis L, van Weel C, van Weel-Baumgarten E. Problem-solving treatment for emotional problems in primary care – a Cochrane review. *Submitted*.

Hassink-Franke LJA, olde Hartman TC, Beek MM, van Weel C, Lucassen PLBJ, van Weel-Baumgarten EM. Problem-solving treatment in general practice residency: a focus group study of registrars' views. Patient education and counseling, *in press*.

Hassink-Franke LJA, van Weel-Baumgarten EM, Wierda E, Engelen MWM, Beek MM, Bor HHJ, van den Hoogen HJM, Lucassen PLBJ, van Weel C. Effectiveness of problem-solving treatment by general practice registrars for patients with emotional symptoms: a controlled trial. *Submitted*.

Hassink RJ, **Hassink-Franke LJA**. Hartritmestoornissen. Accredidact nascholing voor huisartsen en apothekers, *in press*.

Curriculum Vitae

Curriculum Vitae in het Nederlands

Op 4 september 1977 word ik in Oldenzaal geboren als dochter van Harry Franke en Ria Franke-Hams. Ik groei op met mijn twee broers, Lude en Gert. Op mijn 9^e verhuizen we naar Budel en op miin 15^e naar Epe. Ik ga naar het Bisschoppelijk College in Weert en behaal in 1995 mijn gymnasiumdiploma aan het Christelijk Lyceum in Apeldoorn. Vervolgens studeer ik geneeskunde aan de Radboud Universiteit Nijmegen (voorheen 'Katholieke Universiteit Nijmegen'). In 1999 doe ik een wetenschappelijke stage bij Professor Anthony Avery aan de University of Nottingham (Groot-Brittannië), alwaar mijn enthousiasme voor wetenschappelijk onderzoek en het huisartsenvak groeit. Na mijn doctoraaldiploma geneeskunde studeer ik in 1999-2000 een aanvullend jaar Ethiek aan de faculteit Wijsbegeerte te Nijmegen met een studiebeurs van de Stichting Thomas More (voorheen 'Radboudstichting'). Daarna volgen mijn co-schappen welke ik afsluit met een co-schap van 3 maanden in een praktijk op het platteland in El Salvador, Midden-Amerika. Na deze ervaring weet ik zeker dat ik huisarts wil worden. In

2002 behaal ik mijn artsexamen. Na een korte overbruggingsperiode bij Bloedbank Sanguin te Nijmegen, start ik in 2003 met mijn promotieonderzoek naar problem-solving treatment (PST) waarvan de resultaten in dit proefschrift zijn beschreven. Daarnaast werk ik in 2003 als arts-assistent in de ouderenpsychiatrie bij de GGz Nijmegen. In 2004 begin ik met de huisartsopleiding te Nijmegen, welke ik combineer met mijn onderzoek in een zogenaamd AIOTHO-traject (arts in opleiding tot huisarts en onderzoeker). In 2008 rond ik mijn huisartsopleiding af. Vanaf 2004 geef ik PST-trainingen aan diverse hulpverleners in den lande alsmede in de Nijmeegse huisartsopleiding. Momenteel werk ik parttime als huisarts in de regio Utrecht. Daarnaast werk ik binnen het UMC St Radboud te Nijmegen mee aan een onderzoeksproject over ADHD (Afdeling Kinder- en Jeugdpsychiatrie) en onderwijs ik consult- en communicatievaardigheden aan medisch studenten te Nijmegen. Tevens ben ik lid van de werkgroep herziening NHG-standaard angststoornissen. Ik ben getrouwd met Rutger Hassink. Samen genieten wij van onze zonen Jaap (10 juli 2008) en Huub (12 juli 2010).

Huidig woonadres: Lieke Hassink-Franke Rijfveld 38 3993 SN Houten T (0031) 30 2145272 M (0031) 6 29495828 lieke_franke@hotmail.com

Curriculum Vitae in English

I was born on September 4th 1977 in Oldenzaal (the Netherlands) as the daughter of Harry Franke and Ria Franke-Hams. I grew up with my two brothers. Lude and Gert. When I was 9 we moved to Budel and at the age of 15 to Epe. I went to the Bischoppelijk College in Weert and completed my high school (gymnasium) in 1995 at the Christian Lyceum in Apeldoorn, I then studied medicine at the Radboud University Niimegen (previously 'Catholic University of Nijmegen'). In 1999 | conducted a research elective at the University of Nottingham (UK). supervised by Professor Anthony Avery. During this period, my enthusiasm for scientific research and general practice increased. With a grant from the Thomas More Foundation (previously 'Radboudstichting') in 1999-2000, I studied Ethics at the Faculty of Philosophy in Nijmegen for a year. Hereafter I started my internships with a final 3-month internship at a rural health centre in El Salvador, Central America, I obtained my medical degree in 2002. After a brief period working at the Sanguin Blood Bank Nijmegen, I started my PhD project on problemsolving treatment (PST) in 2003. The results of this project are described in this thesis. In addition, in 2003 I also worked as a psychiatry resident for the elderly at the GGz Niimegen. In 2004 I started with the general practice (GP) residency training at the Radboud University Niimegen Medical Centre, being part of a combined residency and research training programme (the 'AIOTHO-programme'). In 2008 I finished my GP training. Since 2004 I have provided PST training to various health professionals in the country, including the Nijmegen GP residency training programme. At the moment I am working part-time as a general practitioner in Utrecht and surroundings. In addition, I am working in the Radboud University Nijmegen Medical Centre on a research project on ADHD (Department of Psychiatry) and I am a GP-teacher in consultation and communication skills for medical students. Also, I am a member of the working party on the revision of the national general practice guideline of anxiety disorders. I am married to Rutger Hassink. Together we enjoy our sons Jaap (July 10th 2008) and Huub (July 12th 2010).

Current home address: Lieke Hassink-Franke Rijfveld 38 3993 SN Houten The Netherlands T: 0031 30 2145272 M: 0031 6 29495828 lieke_franke@hotmail.com

Colophon

This thesis has been prepared by the Department of Primary and Community Care of the Radboud University Nijmegen Medical Centre, the Netherlands. The Department of Primary and Community Care participates in the Netherlands School of Primary Care research (CaRe), which has been acknowledged by the Royal Netherlands Academy of Arts and Sciences (KNAW) in 1995.

- Design Gert Franke Lude Franke
- Printer Gildeprint Drukkerijen www.gildeprint.nl

Financial Support

- This thesis was made possible by grants of
- ZonMw (the Netherlands Organization for Health Research and Development, grant no. 4200.0002)
- Department of Primary and Community Care, Radboud University Nijmegen Medical Centre,
- SBOH, employer of general practitioner trainees



Copyright © L.J.A. Hassink-Franke, Houten, the Netherlands, 2010. All rights reserved. No part of this book may be reproduced or transmitted in any form, by print, photo print, microfilm or otherwise, without prior written permission of the holder of the copyright.

Niets uit deze uitgave mag worden vermenigvuldigd en/of openbaar gemaakt door middel van druk, fotokopie, microfilm, of welke andere wijze dan ook, zonder schriftelijke toestemming van de auteur.

SBN: 978-90-9025598-9